## UNITED STATES SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549

#### FORM 10-Q

### [ X ] QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

#### For the quarterly period ended September 30, 2002

or

## [ ] TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

#### Commission File No. 33-98136

For the transition period from \_\_\_\_\_\_ to \_\_\_

#### CPG PARTNERS, L.P.

(Exact name of registrant as specified in its charter)

### Delaware 22-3258100

(State or other jurisdiction of incorporation or organization)

(I.R.S. Employer Identification No.)

#### 103 Eisenhower Parkway, Roseland, New Jersey 07068

(Address of principal executive offices - zip code)

#### (973) 228-6111

(Registrant's telephone number, including area code)

Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days Yes <u>X</u> No\_\_.

There are no outstanding shares of Common Stock or voting securities.

#### CPG Partners, L.P.

#### Index

#### Part I. Financial Information

| Item 1. | <u>Financial Statements (Unaudited)</u>   | Page |
|---------|---|------|
|         | Condensed Consolidated Balance Sheets as of September 30, 2002 and December 31, 2001                        | 3    |
|         | Condensed Consolidated Statements of Income for the three and nine months ended September 30, 2002 and 2001 | 4    |
|         | Condensed Consolidated Statements of Cash Flows for the nine months ended September 30, 2002 and 2001       | 5    |
|         | Notes to Condensed Consolidated Financial Statements  | 6    |
| Item 2. | Management's Discussion and Analysis of Financial Condition and Results of Operations                       | 20   |
| Item 3. | Quantitative and Qualitative Disclosures about Market Risk  | 28   |
| Item 4. | Controls and Procedures   | 28   |
|         |   |      |

#### Part II. Other Information

#### **Item 1. Financial Statements**

#### CPG Partners, L.P. Condensed Consolidated Balance Sheets (In thousands, except per unit data)

|   | September 30,<br>2002   | December 31,<br>2001  |
|---|---|---|
| ·   | (Unaudited)   |   |
| Assets:   |   |   |
| Rental properties:  | ****  | ****  |
| Land Depreciable property   | \$216,763<br>1,324,535  | \$160,458<br>967,448  |
| Total rental property   | 1,541,298   | 1,127,906   |
| Accumulated depreciation  | (269,849)   | (217, 462)  |
| Rental properties, net  | 1,271,449   | 910,444   |
| Cash and cash equivalents   | 13,693  | 24,604  |
| Restricted cash-escrows   | 4,212   | 3,347   |
| Investments in unconsolidated affiliates  | 74,093  | 93,689  |
| Notes receivable-related parties  | 2,746   | 3,281   |
| Deferred costs, net   | 15,835  | 22,534  |
| Other assets  | 48,656  | 41,409  |
| Total assets  | \$1,430,684   | \$1,099,308   |
|   | =========   | =========   |
| Liabilities and partners' capital: Liabilities: Unsecured bank debt. Unsecured notes. Mortgage debt. Construction payables. Accounts payable and accrued expenses. Accrued dividend and distribution payable Other liabilities. | \$97,035<br>472,647<br>307,677<br>8,034<br>36,026<br>22,711<br>22,676 | \$5,035<br>373,294<br>170,209<br>8,001<br>47,039<br>3,851<br>16,817 |
| Total liabilities   | 966,806   | 624,246   |
| Commitments and contingencies   |   |   |
| Partners' capital:  |   |   |
| General partner units outstanding, 37,957 in 2002 and 37,540 in 2001  | 355,981   | 362,402   |
| Limited partners units outstanding 6,273 in 2002 and 6,300 in 2001  | 52,171  | 52,825  |
| Preferred partner units outstanding, 1,300 in 2002 and 2001   | 63,315  | 63,315  |
| Officer loan  | (488)   | -   |
| Accumulated other comprehensive loss  | (7,101)   | (3,480)   |
| Total partners' capital   | 463,878   | 475,062   |
| Total liabilities and partners' capital   | \$1,430,684   | \$1,099,308   |
|   | =========   | ==========  |

The accompanying notes are an integral part of the financial statements.

# CPG Partners, L.P. Condensed Consolidated Statements of Income for the Three and Nine Months Ended September 30, 2002 and 2001 (Unaudited)

(In thousands, except per unit data)

|  | Three Months Ended<br>September 30, |          | Nine Months Ended<br>September 30, |          |
|--|-------------------------------------|----------|------------------------------------|----------|
|  | 2002                                | 2001     | 2002                               | 2001     |
| Revenues:                                    |                                     |          |                                    |          |
| Base rent                                    | \$46,968                            | \$30,528 | \$128,776                          | \$87,930 |
| Percentage rent                              | 5,402                               | 4,051    | 12,952                             | 9,851    |
| Expense reimbursements                       | 16,128                              | 11,581   | 43,265                             | 34,002   |
| Other income                                 | 2,991                               | 2,905    | 8,207                              | 8,403    |
| Other Income                                 | 2,991                               | 2,905    | 0,201                              | 0,403    |
| Total revenues                               | 71,489                              | 49,065   | 193,200                            | 140,186  |
| Expenses: Operating and maintenance          | 19,389                              | 12,779   | 53.779                             | 37,505   |
| Depreciation and amortization                | 15,045                              | 12,174   | 42,229                             | 35,479   |
| General and administrative                   | 1,558                               | 622      | 4,996                              | 3,053    |
| Other  | 1,027                               | 473      | 3,221                              | 1,713    |
|  |                                     |          |                                    |          |
| Total expenses                               | 37,019                              | 26,048   | 104,225                            | 77,750   |
| Income before unconsolidated investments and |                                     |          |                                    |          |
| interest expense                             | 34,470                              | 23,017   | 88,975                             | 62,436   |
| Income from unconsolidated investments       | 2,032                               | 3,953    | 8,784                              | 10,509   |
|  |                                     |          |                                    |          |

| Loss from Chelsea Interactive  | (3,790)<br>10,911 | (1,477)   | (10,266)<br>10,911 | (3,447)  |
|--|-------------------|-----------|--------------------|----------|
| Interest expense   | (13,098)          | (8,891)   | (33,691)           | (26,326) |
| 2.1.601 вой сиропоститительной поставления |                   | (0,001)   |                    |          |
| Net income   | 30,525            | 16,602    | 64,713             | 43,172   |
| Preferred dividend   | (2,297)           | (2,509)   | (6,974)            | (7,527)  |
|  |                   |           |                    |          |
| Net income to common unitholders   | \$28,228          | \$14,093  | \$57,739           | \$35,645 |
|  | =========         | ========= | ========           | ======== |
| Net income to common unitholders:  |                   |           |                    |          |
| General partner  | \$24,223          | \$11,838  | \$49,518           | \$29,818 |
| Limited partners   | 4,005             | 2,255     | 8,221              | 5,827    |
| Total  | \$28,228          | \$14,093  | \$57,739           | \$35,645 |
|  | =========         | ========= | ========           | =======  |
| Net income per common unit:  |                   |           |                    |          |
| General partner  | \$0.64            | \$0.36    | \$1.31             | \$0.92   |
| Limited partners   | \$0.64            | \$0.36    | \$1.31             | \$0.92   |
| Weighted average units outstanding:  |                   |           |                    |          |
| General partner  | 37,950            | 33,136    | 37,799             | 32,538   |
| Limited partners   | 6,274             | 6,310     | 6,285              | 6,378    |
| Total  | 44,224            | 39,446    | 44,084             | 38,916   |

The accompanying notes are an integral part of the financial statements.

# CPG Partners, L.P. Condensed Consolidated Statements of Cash Flows for the Nine Months Ended September 30, 2002 and 2001 (Unaudited)

(În thousands)

|   | 2002                | 2001               |
|---|---------------------|--------------------|
| Cash flows from operating activities  |                     |                    |
| Net income  | \$64,713            | \$43,172           |
| Depreciation and amortization<br>Equity-in-earnings of unconsolidated investments in  | 42,229              | 35,479             |
| excess of distributions received  | (2,148)             | (1,956)            |
| Loss from Chelsea Interactive   | ì0,266              | 3,447              |
| Gain on sale of unconsolidated investment   | (10,911)            | · -                |
| Proceeds from non-compete receivable  | 4,300               | 4,600              |
| Amortization of non-compete revenue   | (3,852)             | (3,852)            |
| Additions to deferred lease costs   | (603)               | (391)              |
| Other operating activities  | (641)               | (349)              |
| Straight-line rents   | (2,441)             | (1,181)            |
| Due from affiliates   | (806)               | (787)              |
| Other assets  | 3,118               | 4,670              |
| Deferred incentive compensation payout  | (14,401)            |                    |
| Accounts payable and accrued expenses   | (4,747)             | (1,310)            |
| Net cash provided by operating activities   | 84,076              | 81,542             |
|   |                     |                    |
| Cash flows from investing activities  |                     |                    |
| Additions to rental properties  | (133,502)           | (89,572)           |
| Proceeds from sale of centers   | 6,873               | -                  |
| Additions to investments in unconsolidated affiliates   | (26,098)            | (15,960)           |
| Proceeds from sale of investment in unconsolidated affiliate<br>Distributions from investments in   | 11,293              | -                  |
| unconsolidated affiliates in excess of earnings   | 337                 | 5,090              |
| Additions to deferred development costs   | (718)               | (6,074)            |
| Payments from loans to related parties  | 1,085               | 1,725              |
| Loans to related parties  | (550)               | (2,750)            |
| Net cash used in investing activities   | (141,280)           | (107,541)          |
|   |                     |                    |
| Cash flows from financing activities  | 222 224             | 475 455            |
| Debt proceeds   | 233,324             | 175,155            |
| Repayment of debt  Net proceeds from sale of the Company's common stock   | (132,400)           | (134,750)          |
| Distributions   | 5,089<br>(48,851)   | 26,110<br>(40,208) |
| Redemption of the Company's preferred stock   | (9,654)             | (40,200)           |
| Additions to deferred financing costs   | (1,215)             | (2,054)            |
| Additions to deterred financing costs   | (1,213)             | (2,034)            |
| Net cash provided by financing activities   | 46,293              | 24, 253            |
|   |                     |                    |
| Net decrease in cash and cash equivalents   | (10,911)<br>24,604  | (1,746)            |
| Cash and cash equivalents, beginning of period  | 24,604              | 18,036             |
|   |                     | \$16,290           |
| Cash and cash equivalents, end of period  | \$13,693            |                    |
| Cash and cash equivalents, end of period  | \$13,693<br>======= | =========          |
|   |                     | ========           |
|   |                     | ========           |
| Non-cash investing activities:  | \$275,960           | =========          |
| previously held as investments in unconsolidated affiliates   | ========            |                    |
| Non-cash investing activities:  Additions to rental properties on consolidation of properties previously held as investments in unconsolidated affiliates  Non-cash financing activities: | \$275,960           |                    |
| Non-cash investing activities:<br>Additions to rental properties on consolidation of properties<br>previously held as investments in unconsolidated affiliates                            | \$275,960           |                    |

The accompanying notes are an integral part of the financial statements.

### Notes to Condensed Consolidated Financial Statements (Unaudited)

#### 1. Organization and Basis of Presentation

CPG Partners, L.P. (the "Operating Partnership" or "OP") is an operating partnership that specializes in owning, developing, leasing, marketing and managing upscale and fashion-oriented manufacturers' outlet centers. As of September 30, 2002, the OP wholly or partially owned 54 centers in 28 states and Japan containing approximately 12.5 million square feet of gross leasable area ("GLA"). The OP's portfolio is comprised of 27 Premium Outlet Centers containing 8.4 million square feet of GLA, (the "Premium Properties") and 27 other retail centers containing approximately 4.1 million square feet of GLA ("Other Properties") (collectively the "Properties"). The OP's Premium Properties generated 85% of the retail net operating income for the nine months ended September 30, 2002. These Premium Properties generally are located near metropolitan areas including New York City, Los Angeles, Boston, Washington, D.C., San Francisco, Sacramento, Cleveland, Atlanta, Dallas, Portland (Oregon), Tokyo and Osaka, Japan. Some Premium Properties are also located within 20 miles of major tourist destinations including Palm Springs, Napa Valley, Orlando, and Honolulu.

The sole general partner in the OP, Chelsea Property Group, Inc. (the "Company") is a self-administered and self-managed Real Estate Investment Trust ("REIT").

The OP has developed a technology-based e-commerce platform through an unconsolidated affiliate, Chelsea Interactive, Inc. ("Chelsea Interactive"). This platform provides fashion and other retail brands with their own customized direct-to-the-consumer Internet online stores, incorporating e-commerce design, development, fulfillment and customer services.

On May 1, 2002, the Company declared a 2-for-1 stock split of the Company's common shares. The OP simultaneously declared a 2-for-1 split of the OP's Limited Partner units. The stock/unit dividend was paid May 28, 2002 to shareholders/unitholders of record on May 14, 2002. All amounts in the accompanying financial statements reflect the stock/unit split.

Disclosure about the fair value of financial instruments is based on pertinent information available to management as of September 30, 2002, using available market information and appropriate valuation methodologies. Although management is not aware of any factors that would significantly affect the reasonable fair value amounts, such amounts have not been comprehensively revalued for purposes of these financial statements since such date and current estimates of fair value may differ significantly from the amounts presented herein.

Common ownership of the OP as of September 30, 2002 was approximately as follows:

| Total            | 100.0% | 44,230,000       |       |
|------------------|--------|------------------|-------|
| Limited Partners | 14.2%  | <u>6,273,000</u> | units |
| General Partner  | 85.8%  | 37,957,000       | units |

The accompanying unaudited condensed consolidated financial statements have been prepared in accordance with accounting principles generally accepted in the United States for interim financial information and with the instructions to Form 10-Q and Article 10 of Regulation S-X. Accordingly, they do not include all of the information and notes required by accounting principles generally accepted in the United States for complete financial statements. In the opinion of management, all adjustments (consisting of normal recurring accruals) considered necessary for fair presentation have been included.

### Notes to Condensed Consolidated Financial Statements (Unaudited)

#### 1. Organization and Basis of Presentation (continued)

Operating results for the three and nine month periods ended September 30, 2002 are not necessarily indicative of the results that may be expected for the year ending December 31, 2002. The balance sheet at December 31, 2001 has been derived from the audited financial statements at that date but does not include all of the information and footnotes required by accounting principles generally accepted in the United States for complete financial statements. These financial statements should be read in conjunction with the consolidated financial statements and accompanying notes included in the OP's Annual Report on Form 10-K for the year ended December 31, 2001.

In October 2001, the FASB issued Statement of Financial Accounting Standard ("SFAS") No. 144, "Accounting for the Impairment or Disposal of Long-Lived Assets", which addresses financial accounting and reporting for the impairment or disposal of long-lived assets. This standard harmonizes the accounting for impaired assets and resolves some of the implementation issues as originally described in SFAS 121. The new standard became effective and was adopted by the OP on January 1, 2002. The adoption of this pronouncement did not have a material impact on the OP's results of operations or financial position.

In May 2002, the FASB issued SFAS No. 145, "Reporting Gains and Losses from Extinguishment of Debt", which rescinded SFAS No. 4, No. 44 and No. 64 and amended SFAS No. 13. The new standard addresses the income statement classification of gains or losses from the extinguishment of debt and criteria for classification as extraordinary items. The new standard became effective for fiscal years beginning after May 15, 2002. The OP adopted this pronouncement on April 1, 2002. The adoption of this pronouncement did not have a material impact on the OP's results of operations or financial position.

Certain amounts in the 2001 financial statements have been reclassified to conform to the 2002 presentation.

#### 2. Acquisitions and Dispositions

#### **Acquisitions**

On April 1, 2002, the OP became the sole owner of the Orlando Premium Outlets ("OPO") by acquiring Simon Property Group, Inc.'s ("Simon") 50% undivided ownership interest for \$46.6 million in cash and the assumption of \$29.7 million of existing mortgage debt and the guarantee related thereto. On June 16, 2002, the OP repaid the outstanding balance of \$59.4 million and extinguished the mortgage. After closing the transaction the operating results and balance sheet of OPO were consolidated in the accompanying financial statements.

On April 1, 2002, the OP purchased a 305,000 square foot center located in Edinburgh, Indiana for \$27.0 million in cash. The center is classified as Other Properties.

On August 20, 2002, the OP became the sole owner of four premium outlets by acquiring Fortress' 51% undivided ownership interest in a joint venture ("F/C"). The OP paid \$58.9 million in cash and assumed \$86.5 million of existing mortgage debt on the properties. After closing the transaction, the operating results and balance sheet of the four premium outlets, including \$169.6 million of existing nonrecourse mortgage debt, were consolidated in the accompanying financial statements.

On September 27, 2002, the OP signed an agreement to acquire two outlet centers; a 305,000 square-foot center located in Albertville, Minnesota and a 278,000 square foot center located in Johnson Creek, Wisconsin for a total purchase price of approximately \$89.5 million. The transaction is expected to be financed by issuing approximately \$45.5 million of limited partnership units in the OP and the balance from the OP's Senior Credit Facility. Subject to certain conditions, the transaction is expected to close before year-end 2002 but there can be no assurance that the transaction will close.

### Notes to Condensed Consolidated Financial Statements (Unaudited)

#### 2. Acquisitions and Dispositions (continued)

#### **Dispositions**

During June and July 2002, the OP sold four Other Properties and two outparcels generating net proceeds of approximately \$6.8 million, which approximated the net book value of these properties. Accordingly, no gain or loss on the sale was recognized.

On November 1, 2002, the OP sold an additional Other Property that generated net proceeds of approximately \$0.7 million, which resulted in a gain of approximately \$0.3 million.

#### 3. Investments in Affiliates

The OP holds interests in domestic and international joint ventures. Non-controlling investments are accounted for under the equity method. Income from unconsolidated investments and loss from Chelsea Interactive includes equity-in-earnings, management, advisory, license, leasing and guarantee fees earned from these affiliates.

The OP has developed an e-commerce venture through its affiliate, Chelsea Interactive. The Board of Directors has approved a funding commitment of up to \$60.0 million. As of September 30, 2002, the OP has funded \$51.6 million and anticipates that the balance of the funding will be used for development and operating cash shortfalls. The net book value of the OP's investment as of September 30, 2002 was \$35.0 million. While management currently believes that it is possible to recover the net book value of the investment through future cash flows if revenues from existing brands improve and/or additional brands are secured, there is substantial risk that Chelsea Interactive will not: (1) generate sufficient future revenues for the OP to recover its investment; (2) achieve positive cash flow before reaching the OP's committed funding of \$60.0 million; (3) successfully complete a satisfactory arrangement with a financial/business investor; or (4) continue as a going concern. As a result of these risk factors, management may at any time determine that it is necessary to significantly write down or write off the OP's \$35.0 million net book value investment. The OP is actively pursuing and negotiating with potential investors to provide capital and further develop the business. The OP owns 100% of the non-voting preferred stock of Chelsea Interactive and 50% of the non-voting common stock representing 40% of the total common stock.

As of September 30, 2002, the OP's interests in joint ventures also included a 40% interest in Chelsea Japan Co., Ltd. ("Chelsea Japan"), a 50% interest in Las Vegas Premium Outlets ("Simon-Las Vegas") and a 50% interest in Chicago Premium Outlets ("Simon-Chicago") with Simon (collectively "Simon-Ventures") and minority interests in various outlet centers and development projects in Europe operated by Value Retail PLC ("Value Retail").

In April 2002, the OP purchased Simon's 50% undivided ownership interest in OPO for \$46.6 million in cash and assumed Simon's \$29.7 million pro-rata share of existing mortgage debt. The operations and balance sheet of OPO are consolidated in the accompanying financial statements from the buyout date.

In June 2002, the OP and Simon entered into a new 50/50 joint venture to develop and operate Simon–Las Vegas, a 430,000 square-foot single-phase premium outlet center located in Las Vegas, Nevada, scheduled to open in summer 2003. On June 20,

2002, Simon-Las Vegas purchased a 40-acre site and commenced construction. The OP will be responsible for financing its 50% share of development costs which are expected to be approximately \$48.0 million.

### Notes to Condensed Consolidated Financial Statements (Unaudited)

#### 3. Investments in Affiliates (continued)

In July 2002, the joint venture partner in F/C exercised its option under the F/C Limited Liability Company Agreement to invoke the "Buy/Sell Right". This gave the OP the option to sell its 49% ownership interest in F/C or buy the partner's 51% ownership interest. On August 20, 2002, the OP exercised its right to buy the 51% interest for cash of \$58.9 million. The operations and balance sheet of the four F/C premium outlet centers were consolidated in the accompanying financial statements from the buyout date.

In August 2002, the OP and Simon entered into a new 50/50 joint venture to develop and operate Simon-Chicago, a 435,000 square-foot single-phase premium outlet center located in Aurora, Illinois, near Chicago. The center is scheduled to open in mid-2004. On September 23, 2002, Simon-Chicago purchased a 140-acre site, including 80 acres of conservation area, and commenced construction. The OP will be responsible for financing its 50% share of the development costs which are expected to be approximately \$46.0 million.

In July 2002, the OP sold approximately 40% of its holdings in Value Retail to a third party for \$11.4 million, resulting in a gain of approximately \$10.9 million which was recorded as a gain on sale of unconsolidated investment in the accompanying financial statements.

The following is a summary of investments in and amounts due from affiliates as of September 30, 2002 (in thousands):

|  | F/C      | Chelsea<br>Japan | Simon-<br>Orlando | Simon-<br>Ventures | Chelsea<br>Interactive | Other   | Total    |
|--|----------|------------------|-------------------|--------------------|------------------------|---------|----------|
|  |          |                  |                   |                    |                        |         |          |
| Balance 12/31/01<br>Additional                 | \$36,118 | \$ 9,296         | \$9,723           | \$ -               | \$ 34,856              | \$3,696 | \$93,689 |
| investment                                     | -        | 13               | -                 | 24,410             | 10,968                 | 303     | 35,694   |
| Dispositions Income (loss) from unconsolidated | (36,531) | -                | (9,199)           | -                  | -                      | (382)   | (46,112) |
| investments                                    | 4,331    | 3,116            | 1,310             | -                  | (10,266)               | 27      | (1,482)  |
| Distributions and                              |          |                  |                   |                    |                        |         |          |
| fees   | (2,841)  | (1,345)          | (250)             | -                  | -                      | -       | (4,436)  |
| Advances (net)                                 | (1,077)  | (374)            | (1,584)           | 364                | (539)                  | (50)    | (3,260)  |
| Balance 9/30/02                                | \$ -     | \$10,706         | \$ -              | \$24,774           | \$ 35,019              | \$3,594 | \$74,093 |

### Notes to Condensed Consolidated Financial Statements (Unaudited)

#### 3. Investments in Affiliates (continued)

The OP's share of income (loss) before depreciation, depreciation expense and income (loss) from unconsolidated investments for the three and nine months ended September 30, 2002 and 2001 are as follows (in thousands):

|                        |   | For the Three Months Ended September 30, |   |   |                    |   |  |
|------------------------|---|--|---|---|--------------------|---|--|
|                        |   | 2002                                     |   | 2001                                    |                    |   |  |
|                        | Income (loss)<br>before<br>Depreciation | Depr.                                    | Income<br>(loss) from  <br>Unconsol.  <br>Investments | Income (loss)<br>before<br>Depreciation | Depr.              | Income<br>(loss) from<br>Unconsol.<br>Investments |  |
| F/C<br>Chelsea         | \$1,369                                 | \$ 400                                   | \$ 969  | \$2,345                                 | \$ 677             | \$1,668   |  |
| Japan<br>Simon-Orlando | 1,532                                   | 471                                      | 1,061   | 1,214<br>2,077                          | 382<br>569         | 832<br>1,508                                      |  |
| Other                  | 2                                       | -  | 2   | (55)                                    | -                  | (55)  |  |
| Total                  | \$2,903<br>======                       | \$ 871<br>=======                        | \$2,032  <br>=======                                  | \$5,581                                 | \$1,628            | \$3,953<br>========                               |  |
| Chelsea<br>Interactive | (\$1,594)                               | \$2,196<br>======                        | (\$3,790)   | (\$ 580)<br>=======                     | \$ 897<br>======== | (\$1,477)   |  |

|                               | For the Nine Months Ended September 30, |                         |  |   |                           |   |
|-------------------------------|---|-------------------------|--|---|---------------------------|---|
|                               | 2002                                    |                         |  | 2001                                    |                           |   |
|                               | Income (loss)<br>before<br>Depreciation | Depr.                   | Income   (loss) from   Unconsol.   Investments | Income (loss)<br>before<br>Depreciation | Depr.                     | Income<br>(loss) from<br>Unconsol.<br>Investments |
| F/CChelsea JapanSimon-Orlando | \$6,178<br>4,419<br>1,833<br>27         | \$1,847<br>1,303<br>523 | \$4,331  <br>  3,116  <br>  1,310  <br>  27    | \$6,574<br>3,389<br>5,272<br>(26)       | \$1,994<br>1,001<br>1,705 | \$4,580<br>2,388<br>3,567<br>(26)                 |

|                      | =  |            | ======== | =========  |           | ======== | ========= |
|----------------------|----|------------|----------|------------|-----------|----------|-----------|
| Chelsea<br>Interacti | ve | (\$4,502)  | \$5,764  | (\$10,266) | (\$1,497) | \$1,950  | (\$3,447) |
|                      | =  |            | =======  | ========   |           | ======== | ========  |
| Total.               |    | . \$12,457 | \$3,673  | \$8,784    | \$15,209  | \$4,700  | \$10,509  |
|                      | -  |            |          |            |           |          |           |

### Notes to Condensed Consolidated Financial Statements (Unaudited)

#### 3. Investments in Affiliates (continued)

Condensed financial information as of September 30, 2002 and December 31, 2001 and for the three and nine month periods ended September 30, 2002 and 2001 for F/C, Chelsea Japan, Simon-Orlando and Simon-Ventures (which are included in "Retail") and Chelsea Interactive is as follows (in thousands):

| <u></u>                                 | Retail             | Chelsea<br>Interactive |
|---|--------------------|------------------------|
| Property, plant and equipment (net)     |                    |                        |
| September 30, 2002<br>December 31, 2001 | 109,933<br>355,427 | \$32,910<br>29,923     |
| Total assets                            |                    |                        |
| September 30, 2002<br>December 31, 2001 | 151,851<br>409,944 | 34,109<br>34,963       |
| Long term debt (1)                      |                    |                        |
| September 30, 2002 December 31, 2001    | 57,393<br>282,057  | -<br>-                 |
| Total liabilities                       |                    |                        |
| September 30, 2002                      | 95,245<br>321,854  | 1,910<br>2,665         |
| ,                                       | 022,00.            | 2,000                  |
| Net income (loss) Three months ended:   |                    |                        |
| September 30, 2002September 30, 2001    | 2,501<br>5,482     | (3,790)<br>(2,954)     |
| Nine months ended:                      | 0, 102             | (2700.)                |
| September 30, 2002                      | 11,549             | (11,127)               |
| September 30, 2001                      | 13,864             | (8,195)                |
| OP's share of net income (loss)         |                    |                        |
| Three months ended: September 30, 2002  | 1,115              | (3,790)                |
| September 30, 2001                      | 2,642              | (1,477)                |
| Nine months ended:                      |                    | (40,000)               |
| September 30, 2002September 30, 2001    | 5,374<br>6,654     | (10,266)<br>(3,447)    |
| Fee income                              | .,                 | (-, ,                  |
| Three months ended:                     |                    |                        |
| September 30, 2002                      | 915                | -                      |
| September 30, 2001                      | 1,366              | -                      |
| Nine months ended: September 30, 2002   | 2 202              |                        |
| September 30, 2001                      | 3,383<br>3,881     | -<br>-                 |

(1) Long-term debt includes \$57.4 million in Chelsea Japan as of September 30, 2002 and \$170.5 million in F/C, \$58.5 million in Simon-Orlando and \$53.1 million in Chelsea Japan as of December 31, 2001.

### Notes to Condensed Consolidated Financial Statements (Unaudited)

#### 4. Non-Compete Agreement

The OP recognized income from its non-compete agreement with The Mills Corporation of \$1.3 million during the three months ended September 30, 2002 and 2001 and \$3.9 million during the nine months ended September 30, 2002 and 2001. Such amounts are included in other income in the accompanying financial statements.

#### 5. Debt

#### **Unsecured Bank Debt**

In July 2002, the OP increased its senior unsecured revolving bank line of credit to \$200.0 million (the "Senior Credit Facility"). Under the new terms, the Senior Credit Facility's expiration is March 31, 2005, and the OP has the right to extend the maturity until March 31, 2006. The Senior Credit Facility bears interest on the outstanding balance, payable monthly, at a rate equal to the London Interbank Offered Rate ("LIBOR") plus 1.05% (2.88% at September 30, 2002) or the prime rate, at the OP's option. The LIBOR rate spread ranges from 0.85% to 1.50% depending on the OP's Senior Debt rating. A facility fee of 0.125% on the Senior Credit Facility is payable annually in advance. At September 30, 2002, \$92.0 million was outstanding under the Senior Credit Facility.

The OP also has a \$5.0 million term loan (the "Term Loan") that carries the same interest rate and maturity as the Senior Credit Facility.

#### **Unsecured Notes**

A summary of the terms of the unsecured notes outstanding as of September 30, 2002 and December 31, 2001, is as follows (in thousands):

|  | September 30,<br>2002                               | December 31,<br>2001                      | Effective<br>Yield (1)                    |
|--|---|---|---|
| 8.375% Unsecured Notes due August 2005 | \$ 49,915<br>124,833<br>49,930<br>148,780<br>99,189 | \$ 49,892<br>124,809<br>49,923<br>148,670 | 8.44%<br>7.29%<br>8.66%<br>8.40%<br>6.90% |
| Total                                  | \$472,647   | \$373,294<br>=======                      |   |

#### (1) Including discount on the notes

On June 18, 2002, the OP completed a \$100 million debt offering consisting of \$100 million of 6.875% unsecured term notes due June 2012 that were priced to yield 6.90% to investors. Proceeds were used to repay borrowings under the Allen Premium Outlets and OPO construction loans that were to mature in February 2003 and August 2002, respectively, and for general corporate purposes.

### Notes to Condensed Consolidated Financial Statements (Unaudited)

#### 5. Debt (continued)

#### **Mortgage Debt**

A summary of the terms of the mortgage debt outstanding as of September 30, 2002 and December 31, 2001 and the related interest rate and Net Book Value ("NBV") of the associated collateral as of September 30, 2002 are as follows (in thousands):

|                        | September 30,<br>2002 | December 31,<br>2001 | Due Date      | Interest<br>Rate | NBV       |
|------------------------|-----------------------|----------------------|---------------|------------------|-----------|
| Construction Loan (1)  | \$ -                  | \$29,531             | -             | -                | \$ -      |
| F/C Mortgage Loan (2)  | 168,448               | -                    | July 2008     | 6.99%            | 257,589   |
| Bank Mortgage Loan (3) | 67,500                | 68,250               | April 2010    | 7.26%            | 72,662    |
| Mortgage Loan (4)      | 71,729                | 72,428               | December 2012 | 7.67%            | 75,711    |
|                        | \$307,677             | \$170,209            |               |                  | \$405,962 |

- (1) On June 18, 2002, the OP repaid and extinguished the loan.
- The F/C Mortgage Loan was consolidated as part of the August 20, 2002 buyout of Fortress' 51% interest in the F/C joint venture. The mortgage bears interest at 6.99% per annum through July 11, 2008 (the "Optional Prepayment Date") and thereafter at a rate equal to the greater of 8.4% plus 5% or the Treasury Rate, as defined, plus 6.5% until the earlier of the date the mortgage is paid in full or its maturity date of July 11, 2028. The mortgage may be prepaid in whole or in part at any time after the Optional Prepayment Date without a prepayment penalty. The mortgage calls for a \$1.2 million fixed monthly interest plus principal payment based on a 26-year amortization schedule. The F/C Mortgage Loan had a face value of \$169.6 million and a carrying amount of \$168.7 million on the buyout date. During the nine months ended September 30, 2002, the OP recognized \$10,000 in debt discount amortization that is included in interest expense in the accompanying financial statements.
- (3) In April 2000, Chelsea Financing entered into a \$70.0 million Bank Mortgage Loan secured by its four properties. The Bank Mortgage Loan bears interest equal to LIBOR plus 1.50% (3.32% at September 30, 2002) or prime rate plus 1.0% and calls for quarterly principal amortization of \$0.3 million through April 2005 and thereafter \$0.5 million per quarter until maturity. In December 2000, the OP entered into an interest rate swap agreement effective January 2, 2001 to hedge against unfavorable fluctuations in LIBOR rates by fixing the interest rate at 7.26% until January 2006. During the nine months ended September 30, 2002, the OP recognized interest expense of \$2.0 million on the hedge, that is included in interest expense in the accompanying financial statements.
- (4) The Mortgage Loan was assumed as part of a September 2001 acquisition. The stated interest rate of 9.1% was greater than that available to the OP in the public debt markets. Accordingly the OP recorded a \$6.9 million debt premium that will be amortized over the period of the loan and which reduces the effective interest rate to 7.67%. The loan calls for fixed monthly debt service payments of \$0.5 million for interest plus principal based on a 26-year amortization schedule. The Mortgage Loan matures in March 2028 but can be prepaid beginning December 2012. During the nine months ended September 30, 2002, the OP recognized \$0.3 million in debt premium amortization that is included in interest expense in the accompanying financial statements.

### Notes to Condensed Consolidated Financial Statements (Unaudited)

#### 5. Debt (continued)

Interest and loan costs of approximately \$0.8 million and \$0.4 million were capitalized as development costs during the three months ended September 30, 2002 and 2001, respectively; and approximately \$2.1 million and \$1.5 million during the nine months ended September 30, 2002 and 2001, respectively.

#### 6. Financial Instruments: Derivatives and Hedging

The OP employs interest rate and foreign currency forwards or purchased options to hedge qualifying anticipated transactions. Gains and losses are deferred and recognized in net income in the same period that the underlying hedged transaction affects net income, expires or is otherwise terminated or assigned.

At September 30, 2002 the OP's interest rate swap was reported at its fair value and classified as an other liability of \$6.9 million. As of September 30, 2002, there was \$6.9 million in deferred losses that was included in accumulated other comprehensive loss. Within the next twelve months, the OP expects to reclassify to earnings approximately \$2.1 million of the current balance.

| <u>Hedge Type</u> | Notional Value | <u>Rate</u> | <u>Maturity</u> | <u>Fair Value</u> |
|-------------------|----------------|-------------|-----------------|-------------------|
| Swap, Cash Flow   | \$67.5 million | 5.7625%     | 1/1/06          | (\$6.9 million)   |

At March 31, 2002, the OP had Japanese yen forward contracts with a notional value of 255 million yen and a fair value of \$10,000 as a hedge against its yen-denominated receivable due from Chelsea Japan. During the quarter ended June 30, 2002, the receivable and yen forward contracts were settled and the OP received \$1.9 million and recognized a \$0.1 million foreign exchange loss, which is included in income from unconsolidated investments in the accompanying financial statements.

The notional value and fair value of the above hedge provides an indication of the extent of the OP's involvement in financial derivative instruments at September 30, 2002, but do not represent exposure to credit, interest rate, foreign exchange or market risk.

### Notes to Condensed Consolidated Financial Statements (Unaudited)

#### 7. Partner's Capital

Following is a statement of partner's capital for the nine months ended September 30, 2002 (in thousands):

|   | General<br>Partner's<br>Capital | Limited<br>Partners'<br>Capital | Preferred<br>Partner's<br>Capital | Officer<br>Loan | Comp. Income (Loss (1) | Total<br>) Partners'<br>Capital |
|---|---------------------------------|---------------------------------|-----------------------------------|-----------------|------------------------|---------------------------------|
| Balance December 31, 2001                                     | \$362,402                       | \$52,825                        | \$63,315                          | \$ -            | (\$3,480)              | \$475,062                       |
| Net income  | 52,106                          | 12,607                          | -                                 | -               | -                      | 64,713                          |
| Foreign currency translation<br>Interest rate swap            | -<br>-                          | -                               | -                                 | -               | 218<br>(3,839)         | 218<br>(3,839)                  |
| Total comprehensive income                                    |                                 |                                 |                                   |                 |                        | 61,092                          |
| Common distributions  | (52,101)                        | (8,635)                         | -                                 | -               | -                      | (60,736)                        |
| Preferred distribution  | (2,587)                         | (4,387)                         | -                                 | -               | -                      | (6,974)                         |
| Contributions (net of costs)<br>Redemption of preferred units | 5,576                           | -                               | -                                 | -               | -                      | 5,576                           |
| (net of costs)  | (9,654)                         | -                               | -                                 | -               | -                      | (9,654)                         |
| Transfer of limited partners' interest                        | 239                             | (239)                           | -                                 | -               | -                      | - · · · · ·                     |
| Officer loan  | -                               |                                 | -                                 | (488)           | -                      | (488)                           |
| Balance September 30, 2002                                    | \$355,981                       | \$52,171                        | \$63,315                          | (\$488)         | (\$7,101)<br>======    | \$463,878<br>=======            |

(1) At September 30, 2002, accumulated other comprehensive loss included \$0.2 million related to foreign currency translation and \$6.9 million related to interest rate swap.

On May 1, 2002, the Company declared a 2-for-1 stock split of the Company's common shares. The OP simultaneously declared a 2-for-1 split of the OP's Limited Partner units. The stock/unit dividend was payable May 28, 2002 to shareholders/unitholders of record on May 14, 2002.

During the nine months ended September 30, 2002, the Company redeemed and retired 136,500 shares of 8.375% Series A Cumulative Redeemable Preferred Stock at \$47 per share and 66,552 shares at \$47.25 per share.

#### 8. Distribution

On September 18, 2002, the Board of Directors of the Company declared a \$0.485 per share dividend to unitholders of record on September 30, 2002. The distribution, totaling \$21.4 million, was paid on October 15, 2002.

### Notes to Condensed Consolidated Financial Statements (Unaudited)

#### 9. Income Taxes

No provision has been made for income taxes in the accompanying consolidated financial statements since such taxes, if any, are the responsibility of the individual partners.

#### 10. Net Income Per Partnership Unit

Net income per partnership unit is determined by allocating net income to the general partner (including the general partner's preferred unit allocation) and the limited partners based on their weighted average partnership units outstanding during the respective periods presented.

#### 11. Commitments and Contingencies

On May 15, 2002, the OP entered into a 50/50 joint venture agreement with Sordo Madaleno y Asociados and Mr. Carlos Peralta of Mexico City to jointly develop premium outlet centers in Mexico. Subject to entitlements, site work for the 200,000 square-foot first phase of the outlet project is scheduled to commence later in 2002, with construction to begin shortly thereafter. Phase I is expected to open in 2004. The site can support a second phase containing approximately 175,000 square feet of GLA. To date, the OP has committed \$3.0 million to the joint venture.

In connection with the Simon-Ventures, the OP has committed to provide 50% of the development costs of two centers, which are expected to be approximately \$48.0 million to Simon-Las Vegas and \$46.0 million to Simon-Chicago. As of September 30, 2002, the OP had contributed \$17.3 million to Simon-Las Vegas and \$7.1 million to Simon-Chicago.

In June 1999, the OP entered into an agreement with Mitsubishi Estate Co., Ltd. and Nissho Iwai Corporation to jointly develop, own and operate premium outlet centers in Japan. The joint venture is known as Chelsea Japan Co., Ltd. ("Chelsea Japan").

Borrowings related to Chelsea Japan for which the OP and the Company have provided guarantees as of September 30, 2002 are as follows (in thousands):

| Total F         | acility          |             | Ou               | itstanding      |          |               |  |
|-----------------|------------------|-------------|------------------|-----------------|----------|---------------|--|
| Yen             | US \$ Equivalent | ) Yen       | US \$ Equivalent | US \$ Guarantee | Due Date | Interest Rate |  |
|                 |                  |             |                  |                 |          |               |  |
| 4.0 billion (1) | \$32.8 million   | 1.0 billion | \$8.5 million    | \$8.5 million   | 2003     | 1.46%         |  |
| 3.8 billion (2) | 31.2 million     | 3.7 billion | 30.0 million     | 12.0 million    | 2015     | 2.20%         |  |
| 0.6 billion (2) | 4.9 million      | 0.6 billion | 4.9 million      | 2.0 million     | 2012     | 1.50%         |  |

- (1) Facility entered into by an equity investee of the OP and has a one-year extension option; amended in November 2002 to allow for one additional year extension.
- (2) Facilities entered into by Chelsea Japan, secured by Gotemba and Rinku and 40% severally guaranteed by the OP.

### Notes to Condensed Consolidated Financial Statements (Unaudited)

#### 11. Commitments and Contingencies (continued)

The OP agreed under a standby facility to provide up to \$22.0 million in limited debt service guarantees for loans provided to Value Retail and affiliates to construct outlet centers in Europe. The term of the standby facility expired in November 2001 and guarantees shall not be outstanding for longer than five years after project completion. As of September 30, 2002, the OP had provided limited debt service guarantees of approximately \$18.0 million to Value Retail.

At September 30, 2002, other assets includes \$6.9 million and accrued expenses and other liabilities include \$8.0 million related to the 2002 deferred unit incentive program with certain key officers which may be paid in 2007.

The OP is not presently involved in any material litigation nor, to its knowledge, is any material litigation threatened against the OP or its properties, other than routine litigation arising in the ordinary course of business. Management believes the cost incurred by the OP related to any of its litigation will not be material and have been adequately provided for in the consolidated financial statements.

#### 12. Related Party Information

In 1999, the OP established a \$6.0 million secured loan facility that will expire in June 2004 for the benefit of certain unitholders. Each unit holder borrower issues a note that is secured by OP units, bears interest at a rate of LIBOR plus 200 basis points per annum, payable quarterly and is due by the facility expiration date. At December 31, 2001, loans made to unitholders or affiliates of management of the OP, totaled \$3.2 million. During the nine months ended September 30, 2002, the OP received a \$0.5 million payoff of one loan and a \$0.6 million pay down on the other loan outstanding. During March and May of 2002, the OP advanced a total of \$1.0 million to another unitholder \$0.5 million of which was used to exercise Company stock options. The carrying value

of such loans approximated fair value at September 30, 2002. Effective June 1, 2002, the OP changed its policy to eliminate new loans to directors or officers.

#### 13. Segment Information

The OP is principally engaged in the development, ownership, acquisition and operation of manufacturers' outlet centers and has determined that under SFAS No.131 "Disclosures About Segments of an Enterprise and Related Information" it has three reportable retail real estate segments: premium domestic, other domestic and international. The OP evaluates real estate performance and allocates resources based on Net Operating Income ("NOI") defined as total revenue less operating and maintenance expense. The primary sources of revenue are generated from tenant base rents, percentage rents and reimbursement revenue. Operating expenses primarily consist of common area maintenance, real estate taxes and promotional expenses. The retail real estate business segments meet the quantitative threshold for determining reportable segments.

### Notes to Condensed Consolidated Financial Statements (Unaudited)

#### 13. Segment Information (continued)

| (in thousands)             | Premium<br>Domestic | Other<br>Domestic | International | Other    | Total     |
|----------------------------|---------------------|-------------------|---------------|----------|-----------|
|                            |                     | (1)               | (2)           | (3)      |           |
| Total revenues             |                     | ` ,               | ` ,           | ` ,      |           |
| Three months ended:        |                     |                   |               |          |           |
| September 30, 2002         | \$58,302            | \$11,805          | \$ -          | \$1,382  | \$71,489  |
| September 30, 2001         | 46,696              | 713               | -             | 1,656    | 49,065    |
| Nine months ended:         |                     |                   |               |          |           |
| September 30, 2002         | 153,489             | 35,589            | -             | 4,122    | 193,200   |
| September 30, 2001         | 134,159             | 713               | -             | 5,314    | 140,186   |
| Interest income            |                     |                   |               |          |           |
| Three months ended:        |                     |                   |               |          |           |
| September 30, 2002         | 252                 | 76                | -             | 105      | 433       |
| September 30, 2001         | 204                 | -                 | -             | 372      | 576       |
| Nine months ended:         |                     |                   |               |          |           |
| September 30, 2002         | 792                 | 140               | -             | 276      | 1,208     |
| September 30, 2001         | 617                 | -                 | -             | 1,451    | 2,068     |
| Income from unconsolidated |                     |                   |               |          |           |
| investments                |                     |                   |               |          |           |
| Three months ended:        |                     |                   |               |          |           |
| September 30, 2002         | 971                 | -                 | 1,061         | (3,790)  | (1,758)   |
| September 30, 2001         | 3,176               | -                 | 777           | (1,477)  | 2,476     |
| Nine months ended:         |                     |                   |               |          |           |
| September 30, 2002         | 5,643               | -                 | 3,141         | (10,266) | (1,482)   |
| September 30, 2001         | 8,147               | -                 | 2,362         | (3,447)  | 7,062     |
| NOI                        |                     |                   |               |          |           |
| Three months ended:        |                     |                   |               |          |           |
| September 30, 2002         | 45,488              | 10,350            | 2,064         | (6,546)  | 51,356    |
| September 30, 2001         | 37,743              | 548               | 1,657         | 633      | 40,581    |
| Nine months ended:         |                     |                   |               |          |           |
| September 30, 2002         | 122,527             | 21,297            | 5,915         | (9,109)  | 140,630   |
| September 30, 2001         | 108,010             | 548               | 4,601         | (403)    | 112,756   |
| Fixed asset additions      |                     |                   |               |          |           |
| Nine months ended:         |                     |                   |               |          |           |
| September 30, 2002         | 103,520             | 28,550            | -             | 1,432    | 133,502   |
| September 30, 2001         | 46,849              | 41,451            | -             | 1,272    | 89,572    |
| Total assets               |                     |                   |               |          |           |
| September 30, 2002         | 1,179,977           | 214,542           | 14,307        | 21,858   | 1,430,684 |
| December 31, 2001          | 866,738             | 194,406           | 15,092        | 23,072   | 1,099,308 |

- (1) Approximately 25% of the GLA is occupied by and approximately 13% of annualized revenue is derived from one tenant.
- (2) Principally comprised of the OP's interest in Chelsea Japan.
- (3) Includes corporate overhead assets and results from Chelsea Interactive.

### Notes to Condensed Consolidated Financial Statements (Unaudited)

#### 13. Segment Information (continued)

Following is a reconciliation of net operating income to net income for the three and nine months ended September 30, 2002 and 2001 (in thousands):

|  | Three Months Ended<br>September 30,       |  | Nine Months<br>September                   | 30,  |
|--|---|--|--|--|
|  | 2002                                      | 2001                                     | 2002                                       | 2001                                       |
| Segment NOI  | \$51,356<br>(13,098)<br>(146)<br>(15,045) | \$40,581<br>(8,891)<br>(145)<br>(12,174) | \$140,630<br>(33,691)<br>(413)<br>(42,229) | \$112,756<br>(26,326)<br>(445)<br>(35,479) |
| Depreciation and amortization  | (15,045)                                  | (12,174)                                 | (42,229)                                   | (35,479)                                   |
| expense - unconsolidated investments  Depreciation and amortization    | (871)                                     | (1,628)                                  | (3,673)                                    | (4,700)                                    |
| expense - Chelsea Interactive  Income tax - unconsolidated investments | (2,196)<br>(386)                          | (897)<br>(244)                           | (5,764)<br>(1,058)                         | (1,950)<br>(684)                           |

| Gain on sale of | asset - | unconsolidated | investments | 10,911   | -        | 10,911   | -        |
|-----------------|---------|----------------|-------------|----------|----------|----------|----------|
| Net income      |         |                |             | \$30,525 | \$16,602 | \$64,713 | \$43,172 |
|                 |         |                |             |          |          |          |          |

#### 14. Subsequent Event

On November 12, 2002, the OP announced that it had signed a definitive agreement to acquire four outlet centers from New Plan Excel Realty Trust, Inc. for an all cash aggregate price of \$193.0 million. The four properties, totaling 1.3 million square feet of GLA, are a 293,000 square-foot center located in Jackson, New Jersey; a 400,000 square-foot center located in Osage Beach, Missouri; a 329,000 square-foot center located in St. Augustine, Florida; and a 317,000 square-foot center located in Branson, Missouri. Subject to certain conditions, the transaction is expected to close on or about December 31, 2002 but there can be no assurance that the transaction will close.

#### CPG Partners, L.P.

#### Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations

The following discussion should be read in connection with the accompanying unaudited condensed consolidated financial statements and notes thereto. These financial statements include all adjustments, which in the opinion of management are necessary to reflect a fair statement of results for all interim periods presented, and all such adjustments are of a normal recurring nature.

#### **General Overview**

From October 1, 2001 to September 30, 2002, the OP grew by increasing rents at its operating centers, expanding three centers, acquiring 32 retail centers and purchasing ownership interests in five centers previously held as unconsolidated investments. Increasing rents at operating centers resulted in base rent growth of \$2.5 million. The OP re-leased or renewed approximately 1.0 million square feet of Premium Property GLA during the twelve months ended September 30, 2002, for which initial contractual cash rents under new leases were 12% higher than expiring leases. The expansion of two wholly-owned centers increased base rent by \$1.4 million. The acquisition of 32 retail centers increased base rent by \$26.4 million. The purchase of ownership interests in the five centers previously held as unconsolidated investments resulted in base rent growth of \$9.3 million and a decrease in income from unconsolidated investments of \$1.7 million during the nine months ended September 30, 2002. Income from unconsolidated investments was positively impacted by the results of Chelsea Japan due to higher earnings, increased fees and the opening of the expansion of Rinku Premium Outlets during March 2002.

The OP operated GLA of 12.5 million square feet at September 30, 2002 and 2001. The OP's Premium Properties consisted of 27 wholly or partially owned centers containing 8.4 million square feet of GLA at September 30, 2002, and 28 wholly or partially owned centers containing 8.2 million square feet of GLA at September 30, 2001. The OP's wholly-owned Other Properties consisted of 27 centers containing 4.1 million square feet of GLA at September 30, 2002, and 30 wholly-owned centers containing 4.3 million square feet of GLA at September 30, 2001.

Net GLA added since October 1, 2001 is detailed as follows:

|   | 12 Months Ended<br>September 30, 2002 | 9 Months Ended<br>September 30, 2002 | 3 Months Ended<br>December 31, 2001 |
|---|---------------------------------------|--------------------------------------|-------------------------------------|
| Changes in GLA (sf in 000's):   |                                       |                                      |                                     |
| Centers expanded: Rinku Premium Outlets (40% owned) Napa Premium OutletsAllen Premium Outlets Other | . 9<br>. 115                          | 70<br>9<br>(4)<br>(4)                | -<br>-<br>119<br>(3)                |
| Total centers expanded  | . 187                                 | 71                                   | 116                                 |
| Centers acquired: Edinburgh Outlets   | . 305                                 | 305                                  | -                                   |
| Centers sold: Other Properties Mammoth Premium Outlets  |                                       | (433)                                | -<br>(35)                           |
| Total centers sold  | . (468)                               | (433)                                | (35)                                |
| Net GLA change during the period  | . 24                                  | (57)                                 | 81                                  |
| GLA at end of period  | . 12,517                              | 12,517                               | 12,574                              |

#### **Results of Operations**

Comparison of the three months ended September 30, 2002 to the three months ended September 30, 2001.

Income increased \$13.9 million, or 83.9%, to \$30.5 million for the three months ended September 30, 2002, from \$16.6 million for the three months ended September 30, 2001. This increase was primarily the result of the acquisition of the 31 retail centers on September 25, 2001, the acquisition of one center in April 2002, the buyout and consolidation of five centers previously held as unconsolidated investments, a gain resulting from the partial sale of an unconsolidated investment, and higher rents on releasing and renewals during 2002, partially offset by higher interest costs and increased losses from Chelsea Interactive.

Base rentals increased \$16.4 million, or 53.9%, to \$46.9 million for the three months ended September 30, 2002, from \$30.5 million for the three months ended September 30, 2001, due to the acquisition of the 32 retail centers and the buyouts of ownership interests in five centers in 2002, higher average rents on releasing and renewals and the expansion of a center in late 2001.

Percentage rents increased \$1.3 million, or 33.3%, to \$5.4 million for the three months ended September 30, 2002, from \$4.1 million for the three months ended September 30, 2001, primarily due to higher tenant sales, the acquisition of the 32 retail centers and the buyouts of ownership interests in five centers during 2002.

Expense reimbursements, representing contractual recoveries from tenants of certain common area maintenance, operating, real estate tax, promotional and management expenses, increased \$4.5 million, or 39.3%, to \$16.1 million for the three months ended September 30, 2002, from \$11.6 million for the three months ended September 30, 2001, due to the recovery of operating and maintenance costs from increased GLA. The average recovery of reimbursable expenses for the Premium Properties was 89.9% for the three months ended September 30, 2002 compared to 90.6% for the three months ended September 30, 2001. The average recovery of reimbursable expenses for the Other Properties was 57.7% for the three months ended September 30, 2002.

Other income increased \$0.1 million, to \$3.0 million for the three months ended September 30, 2002, from \$2.9 million for the three months ended September 30, 2001, primarily due to increased ancillary operating income from the acquisition of the 32 new retail centers and the buyouts of ownership interests in five centers in 2002 partially offset by decreased interest income from lower interest rates and the sale of an outparcel during 2001.

Operating and maintenance expenses increased \$6.6 million, or 51.7%, to \$19.4 million for the three months ended September 30, 2002, from \$12.8 million for the three months ended September 30, 2001. The increase was primarily due to costs related to increased GLA and the buyouts of ownership interests in five centers during 2002.

Depreciation and amortization expense increased \$2.8 million, or 23.6%, to \$15.0 million for the three months ended September 30, 2002, from \$12.2 million for the three months ended September 30, 2001. The increase was due to additional depreciation of the acquisition of the 32 retail centers, the buyouts of ownership interests during 2002 and expansions.

General and administrative expense increased \$1.0 million, to \$1.6 million for the three months ended September 30, 2002, from \$0.6 million for the three months ended September 30, 2001, primarily due to increased unit program accrual and professional fees.

Other expenses increased \$0.5 million, to \$1.0 million for the three months ended September 30, 2002, from \$0.5 million for the three months ended September 30, 2001, due to ground lease expenses assumed with the acquisition of the 31 retail centers, legal expenses and increased reserve for bad debt.

Income from unconsolidated investments decreased \$2.0 million, to \$2.0 million for the three months ended September 30, 2002, from \$4.0 million for the three months ended September 30, 2001, due to the buyouts of ownership interests in five centers which were previously held as unconsolidated investments partially offset by higher earnings from Chelsea Japan.

The loss from Chelsea Interactive increased \$2.3 million, or 156.6%, to \$3.8 million for the three months ended September 30, 2002, from \$1.5 million for the three months ended September 30, 2001. The increase was due to increased payroll, general and administrative, depreciation and amortization expense and lack of third party participation in the losses.

Gain on sale of unconsolidated investment of \$10.9 million for the three months ended September 30, 2002 resulted from the sale of approximately 40% of the OP's partial interest in Value Retail.

Interest in excess of amounts capitalized increased \$4.2 million, or 47.3%, to \$13.1 million for the three months ended September 30, 2002, from \$8.9 million for the three months ended September 30, 2001, due to higher debt balances from acquisitions, debt assumption in the F/C buyout and bond issue partially offset by lower interest rates.

Comparison of the nine months ended September 30, 2002 to the nine months ended September 30, 2001.

Income increased \$21.5 million, or 49.9%, to \$64.7 million for the nine months ended September 30, 2002, from \$43.2 million for the nine months ended September 30, 2001. This increase was primarily the result of the acquisition of the 31 retail centers on September 25, 2001, the acquisition of one center in April 2002, the buyout and consolidation of five centers previously held as unconsolidated investments, a gain resulting from the partial sale of an unconsolidated investment, and higher rents on releasing and renewals during 2002, partially offset by higher interest costs and increased losses from Chelsea Interactive.

Base rentals increased \$40.9 million, or 46.5%, to \$128.8 million for the nine months ended September 30, 2002, from \$87.9 million for the nine months ended September 30, 2001, due to the acquisition of the 32 retail centers and the buyouts of ownership interests in five centers in 2002 and higher average rents on new leases and renewals.

Percentage rents increased \$3.1 million or 31.5% to \$13.0 million for the nine months ended September 30, 2002, from \$9.9 million for the nine months ended September 30, 2001, primarily due to higher tenant sales, acquisition of the 32 retail centers and the buyouts of ownership interests in five centers during 2002.

Expense reimbursements, representing contractual recoveries from tenants of certain common area maintenance, operating, real estate tax, promotional and management expenses, increased \$9.3 million, or 27.2%, to \$43.3 million for the nine months ended September 30, 2002, from \$34.0 million for the nine months ended September 30, 2001, due to the recovery of operating and

maintenance costs from increased GLA. The average recovery of reimbursable expenses for the Premium Properties was 88.4% for the nine months ended September 30, 2002, compared to 90.7% for the nine months ended September 30, 2001. The average recovery of reimbursable expense for the Other Properties was 55.1% for the nine months ended September 30, 2002.

Other income decreased \$0.2 million, to \$8.2 million for the nine months ended September 30, 2002, from \$8.4 million for the nine months ended September 30, 2001. The decrease is primarily the result of decreased interest income from lower interest rates and the sale of an outparcel in 2001 partially offset by an increase in ancillary income related to the acquisition of the 32 new retail centers and the buyouts of ownership interests in five centers during 2002.

Operating and maintenance expenses increased \$16.3 million, or 43.4%, to \$53.8 million for the nine months ended September 30, 2002, from \$37.5 million for the nine months ended September 30, 2001. The increase was primarily due to costs related to increased GLA, and the buyouts of ownership interests in five centers during 2002.

Depreciation and amortization expense increased \$6.7 million, or 19.0%, to \$42.2 million for the nine months ended September 30, 2002, from \$35.5 million for the nine months ended September 30, 2001. The increase was due to additional depreciation of the acquisition of the 32 retail centers and the buyouts of ownership interests in five centers during 2002 and expansions.

General and administrative expense increased \$1.9 million, or 63.6%, to \$5.0 million for the nine months ended September 30, 2002, from \$3.1 million for the nine months ended September 30, 2001, primarily due to increased head count, accrual for unit program and professional fees.

Other expenses increased \$1.5 million, or 88.0%, to \$3.2 million for the nine months ended September 30, 2002, from \$1.7 million for the nine months ended September 30, 2001. The increase was primarily due to ground lease expenses assumed with the acquisition of the 31 retail centers, legal expenses and increased reserve for bad debt.

Income from unconsolidated investments decreased \$1.7 million, to \$8.8 million for the nine months ended September 30, 2002, from \$10.5 million for the nine months ended September 30, 2001, due to the buyouts of ownership interests in five centers which were previously held as unconsolidated investments, offset by increased earnings from Chelsea Japan, including a one time leasing fee of \$0.2 million from Rinku phase two.

The loss from Chelsea Interactive increased \$6.8 million, or 197.8%, to \$10.2 million for the nine months ended September 30, 2002, from \$3.4 million for the nine months ended September 30, 2001, due to increased depreciation, administrative and maintenance expenses and lack of third party participation in the losses.

Gain on sale of unconsolidated investment of \$10.9 million for the nine months ended September 30, 2002 resulted from the sale of approximately 40% of the OP's interest in Value Retail.

Interest in excess of amounts capitalized increased \$7.4 million, or 28.0%, to \$33.7 million for the nine months ended September 30, 2002, from \$26.3 million for the nine months ended September 30, 2001, primarily due to higher debt balances from acquisitions, debt assumption in the F/C buyout, lower construction activity and new bond issue partially offset by lower interest rates.

#### **Liquidity and Capital Resources**

The OP believes it has adequate financial resources to fund operating expenses, distributions, and planned development, construction and acquisition activities over the short term, which is less than 12 months and the long term, which is 12 months or more. Operating cash flow in 2001 of \$121.7 million is expected to increase with a full year of operations of the 4.4 million square feet of GLA added during 2001, the acquisition of one center, the buyouts of ownership interests in five centers and scheduled openings of approximately 127,000 square feet of GLA in 2002. The OP has adequate funding sources to complete and open all planned acquisition or development projects, fund committed amounts for Chelsea Interactive and the Simon-Ventures and to acquire six centers for \$282.5 million that are expected to close during the fourth quarter 2002. These transactions are expected to be accomplished with available cash of \$13.7 million and \$108 million available under the increased Senior Credit Facility, additional bank borrowings, issuing OP units or accessing the public markets through the Company's \$350 million debt shelf registration and its \$185 million equity shelf registration.

Operating cash flow is expected to provide sufficient funds for dividends and distributions in accordance with REIT federal income tax requirements. In addition, the OP anticipates retaining sufficient operating cash to fund re-tenanting and lease renewal tenant improvement costs, as well as capital expenditures to maintain the quality of its centers, and meet funding requirements for Chelsea Interactive.

Common distributions declared and recorded during the nine months ended September 30, 2002, were \$60.7 million, or \$1.38 per share or unit. The OP's common dividend payout ratio as a percentage of net income before depreciation and amortization (exclusive of amortization of deferred financing costs ("FFO")) was 66.8% for the nine months ended September 30, 2002. The Senior Credit Facility limits aggregate dividends and distributions to the lesser of (i) 90% of FFO on an annual basis or (ii) 100% of FFO for any two consecutive quarters.

At September 30, 2002, the OP had a \$200.0 million senior unsecured revolving bank line of credit, (the "Senior Credit Facility"). Under the terms, the Senior Credit Facility's expiration is March 31, 2005, and the OP has the right to extend the maturity date until March 31, 2006. The Senior Credit Facility bears interest on the outstanding balance, payable monthly, at a rate equal to the

London Interbank Offered Rate ("LIBOR") plus 1.05% (2.88% at September 30, 2002) or the prime rate, at the OP's option. The LIBOR rate spread ranges from 0.85% to 1.50% depending on the OP's Senior Debt rating. A facility fee of 0.125% on the Senior Credit Facility is payable annually in advance. \$108.0 million of the Senior Credit Facility was available at September 30, 2002.

On August 20, 2002, the OP became the sole owner of the four premium outlets held in the F/C joint venture by acquiring Fortress' 51% undivided ownership interest for \$58.9 million in cash which was funded by borrowing from the Senior Credit Facility.

On June 18, 2002, the OP completed a \$100 million debt offering consisting of \$100 million of 6.875% unsecured term notes due June 2012 that were priced to yield 6.90% to investors. Proceeds were used to repay borrowings under the Allen Premium Outlets and OPO construction loans that were to mature in February 2003 and August 2002, respectively, and for general corporate purposes.

On April 1, 2002, the OP acquired a 305,000 square foot center located in Edinburgh, Indiana for \$27.0 million in cash. The center is located 40 miles south of downtown Indianapolis and was 98% leased at the acquisition date. The OP funded the transaction from available cash and the Senior Credit Facility. Also on April 1, 2002, the OP became the sole owner of the Orlando Premium Outlets by acquiring Simon's 50% undivided ownership interest for \$46.6 million in cash and the assumption of \$29.7 million of existing mortgage debt and the guarantee related thereto. The cash portion of the transaction was funded by borrowing from the Senior Credit Facility. On June 16, 2002, the OP repaid the outstanding balance of \$59.4 million and extinguished the mortgage.

During 2002 the Company redeemed and retired 136,500 shares of 8.375% Series A Cumulative Redeemable Preferred Stock at \$47 per share and 66,552 shares at \$47.25 per share using available cash.

During June and July 2002, the OP sold four Other Properties and two outparcels generating net proceeds of approximately \$6.8 million, which approximated the net book value of the properties. Accordingly, no gain or loss on these sales was recognized.

Development activity for real estate operations as of September 30, 2002 includes additional phases of two existing centers, totaling 47,000 square feet scheduled to open in late 2002. Construction is also underway on the 50% joint venture project outside Las Vegas, Nevada scheduled to open in summer 2003, the 175,000 square-foot second phase of Gotemba Premium Outlets scheduled to open mid–2003, and the 50% joint venture project in Chicago, Illinois, scheduled to open in mid 2004; development is underway on a project in Seattle, Washington scheduled to open in 2005 and a project north of Tokyo, Japan scheduled to open in mid-2003. These projects are under development and there can be no assurance that they will be completed or opened, or that there will not be delays in opening or completion. All current development activity will be financed either through project specific secured construction financing, the yen denominated line of credit, available cash or through the Senior Credit Facility. The OP will seek to obtain permanent financing once the projects are completed and income has been stabilized.

On May 15, 2002, the OP entered into a 50/50 joint venture agreement with Sordo Madaleno y Asociados and Mr. Carlos Peralta of Mexico City to jointly develop premium outlet centers in Mexico. Subject to entitlements, site work for the 200,000 square-foot first phase of the outlet project is scheduled to commence later in the year, with construction to begin shortly thereafter. Phase I is expected to open in 2004. The site can support a second phase containing approximately 175,000 square feet of GLA. To date, the OP has committed \$3.0 million to the joint venture.

In connection with the Simon-Ventures, the OP has committed to provide 50% of the development costs of the centers, which are expected to be approximately \$48.0 million to Simon-Las Vegas and \$46.0 million to Simon-Chicago. As of September 30, 2002, the OP had contributed \$17.3 million to Simon-Las Vegas and \$7.1 million to Simon-Chicago.

The OP has minority interests ranging from 3% to 8% in several outlet centers and outlet development projects in Europe operated by Value Retail. The OP's total investment in Europe as of September 30, 2002 was \$3.6 million. The OP has also agreed to provide up to \$22.0 million in limited debt service guarantees under a standby facility for loans arranged by Value Retail to construct outlet centers in Europe. The term of the standby facility for new guarantees expired in November 2001 and these guarantees shall not be outstanding for longer than five years after project completion. In July 2002, the OP sold approximately 40% of its holdings in Value Retail to a third party for \$11.4 million that resulted in a gain of approximately \$10.9 million.

In June 1999, the OP entered into an agreement with Mitsubishi Estate Co., Ltd. and Nissho Iwai Corporation to jointly develop, own and operate premium outlet centers in Japan. The joint venture is known as Chelsea Japan Co., Ltd. ("Chelsea Japan").

Borrowings related to Chelsea Japan for which the OP and the Company have provided guarantees as of September 30, 2002, are as follows (in thousands):

| Total Facilit   | у                |             | Outstanding      |                 |          |               |
|-----------------|------------------|-------------|------------------|-----------------|----------|---------------|
| Yen             | US \$ Equivalent | <br>  Yen   | US \$ Equivalent | US \$ Guarantee | Due Date | Interest Rate |
|                 |                  |             |                  |                 |          |               |
| 4.0 billion (1) | \$32.8 million   | 1.0 billion | \$8.5 million    | \$8.5 million   | 2003     | 1.46%         |
| 3.8 billion (2) | 31.2 million     | 3.7 billion | 30.0 million     | 12.0 million    | 2015     | 2.20%         |
| 0.6 billion (2) | 4.9 million      | 0.6 billion | 4.9 million      | 2.0 million     | 2012     | 1.50%         |

- (1) Facility entered into by an equity investee of the OP and has a one-year extension option; amended in November 2002 to allow for one additional year extension.
- (2) Facilities entered into by Chelsea Japan, secured by Gotemba and Rinku and 40% severally guaranteed by the OP.

The OP has developed an e-commerce venture through its affiliate, Chelsea Interactive. The Board of Directors has approved a funding commitment of up to \$60.0 million. As of September 30, 2002, the OP has funded \$51.6 million and anticipates that the balance of the funding will be used for development and operating cash shortfalls. The net book value of the OP's investment as of September 30, 2002 was \$35.0 million. While management currently believes that it is possible to recover the net book value of the investment through future cash flows if revenues from existing brands improve and/or additional brands are secured, there is substantial risk that Chelsea Interactive will not: (1) generate sufficient future revenues for the OP to recover its investment; (2) achieve positive cash flow before reaching the OP's committed funding of \$60.0 million; (3) successfully complete a satisfactory arrangement with a financial/technology investor; or (4) continue as a going concern. As a result of these risk factors, management may at any time determine that it is necessary to significantly write down or write off the OP's \$35.0 million investment. The OP is actively pursuing and negotiating with potential investors to provide capital and further develop the business.

To achieve planned growth and favorable returns in both the short and long-term, the OP's financing strategy is to maintain a strong, flexible financial position by: (i) maintaining a conservative level of leverage; (ii) extending and sequencing debt maturity dates; (iii) managing exposure to floating interest rates; and (iv) maintaining liquidity. Management believes these strategies will enable the OP to access a broad array of capital sources, including bank or institutional borrowings and secured and unsecured debt and equity offerings, subject to market conditions.

The OP's cash balances decreased \$10.9 million to \$13.7 million at September 30, 2002 from \$24.6 million at December 31, 2001. Net cash provided by operating activities increased \$2.5 million for the nine months ended September 30, 2002, compared to the corresponding 2001 period. The increase was primarily due to the increased operating cash flow generated from the growth of the OP's GLA substantially offset by the payout of the deferred incentive compensation in March 2002. Net cash used in investing activities increased \$33.7 million for the nine months ended September 30, 2002, compared to the corresponding 2001 period. The increase was primarily due to the acquisition of one center and the buyout of ownership interests in five centers partially offset by proceeds from sale of an unconsolidated investment. Net cash provided by financing activities increased by \$22.0 million compared to the corresponding period in 2001. The increase was primarily due to increased debt issuance, partially offset by decreased offerings of the Company's common stock and retirements of the Company's preferred stock.

#### **Funds from Operations**

Management believes that FFO should be considered in conjunction with net income, as presented in the statements of operations included elsewhere herein, to facilitate a clearer understanding of the operating results of the OP. The White Paper on FFO approved by the Board of Governors of NAREIT in October 1999 defines FFO as net income (loss) (computed in accordance with GAAP), excluding gains (or losses) from debt restructuring and sales of properties, plus real estate related depreciation and amortization and after adjustments for unconsolidated partnerships and joint ventures. The OP believes that FFO is helpful to investors as a measure of the performance of an equity REIT because, along with cash flow from operating activities, financing activities and investing activities, it provides investors with an indication of the ability of the OP to incur and service debt, to make capital expenditures and to fund other cash needs. The OP computes FFO in accordance with the current standards established by NAREIT which may not be comparable to FFO reported by other REITs that do not define the term in accordance with the current NAREIT definition or that interpret the current NAREIT definition differently than the OP. FFO does not represent cash generated from operating activities in accordance with GAAP and should not be considered as an alternative to net income (determined in accordance with GAAP) as an indication of the OP's financial performance or to cash flow from operating activities (determined in accordance with GAAP) as a measure of the OP's liquidity, nor is it indicative of funds available to fund the OP's cash needs, including its ability to make cash distributions.

|  | Three Months Ended<br>September 30, |                  | Nine Month<br>Septembe |                  |
|--|-------------------------------------|------------------|------------------------|------------------|
|  | 2002                                | 2001             | 2002                   | 2001             |
| Net income to common unitholdersAdd (deduct):  | \$28,228                            | \$14,093         | \$57,739               | \$35,645         |
| Depreciation and amortization - wholly owned   | 15,045                              | 12,174           | 42,229                 | 35,479           |
| Depreciation and amortization - joint ventures  Amortization of deferred financing costs and | 871                                 | 1,628            | 3,673                  | 4,700            |
| depreciation of non-rental real estate assets  | (573)                               | (480)            | (1,741)                | (1,326)          |
| Gain on sale of unconsolidated investment  | (10,911)                            | ` - ′            | (10,911)               | ` ' - '          |
| FF0  | \$32,660                            | \$27,415         | \$90,989               | \$74,498         |
| Average units outstanding (1)  | 45,883<br>\$0.485                   | 40,630<br>\$0.39 | 45,538<br>\$1.375      | 39,912<br>\$1.17 |

(1) Assumes May 28, 2002, 2-for-1 stock/unit split had occurred on January 1, 2001.

#### **Recent Accounting Pronouncements**

In October 2001, the FASB issued Statement of Financial Accounting Standard No. 144, "Accounting for the Impairment or Disposal of Long-Lived Assets", which addresses financial accounting and reporting for the impairment or disposal of long-lived assets. This standard harmonizes the accounting for impaired assets and resolves some of the implementation issues as originally described in SFAS 121. The new standard became effective and was adopted by the OP on January 1, 2002. The adoption of this pronouncement has not had a material impact on the OP's results of operations or financial position.

In May 2002, the FASB issued SFAS No. 145, "Reporting Gains and Losses from Extinguishment of Debt", which rescinded SFAS No. 4, No. 44 and No. 64 and amended SFAS No. 13. The new standard addresses the income statement classification of gains or

losses from the extinguishment of debt and criteria for classification as extraordinary items. The new standard became effective for fiscal years beginning after May 15, 2002. The OP adopted this pronouncement on April 1, 2002. The adoption of this pronouncement did not have a material impact on the OP's results of operations or financial position.

#### **Critical Accounting Policies and Estimates**

The OP's discussion and analysis of its financial condition and results of operations are based upon the OP's consolidated financial statements which have been prepared in accordance with accounting principles generally accepted in the United States. The preparation of these financial statements requires the OP to make estimates and judgments that affect the reported amounts of assets, liabilities, revenues and expenses. The OP bases its estimates on historical experience and on various other assumptions that are believed to be reasonable under the circumstances, the results of which form the basis for making judgments about the carrying values of assets and liabilities that are not readily apparent from other sources. Actual results may differ from these estimates under different assumptions or conditions.

The OP believes the following critical accounting policies affect its more significant judgments and estimates used in the preparation of its consolidated financial statements.

#### **Bad Debt**

The OP maintains an allowance for doubtful accounts for estimated losses resulting from the inability of its tenants to make required rent payments. If the financial condition of the OP's tenants were to deteriorate, resulting in an impairment of their ability to make payments, additional allowances may be required.

#### Foreign Exchange Exposure

From time to time, the OP may enter into hedging contracts to protect against fluctuations in foreign exchange rates. Receivables from Chelsea Japan are recorded in the accompanying financial statements at the exchange rate on the date of valuation. Significant changes in the exchange rate may result in changes to the receivable realized.

#### **Valuation of Investments**

The OP reviews its investments in unconsolidated affiliates for impairment wherever events or changes in circumstances indicate that the carrying value of the asset may not be recoverable.

#### **Economic Conditions**

Substantially all leases contain provisions, including escalations of base rents and percentage rentals calculated on gross sales, to mitigate the impact of inflation. Inflationary increases in common area maintenance and real estate tax expenses are substantially all reimbursed by tenants.

Virtually all tenants have met their lease obligations and the OP continues to attract and retain quality tenants. The OP intends to reduce operating and leasing risks by continually improving its tenant mix, rental rates and lease terms, and by pursuing contracts with creditworthy upscale and national brand-name tenants.

#### Item 3. Quantitative and Qualitative Disclosures about Market Risk

The OP is exposed to changes in interest rates primarily from its floating rate debt arrangements. In December 2000, the OP implemented a policy to protect against interest rate and foreign exchange risk. The OP's primary strategy is to protect against this risk by using derivative transactions as appropriate to minimize the variability that floating rate interest and foreign currency fluctuations could have on cash flow. In December 2000, a wholly-owned subsidiary of the OP entered into an interest rate swap agreement effective January 2, 2001 with a financial institution for a notional amount of \$69.3 million amortizing to \$64.1 million to hedge against unfavorable fluctuations in the LIBOR rates of its secured mortgage loan facility. The hedge effectively produces a fixed interest rate of 7.2625% on the notional amount until January 1, 2006.

At September 30, 2002 a hypothetical 100 basis point adverse move (increase) in U.S. Treasury and LIBOR rates applied to unhedged debt would adversely affect the OP's annual interest cost by approximately \$1.0 million annually.

Following is a summary of the OP's debt obligations at September 30, 2002 (in thousands):

|                       |      | Expected Maturity Date |      |        |      |            |         |            |  |
|-----------------------|------|------------------------|------|--------|------|------------|---------|------------|--|
|                       | 2002 | 2003                   | 2004 | 2005   | 2006 | Thereafter | Total   | Fair Value |  |
|                       |      |                        |      |        |      |            |         |            |  |
| Fixed Rate Debt       | -    | -                      | -    | 49,915 | -    | 662,909    | 712,824 | 727,719    |  |
| Average Interest Rate | -    | -                      | -    | 8.38%  | -    | 7.50%      | 7.56%   | -          |  |
| Variable Rate Debt    | -    | -                      | -    | 97,035 | -    | 67,500     | 164,535 | 164,535    |  |
| Average Interest Rate | -    | _                      | _    | 2.88%  | -    | 3.32%      | 3.06%   |            |  |

#### **Item 4. Controls and Procedures**

The Company's chief executive officer and chief financial officer, based on their evaluation of the OP's disclosure controls and procedures within the past 90 days, have concluded that such disclosure controls and procedures were effective to ensure that

material information relating to the OP and required to be disclosed by the OP has been made known to them and has been recorded, processed, summarized and reported timely. There have not been any significant changes in the OP's internal controls or in other factors that could significantly affect these controls subsequent to the date of evaluation.

#### CPG Partners, L.P.

#### Part II. Other Information

#### Item 6. Exhibits and Reports on Form 8-K

(a)

- Amended and Restated Credit Agreement dated July 31, 2002 among CPG Partners, L.P. and Fleet National Bank, individually and as an agent, and other Lending Institutions listed therein.
- (b) Current Report on Form 8K and 8K/A reporting under Items 2 and 7 on an event that occurred on August 20, 2002.

**Signatures** 

Pursuant to the requirements of the Securities Exchange Act of 1934, the Registrant has duly caused this Report to be signed on its behalf by the undersigned thereunto duly authorized.

CPG PARTNERS, L.P.

By: CHELSEA PROPERTY GROUP, INC., General Partner

By: <u>/s/ Michael J. Clarke</u>
Michael J. Clarke
Chief Financial Officer

Date: November 13, 2002

#### **CERTIFICATION**

I, David C. Bloom, Chief Executive Officer of Chelsea Property Group, Inc., the sole general partner of CPG Partners, L.P., (the "OP"), certify that:

- 1. I have reviewed this quarterly report on Form 10-Q of the OP;
- 2. Based on my knowledge, this quarterly report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this quarterly report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this quarterly report, fairly present in all material respects the financial condition, results of operations and cash flows of the OP as of, and for, the periods presented in this quarterly report;
- 4. The OP's other certifying officers and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-14 and 15d-14) for the OP and we have:
  - (a) designed such disclosure controls and procedures to ensure that material information relating to the OP, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this quarterly report is being prepared;
  - (b) evaluated the effectiveness of the OP's disclosure controls and procedures as of a date within 90 days prior to the filing date of this quarterly report (the "Evaluation Date"); and
  - (c) presented in this quarterly report our conclusions about the effectiveness of the disclosure controls and procedures based on our evaluation as of the Evaluation Date;
- 5. The OP's other certifying officers and I have disclosed, based on our most recent evaluation, to the OP's auditors and the audit committee of the OP's board of directors (or persons performing the equivalent function):
  - (a) all significant deficiencies in the design or operation of internal controls which could adversely affect the OP's ability to record, process, summarize and report financial data and have identified for the OP's auditors any material weaknesses in internal controls; and

- (b) any fraud, whether or not material, that involves management or other employees who have a significant role in the OP's internal controls; and
- 6. The OP's other certifying officers and I have indicated in this quarterly report whether or not there were significant changes in internal controls or in other factors that could significantly affect internal controls subsequent to the date of our most recent evaluation, including any corrective actions with regard to significant deficiencies and material weaknesses.

Date: November 13, 2002

/s/ David C. Bloom
David C. Bloom
Chief Executive Officer

#### CERTIFICATION

I, Michael J. Clarke, Chief Financial Officer of Chelsea Property Group, Inc., the sole general partner of CPG Partners, L.P., (the "OP"), certify that:

- 1. I have reviewed this quarterly report on Form 10-Q of the OP;
- 2. Based on my knowledge, this quarterly report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this quarterly report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this quarterly report, fairly present in all material respects the financial condition, results of operations and cash flows of the OP as of, and for, the periods presented in this quarterly report;
- 4. The OP's other certifying officers and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-14 and 15d-14) for the OP and we have:
  - (a) designed such disclosure controls and procedures to ensure that material information relating to the OP, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this quarterly report is being prepared;
  - (b) evaluated the effectiveness of the OP's disclosure controls and procedures as of a date within 90 days prior to the filing date of this quarterly report (the "Evaluation Date"); and
  - (c) presented in this quarterly report our conclusions about the effectiveness of the disclosure controls and procedures based on our evaluation as of the Evaluation Date;
- 5. The OP's other certifying officers and I have disclosed, based on our most recent evaluation, to the OP's auditors and the audit committee of the OP's board of directors (or persons performing the equivalent function):
  - (a) all significant deficiencies in the design or operation of internal controls which could adversely affect the OP's ability to record, process, summarize and report financial data and have identified for the OP's auditors any material weaknesses in internal controls; and
  - (b) any fraud, whether or not material, that involves management or other employees who have a significant role in the OP's internal controls; and
- 6. The OP's other certifying officers and I have indicated in this quarterly report whether or not there were significant changes in internal controls or in other factors that could significantly affect internal controls subsequent to the date of our most recent evaluation, including any corrective actions with regard to significant deficiencies and material weaknesses.

Date: November 13, 2002

/s/ Michael J. Clarke
Michael J. Clarke
Chief Financial Officer

#### **CERTIFICATION**

- I, David C. Bloom, Chief Executive Officer of Chelsea Property Group, Inc., the sole general partner of CPG Partners, L.P., (the "OP"), pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, do hereby certify as follows:
- 1. The quarterly report on Form 10-Q of the OP for the period ended September 30, 2002 fully complies with the requirements

- of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- 2. The information contained in such Form 10-Q fairly presents, in all material respects, the financial condition and results of operations of the OP.

IN WITNESS WHEREOF, I have executed this Certification this 13th day of November 2002.

/s/ David C. Bloom
David C. Bloom
Chief Executive Officer

#### **CERTIFICATION**

- I, Michael J. Clarke, Chief Financial Officer of Chelsea Property Group, Inc., the sole general partner of CPG Partners, L.P., (the "OP"), pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, do hereby certify as follows:
- 1. The quarterly report on Form 10-Q of the OP for the period ended September 30, 2002 fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- 2. The information contained in such Form 10-Q fairly presents, in all material respects, the financial condition and results of operations of the OP.

IN WITNESS WHEREOF, I have executed this Certification this 13th day of November, 2002.

/s/ Michael J. Clarke
Michael J. Clarke
Chief Financial Officer

#### AMENDED AND RESTATED CREDIT AGREEMENT

#### **AMONG**

#### CPG PARTNERS, L.P., A DELAWARE LIMITED PARTNERSHIP, AS BORROWER,

#### **AND**

FLEET NATIONAL BANK
TOGETHER WITH THOSE LENDERS NAMED ON THE
SIGNATURE PAGES HERETO AND ASSIGNEES
BECOMING PARTIES HERETO PURSUANT
TO SECTION 11.12, AS LENDERS,

#### **AND**

#### FLEET NATIONAL BANK AS ADMINISTRATIVE AGENT

#### **AND**

#### BANK OF AMERICA, N.A. AS CO-SYNDICATION AGENT

#### **AND**

### COMMERZBANK AG, NEW YORK BRANCH AS CO-SYNDICATION AGENT

#### **AND**

### WACHOVIA BANK, NATIONAL ASSOCIATION AS CO-DOCUMENTATION AGENT

#### **AND**

### PNC BANK, NATIONAL ASSOCIATION AS CO-DOCUMENTATION AGENT

#### Dated as of July 31, 2002

#### **TABLE OF CONTENTS**

|   |   | <u>Page</u>  |
|---|---|--|
| ARTICLE I   | <u>DEFINITIONS</u>  | 1  |
| 1.1<br>1.2<br>1.3   | Certain Defined Terms Computation of Time Periods Terms   | 1<br>23<br>23                                      |
| ARTICLE I   | LOANS AND LETTERS OF CREDIT   | 24   |
| 2.1<br>2.2<br>2.3<br>2.4<br>2.5<br>2.6<br>2.7<br>2.8<br>2.9 | Loan Advances and Repayment Authorization to Obtain Loans Lenders' Accounting Interest on the Loans Fees Payments Increased Capital Notice of Increased Costs Letters of Credit | 24<br>26<br>26<br>27<br>31<br>32<br>33<br>33<br>33 |
| ARTICLE I   | I UNENCUMBERED PROPERTIES AND PORTFOLIO PROPERTIES  | 36   |
| 3.1   | <u>Listing of Unencumbered Properties</u>   | 36   |

| 3.2<br>3.3<br>3.4                 | Rejection of Unencumbered Properties      | 37<br>37<br>37 |  |  |
|-----------------------------------|---|----------------|--|--|
| ARTICLE IV                        | V CONDITIONS TO LOANS                     | 37             |  |  |
| 4.1<br>4.2                        |   | 37<br>39       |  |  |
| ARTICLE V                         | REPRESENTATIONS AND WARRANTIES            | 40             |  |  |
| 5.1                               | Borrower Organization; Partnership Powers | 41             |  |  |
| 5.2                               |   | 41             |  |  |
| 5.3                               | REIT Organization; Corporate Powers       | 41             |  |  |
| 5.4                               |   | 41             |  |  |
| 5.5                               |   | 41             |  |  |
| 5.6                               |   | 42             |  |  |
| 5.7                               |   | 42             |  |  |
| 5.8                               |   | 42             |  |  |
| 5.9                               |   | 42             |  |  |
| 5.10                              |   | 42             |  |  |
| 5.11                              | <u></u>                                   | 43             |  |  |
| 5.12                              |   | 43             |  |  |
| 5.13<br>5.14                      |   | 43<br>43       |  |  |
| 5.14                              | <del>-</del>                              | 43<br>43       |  |  |
| 5.16                              |   | 43<br>44       |  |  |
| 5.17                              |   | <del>44</del>  |  |  |
| 5.18                              |   | 44             |  |  |
| 5.19                              |   | 44             |  |  |
| 5.20                              |   | 44             |  |  |
| 5.21                              |   | 45             |  |  |
| 5.22                              |   | 45             |  |  |
| 5.23                              |   | 45             |  |  |
| 5.24                              | <u>Solvency</u>                           | 45             |  |  |
| 5.25                              |   | 45             |  |  |
| 5.26                              |   | 45             |  |  |
| 5.27                              |   | 46             |  |  |
| 5.28                              |   | 46             |  |  |
| 5.29                              |   | 46             |  |  |
| 5.30                              | <u></u>                                   | 46             |  |  |
| 5.31                              | 8   | 46             |  |  |
| 5.32                              | Current Construction Projects             | 46             |  |  |
| ARTICLE VI REPORTING COVENANTS 47 |   |                |  |  |
| 6.1                               | <u> </u>                                  | 47             |  |  |
| 6.2                               |   | 51             |  |  |
| 6.3                               | <u>Confidentiality</u>                    | 51             |  |  |
| ARTICLE VII AFFIRMATIVE COVENANTS |   | 52             |  |  |
| 7.1                               | Existence                                 | 52             |  |  |
| 7.2                               |   | 52             |  |  |
| 7.3                               | Compliance with Laws, Etc.                | 52             |  |  |
| 7.4                               | Payment of Taxes and Claims               | 52             |  |  |
| 7.5                               | <u>.</u>                                  | 52             |  |  |
| 7.6                               |   | 53             |  |  |
| 7.7                               |   | 53             |  |  |
| 7.8                               |   | 54             |  |  |
| 7.9                               |   | 54             |  |  |
| 7.10                              | <u></u>                                   | 54<br>54       |  |  |
| 7.11                              |   | 54<br>= 4      |  |  |
| 7.12<br>7.13                      |   | 54<br>54       |  |  |
| 7.13<br>7.14                      | — <u>1—3——0——</u>                         | 54<br>54       |  |  |
| / <b>. 1</b> <del>'</del>         | interest rate contracts                   | J <del>·</del> |  |  |
| ARTICLE V                         | <u>NEGATIVE COVENANTS</u>                 | 55             |  |  |

| 8.1         | Restrictions on Fundamental Changes  | 55       |
|-------------|--|----------|
| 8.2         | ERISA  | 55       |
| 8.3         | Amendment of Constituent Documents   | 56       |
| 8.4         | Margin Regulations   | 56       |
| 8.5         | With Respect to the REIT   | 56       |
| 8.6         | Restrictions on Indebtedness   | 57       |
| 8.7         |  | 57       |
|             | Construction Projects  Discontinuity in Management   |          |
| 8.8         | <u>Discontinuity in Management</u>   | 57       |
| ARTICLE IX  | X FINANCIAL COVENANTS  | 58       |
| THEFTCEL 12 | THAIRCINE GOVERNIVIO   | 50       |
| 9.1         | Value of All Unencumbered Properties   | 58       |
| 9.2         | Minimum Debt Service Coverage  | 58       |
| 9.3         | Minimum Fair Market Net Worth  | 58       |
| 9.4         | Total Liabilities to Adjusted Asset Value Ratio  | 58       |
| 9.5         | Maximum Secured Borrower Debt  | 58       |
|             |  | 56<br>58 |
| 9.6         | Operating Cash Flow to Debt Service Ratio  |          |
| 9.7         | EBITDA to Fixed Charges Ratio  | 58       |
| 9.8         | Aggregate Occupancy Rate   | 58       |
| 9.9         | <u>Distributions</u>   | 58       |
| 9.10        | <u>Permitted Investments</u>   | 59       |
| 9.11        | Repurchase of Equity Interests   | 59       |
| 9.12        | <u>Calculation</u>   | 59       |
|             |  |          |
| ARTICLE X   | EVENTS OF DEFAULT; RIGHTS AND REMEDIES   | 60       |
|             |  |          |
| 10.1        | Events of Default  | 60       |
| 10.2        | Rights and Remedies  | 63       |
| 10.3        | <u>Rescission</u>  | 64       |
|             |  |          |
| ARTICLE X   | <u>AGENCY PROVISIONS</u>   | 65       |
|             |  |          |
| 11.1        | <u>Appointment</u>   | 65       |
| 11.2        | Nature of Duties   | 65       |
| 11.3        | <u>Loan Disbursements</u>  | 66       |
| 11.4        | <u>Distribution and Apportionment of Payments</u>  | 67       |
| 11.5        | Rights, Exculpation, Etc   | 68       |
| 11.6        | Reliance   | 69       |
| 11.7        | Indemnification  | 69       |
| 11.8        | Agent Individually   | 69       |
| 11.9        | Successor Agent; Resignation of Agent; Removal of Agent  | 70       |
| 11.10       | Consent and Approvals  | 71       |
| 11.11       | Agency Provisions Relating to Certain Enforcement Actions                                      | 72       |
| 11.12       | Assignments and Participations   | 73       |
| 11.12       | Ratable Sharing  | 75<br>75 |
| 11.13       | Delivery of Documents  | 76       |
| 11.14       | Notice of Events of Default  | 76<br>76 |
| 11.15       | Notice of Events of Default  | 70       |
| ARTICLE X   | III MISCELLANEOUS  | 76       |
| ARTICLEA    | MI MISCELLAIVEOUS  | 70       |
| 12.1        | Expenses   | 76       |
| 12.2        | <u>Indemnity</u>   | 77       |
| 12.3        | <u>Change in Accounting Principles</u>   | 78       |
| 12.3        | Amendments and Waivers   | 78<br>78 |
|             |  |          |
| 12.5        | Independence of Covenants  Notices and Delivery  | 80       |
| 12.6        | Notices and Delivery   | 80       |
| 12.7        | Survival of Warranties, Indemnities and Agreements  Entire and Indemnities and Agreements      | 80       |
| 12.8        | Failure or Indulgence Not Waiver; Remedies Cumulative  | 80       |
| 12.9        | Setoff; Payments Set Aside   | 81       |
| 12.10       | <u>Severability</u>  | 81       |
| 12.11       | Heading  | 81       |
| 12.12       | Governing Law  | 81       |
| 12.13       | <u>Limitation of Liability</u>   | 81       |
| 12.14       | Successors and Assigns   | 82       |
| 12.15       | Consent to Jurisdiction and Service of Process; Waiver of Jury Trial and Certain Damage Claims | 82       |
| 12.16       | Counterparts; Effectiveness; Inconsistencies   | 83       |
| 12.17       | Construction   | 83       |

| 12.18 | Obligations Unsecured     | 84 |
|-------|---------------------------|----|
| 12.19 | Replacement of Loan Notes | 84 |
| 12.20 | Entire Agreement          | 84 |

#### LIST OF EXHIBITS AND SCHEDULES

#### **Exhibits:**

A - Form of Assignment and AssumptionB -- Form of Compliance Certificate

C -- Form of Loan Notes

-- Form of Notice of Borrowing
-- Form of Fixed Rate Notice

F -- Form of Letter of Credit Request

#### Schedules:

E

1 - List of Unencumbered Properties

1.1 -- List of Portfolio Properties which are not

**Unencumbered Properties** 

5.5 -- Unconsolidated Affiliates and Subsidiaries

5.22 -- Environmental Matters

5.32 -- Current Construction Projects

#### AMENDED AND RESTATED CREDIT AGREEMENT

THIS AMENDED AND RESTATED CREDIT AGREEMENT is dated as of July 31, 2002 and is among CPG PARTNERS, L.P., a Delaware limited partnership ("Borrower"), each of the Lenders named on the signature pages hereof (the "Lenders") and FLEET NATIONAL BANK, a national banking association ("Fleet") in its capacity as Agent for the Lenders (the "Agent").

#### **RECITALS**

- A. The Borrower is the borrower under that certain Credit Agreement dated as of March 30, 1998 among Borrower, certain of the Lenders and the Agent, as amended by First Amendment to Credit Agreement dated as of March 10, 1999, by Second Amendment to Credit Agreement dated as of March 30, 1999, by Third Amendment to Credit Agreement dated as of October 8, 1999, by Fourth Amendment to Credit Agreement dated as of March 30, 2000, by Fifth Amendment to Credit Agreement dated as of December 29, 2000, by Sixth Amendment to Credit Agreement dated as of June 27, 2001, by Seventh Amendment to Credit Agreement dated as of October 29, 2001 (as so amended, the "1998 Credit Agreement"), pursuant to which certain of the Lenders agreed to provide Borrower with an unsecured revolving loan facility in the maximum principal amount of \$160,000,000 on the terms and conditions set forth therein (the "1998 Facility").
- B. Borrower has requested that the 1998 Facility be increased to a maximum principal amount of \$200,000,000 and be modified based on the terms and conditions provided for herein (the "<u>Facility</u>"), and the Lenders are willing to provide the requested Facility on the terms and conditions set forth herein.
- C. The existing \$5,034,536 term loan from Fleet to Borrower pursuant to a Term Loan Agreement dated as of March 30, 1998 as amended is also being modified as set forth in an Amended and Restated Term Loan Agreement of even date herewith (the "Fleet Term Loan").

NOW, THEREFORE, the parties hereto agree as follows:

#### **ARTICLE I**

#### **DEFINITIONS**

1.1 <u>Certain Defined Terms</u>. The following terms used in this Agreement shall have the following meanings (such meanings to be applicable, except to the extent otherwise indicated in a definition of a particular term, both to the singular and the plural forms of the terms defined):

"Accommodation Obligations", as applied to any Person, means any Indebtedness or other contractual obligation or liability, contingent or otherwise, of another Person in respect of which that Person is liable, including, without limitation, any such Indebtedness, obligation or liability directly or indirectly guaranteed, endorsed (otherwise than for collection or deposit in the ordinary course of business), co-made or discounted or sold with recourse by that Person, or in respect of which that Person is otherwise directly or indirectly liable, including in respect of any partnership in which that Person is a general partner, Contractual Obligations (contingent or otherwise) arising through any agreement to purchase, repurchase or otherwise acquire such

Indebtedness, obligation or liability or any security therefor, or to provide funds for the payment or discharge thereof (whether in the form of loans, advances, stock purchases, capital contributions or otherwise), or to maintain solvency, assets, level of income, or other financial condition, or to make payment other than for value received.

"<u>Accountants</u>" means Ernst & Young LLP, any other "big five" accounting firm or another firm of certified public accountants of national standing selected by Borrower and acceptable to Agent.

"<u>Acquisition Price</u>" means the aggregate purchase price for an asset including bona fide purchase money financing provided by the seller and all existing Indebtedness pertaining to such asset.

"Adjusted Asset Value" means, as at any date of determination, the sum (without duplication of any item) of (i) cash and Cash Equivalents owned by Borrower (excluding any tenant deposits or other restricted cash or escrows), (ii) an amount equal to 50% of the construction-in-progress as reported in the Financial Statements and 50% of the Borrower's Share of the construction-in-progress of Unconsolidated Affiliates, (iii) the Value of All Premium Centers, (iv) the Value of All Other Retail Centers, and (v) the Value of All Other Properties.

"Affiliates" as applied to any Person, means any other Person directly or indirectly controlling, controlled by, or under common control with, that Person. For purposes of this definition, "control" (including, with correlative meanings, the terms "controlling", "controlled by" and "under common control with"), as applied to any Person, means (i) the possession, directly or indirectly, of the power to vote twenty-five percent (25%) or more of the Securities having voting power for the election of directors of such Person or otherwise to direct or cause the direction of the management and policies of that Person, whether through the ownership of voting securities or by contract or otherwise, or (ii) the ownership of a general partnership interest or a limited partnership interest (or other ownership interest) representing twenty-five percent (25%) or more of the outstanding limited partnership interests or other ownership interests of such Person. In addition, any corporation, partnership or other entity in which the ownership interests of Borrower and its Subsidiaries represents ten percent (10%) or more of the outstanding ownership interests shall be deemed to be an Affiliate of Borrower.

"<u>Agent</u>" means Fleet in its capacity as agent for the Lenders under this Agreement, and shall include any successor Agent appointed pursuant hereto and shall be deemed to refer to Fleet in its individual capacity as a Lender where the context so requires.

"Aggregate Occupancy Rate" means, with respect to the Unencumbered Properties at any time, the ratio, as of such date, expressed as a percentage, of (i) the gross leasable area of all Unencumbered Properties occupied by tenants paying rent pursuant to Leases other than Materially Defaulted Leases, to (ii) the aggregate gross leasable area of all Unencumbered Properties, excluding from both (i) and (ii) the gross leasable area of Construction Projects prior to the date which is 3 months after the Rent Stabilization Date for such Construction Project. Only premises which are actually open for business shall be counted as occupied.

"<u>Agreement</u>" means this Amended and Restated Credit Agreement as the same may be amended, supplemented or modified from time to time.

"Applicable LIBOR Rate Margin" means, as of any date of determination: (i) 0.85%, if Borrower's senior long-term unsecured debt obligations are rated at least BBB+ or Baa1 by one or both of the Rating Agencies, (ii) 0.95%, if Borrower's senior long-term unsecured debt obligations are rated at least BBB or Baa2 by one or both of the Rating Agencies, (iii) 1.05%, if Borrower's senior long-term unsecured debt obligations are rated at least BBB- or Baa3 by one or both of the Rating Agencies, (iv) 1.25% if Borrower's senior long-term unsecured debt obligations are rated at least Ba1/Ba2 or BB+/BB by one or both of the Rating Agencies or (v) 1.50%, in any other case (including, without limitation, if Borrower's senior long-term unsecured debt obligations are not rated by either of the Rating Agencies).

"<u>Assignment and Assumption</u>" means an Assignment and Assumption in the form of <u>Exhibit A</u> hereto (with blanks appropriately filled in) delivered to Agent in connection with each assignment of a Lender's interest under this Agreement pursuant to Section 11.12.

"ATC Partnership" means Cannery Row Associates, a California limited partnership in which the Borrower is the sole limited partner (having a 99% interest) and the REIT is the sole general partner (having a 1% interest).

"Base Rate" means, on any day, the higher of (i) the Federal Funds Rate in effect on such day <u>plus</u> one-half percent (0.5%) per annum or (ii) the variable per annum rate of interest so designated from time to time by Fleet as its prime rate. The prime rate is a reference rate and does not necessarily represent the lowest or best rate being charged to any customer.

"Base Rate Loans" means those Loans bearing interest at the Base Rate.

"Benefit Plan" means any employee pension benefit plan as defined in Section 3(2) of ERISA (other than a Multiemployer Plan) in respect of which a Person or an ERISA Affiliate is, or within the immediately preceding five (5) years was, an "employer" as defined in Section 3(5) of ERISA.

"<u>Borrower</u>" means CPG Partners, L.P., a Delaware limited partnership that was formerly known as Chelsea GCA Realty Partnership, L.P.

"Borrower Debt" means (without duplication) all Indebtedness of Borrower or any Subsidiary of Borrower, without offset or reduction in respect of prepaid interest, restructuring fees or similar items, minus, in the case of Nonrecourse Indebtedness of a Subsidiary that is not a Wholly-Owned Subsidiary, the amount of such Indebtedness in excess of Borrower's Share thereof, plus the Borrower's Share of the Indebtedness of all Unconsolidated Affiliates.

"Borrower's Share" means, in the case of an Unconsolidated Affiliate or a Subsidiary that is not a Wholly-Owned Subsidiary, the percentage ownership interest in such Unconsolidated Affiliate or a Subsidiary that is directly or indirectly owned by Borrower.

"Borrowing" means a borrowing of Loans under the Facility.

"Business Day" means any day other than a Saturday, Sunday or day which shall be in the Commonwealth of Massachusetts a legal holiday or day on which banking institutions are required or authorized to close. When the term Business Day is used with respect to LIBOR Loans, such day must also be a day on which commercial banks are open for international business (including dealings in dollar deposits) in London, England.

"<u>Capital Leases</u>", as applied to any Person, means any lease of any property (whether real, personal or mixed) by that Person as lessee which, in conformity with GAAP, is or should be accounted for as a capital lease on the balance sheet of that Person.

"Cash Equivalents" means (i) marketable direct obligations issued or unconditionally guaranteed by the United States Government or issued by an agency thereof and backed by the full faith and credit of the United States, in each case maturing within one (1) year after the date of acquisition thereof; (ii) marketable direct obligations issued by any state of the United States of America or any political subdivision of any such state or any public instrumentality thereof maturing within ninety (90) days after the date of acquisition thereof and, at the time of acquisition, having one of the two highest ratings obtainable from any two of Standard & Poor's Corporation, Moody's Investors Services, Inc., Duff and Phelps, or Fitch Investors (or, if at any time no two of the foregoing shall be rating such obligations, then from such other nationally recognized rating services as may be acceptable to Agent) and not listed for possible down-grade in Credit Watch published by Standard & Poor's Corporation; (iii) commercial paper, other than commercial paper issued by Borrower or any of its Affiliates, maturing no more than ninety (90) days after the date of creation thereof and, at the time of acquisition, having a rating of at least A-1 or P-1 from either Standard & Poor's Corporation or Moody's Investor's Service, Inc. (or, if at any time neither Standard & Poor's Corporation nor Moody's Investor's Service, Inc. shall be rating such obligations, then the highest rating from such other nationally recognized rating services as may be acceptable to Agent); (iv) domestic and Eurodollar certificates of deposit or time deposits or bankers' acceptances maturing within ninety (90) days after the date of acquisition thereof, overnight securities repurchase agreements, or reverse repurchase agreements secured by any of the foregoing types of securities or debt instruments issued, in each case, by (A) any commercial bank organized under the laws of the United States of America or any state thereof or the District of Columbia having combined capital and surplus of not less than Two Hundred Fifty Million Dollars (\$250,000,000) or (B) any Lender, and (v) deposits in existing Merrill Lynch money market accounts maintained by Borrower, such deposits being the proceeds from the exercise of stock options pursuant to Borrower's employee stock option plans.

"CIP Budget Amount" means the total budgeted cost (as such budget shall be updated from time to time) of all Current Construction Projects (excluding Expansion Projects) owned by Borrower or any of its Wholly-Owned Subsidiaries plus the Borrower's Share of the total budgeted cost (as such budget shall be updated from time to time) of all Current Construction Projects (excluding Expansion Projects) owned by any Unconsolidated Affiliate or by any Subsidiary of Borrower that is not a Wholly-Owned Subsidiary provided, however, that with respect to (i) Current Construction Projects owned by any such Unconsolidated Affiliate or Subsidiary for which the respective financial responsibilities of the other owners of such Unconsolidated Affiliate or Subsidiary on the one hand and Borrower and its Wholly-Owned Subsidiaries on the other hand are not in proportion to the respective ownership interests in the applicable Unconsolidated Affiliate or Subsidiary, then the portion of budgeted cost of such project included in the CIP Budget Amount shall be adjusted to reflect Borrower's actual financial responsibility related to such project. Such costs shall include, without limitation, all land acquisition costs (but may exclude costs of land used for expansion projects which was not purchased for the purpose of such expansion project), design and permitting costs, construction period real estate taxes, leasing costs including brokers' commissions and tenant improvements, allowances or reimbursements, construction costs and opening costs. With respect to any Construction Projects financed with Indebtedness other than this Facility, such costs shall also include construction period interest and all fees and expenses associated with such Indebtedness.

"Closing Date" means the date that this Amended and Restated Credit Agreement shall become effective pursuant to Section 4.1.

"Commission" means the Securities and Exchange Commission.

"Commitment" means, with respect to any Lender, the principal amount set out under such Lender's name under the heading "Loan Commitment" on the signature pages attached to this Agreement or as set forth on an Assignment and Assumption executed by such Lender, as assignee.

"Completion Date" means, with respect to a Construction Project, the date on which certificates of occupancy (or the equivalent) have been issued for at least 90% of the gross leasable area of such Construction Project.

"Compliance Certificate" means a certificate in the form of Exhibit B delivered to Agent by Borrower pursuant to Section 3.4, Section 6.1.4, Section 6.1.11, Section 8.1 or any other provision of this Agreement and covering Borrower's compliance with the financial covenants contained in Article IX.

"Confidential Information" has the meaning ascribed to such term in Section 6.3.

"Construction Project" means a project consisting of the construction of new buildings, additions to existing buildings, and/or rehabilitation of existing buildings (other than normal refurbishing and tenant fit-up work when one retail tenant leases space previously occupied by another retail tenant).

"Contaminant" means any pollutant (as that term is defined in 42 U.S.C. 9601(33)) or toxic pollutant (as that term is defined in 33 U.S.C. 1362(13)), hazardous substance (as that term is defined in 42 U.S.C. 9601(14)), hazardous chemical (as that term is defined by 29 CFR Section 1910.1200(c)), toxic substance, hazardous waste (as that term is defined in 42 U.S.C. 6903(5)), radioactive material, special waste, petroleum (including crude oil or any petroleum-derived substance, waste, or breakdown or decomposition product thereof), any constituent of any such substance or waste, including, but not limited to, polychlorinated biphenyls and asbestos, or any other substance or waste deleterious to the environment the release, disposal or remediation of which is now or at any time becomes subject to regulation under any Environmental Law.

"Contractual Obligation", as applied to any Person, means any provision of any Securities issued by that Person or any indenture, mortgage, deed of trust, lease, contract, undertaking, document or instrument to which that Person is a party or by which it or any of its properties is bound, or to which it or any of its properties is subject (including, without limitation, any restrictive covenant affecting such Person or any of its properties).

"Court Order" means any judgment, writ, injunction, decree, rule or regulation of any court or Governmental Authority binding upon or applicable to the Person in question.

"<u>Current Construction Project</u>" means a Construction Project from the time of commencement of construction of footings and foundations for such Construction Project until the Rent Stabilization Date of such Construction Project.

"<u>Debt Service</u>" means, for any period, Interest Expense for such <u>period</u> plus scheduled principal amortization (<u>i.e.</u>, excluding any balloon payment due at maturity) for such period on all Borrower Debt.

"December 31, 2001 Financials" has the meaning given to such term in Section 5.9.

"<u>Defaulting Lender</u>" means any Lender which fails or refuses to perform its obligations under this Agreement within the time period specified for performance of such obligation or, if no time frame is specified, if such failure or refusal continues for a period of five (5) Business Days after notice from Agent.

"DOL" means the United States Department of Labor and any successor department or agency.

"Dollars" and "\$" means the lawful money of the United States of America.

"<u>Drawing Date</u>" means the date on which a draft under a Letter of Credit is paid by the Agent.

"EBITDA" means, at any time, for the most recent Fiscal Quarter, without duplication, the Borrower's earnings (or loss) before interest, taxes, depreciation and amortization, calculated for such period on a consolidated basis in conformity with GAAP, including only the Borrower's Share of the earnings (or loss) before interest, taxes, depreciation and amortization of all Unconsolidated Affiliates and of any Subsidiary that is not a Wholly-Owned Subsidiary minus gains (and plus losses) from extraordinary items or asset sales or write-ups or forgiveness of Indebtedness, minus percentage rent income for such Fiscal Quarter, plus twenty-five percent (25%) of the total percentage rent income during such Fiscal Quarter and the three immediately preceding Fiscal Quarters.

"Eligible Affiliate" means an Affiliate of Borrower which is not a Wholly-Owned Subsidiary that does not have any Indebtedness other than Indebtedness in connection with accounts payable to unsecured trade creditors for goods and services and current operating liabilities (not the result of the borrowing of money) incurred in the ordinary course of business of such Affiliate. Eligible Affiliates as of the date hereof are listed on Schedule 5.5.

"Eligible Assignee" means (i) (A) (1) a commercial bank organized under the laws of the United States or any state thereof; (2) a savings and loan association or savings bank organized under the laws of the United States or any state thereof; or (3) a commercial bank organized under the laws of any other country or a political subdivision thereof, provided that (x) such bank is acting through a branch or agency located in the United States, or (y) such bank is organized under the laws of a country that is a member of the Organization for Economic Cooperation and Development or a political subdivision of such country; that (B) in each case, is (1) reasonably acceptable to Agent and Borrower, and (2) has total assets in excess of \$10,000,000,000 and a rating on its (or its parent's) senior unsecured debt obligations of at least BBB by one of the Rating Agencies; or (ii) any Lender or Affiliate of any Lender; provided that no Affiliate of Borrower shall be an Eligible Assignee.

"Environmental Laws" has the meaning set forth in Section 5.22.

"<u>Environmental Lien</u>" means a Lien in favor of any Governmental Authority for (i) any liability under Environmental Laws, or (ii) damages arising from, or costs incurred by such Governmental Authority in response to, a Release or threatened

Release of a Contaminant into the environment.

"<u>ERISA</u>" means the Employee Retirement Income Security Act of 1974, as amended from time to time, and any successor statute.

"ERISA Affiliate" of any Person means any (i) corporation which is, becomes, or is deemed to be a member of the same controlled group of corporations (within the meaning of Section 414(b) of the Internal Revenue Code) as such Person, (ii) partnership, trade or business (whether or not incorporated) which is, becomes or is deemed to be under common control (within the meaning of Section 414(c) of the Internal Revenue Code) with such Person, (iii) other Person which is, becomes or is deemed to be a member of the same "affiliated service group" (as defined in Section 414(m) of the Internal Revenue Code) as such Person, or (iv) any other organization or arrangement described in Section 414(o) of the Internal Revenue Code which is, becomes or is deemed to be required to be aggregated pursuant to regulations issued under Section 414(o) of the Internal Revenue Code with such Person pursuant to Section 414(o) of the Internal Revenue Code.

"<u>Event of Default</u>" means any of the occurrences set forth in <u>Article X</u> after the expiration of any applicable grace period expressly provided therein.

"<u>Executive Officers</u>" mean David C. Bloom, William D. Bloom, Leslie T. Chao, Michael J. Clarke and Thomas J. Davis.

"Expansion Projects" means Construction Projects consisting of the construction of additional buildings or building additions at a Portfolio Property which has been open and operating prior to the of such Construction Project so long as the increase in the gross leasable area of such Property as a result of such Construction Project does not exceed 150,000 square feet.

"Facility" means the loan facility of Two Hundred Million Dollars (\$200,000,000) described in Section 2.1.1.

"Facility Fee" has the meaning given to such term in Section 2.5.1.

"Fair Market Net Worth" means the Borrower's Adjusted Asset Value less Total Liabilities.

"FDIC" means the Federal Deposit Insurance Corporation or any successor thereto.

"Federal Funds Rate" means, for any period, a fluctuating interest rate per annum equal for each day during such period to the weighted average of the rates on overnight Federal Funds transactions with members of the Federal Reserve System arranged by Federal Funds brokers, as published for such day (or, if such day is not a Business Day, for the next preceding Business Day) by the Federal Reserve Bank of New York, or, if such rate is not so published for any day which is a Business Day, the average of the quotations for such day on such transactions received by Agent from three Federal Funds brokers of recognized standing selected by Agent.

"<u>Federal Reserve Board</u>" means the Board of Governors of the Federal Reserve System or any governmental authority succeeding to its functions.

"<u>Financial Statements</u>" means the consolidated financial statements of the Borrower, including the notes thereto, as contained in the most recent Form 10-K or Form 10-Q filed with the Commission.

"Fiscal Quarter" means each three-month period ending on March 31, June 30, September 30 and December 31.

"<u>Fiscal Year</u>" means the fiscal year of Borrower which shall be the twelve (12) month period ending on the last day of December in each year.

"<u>Fixed Charges</u>" means, for any period, Debt Service <u>plus</u> scheduled dividends or distributions due with respect to preferred partnership units in the Borrower or preferred stock in the REIT.

"<u>Fixed Rate Notice</u>" means, with respect to a LIBOR Loan pursuant to <u>Section 2.1.2</u>, a notice substantially in the form of <u>Exhibit E</u>.

"<u>Fleet Term Loan</u>" means the term loan from <u>Fleet</u> to Borrower in the principal amount of \$5,046,536 pursuant to an Amended and Restated Term Loan Agreement of even date herewith and any replacements, extensions or refinancing thereof. Such Term Loan Agreement provides for the same interest rate and substantially the same terms and conditions as set forth in this Agreement.

"<u>Foreign Affiliates</u>" means Unconsolidated Affiliates which directly or indirectly develop, own or finance one or more Foreign Properties. Foreign Affiliates existing on the date hereof are listed on <u>Schedule 5.5</u>.

"Foreign Investments" means the sum, without duplication, of the following: (i) the aggregate amount of all Investments by Borrower in Borrower's Foreign Affiliates, (ii) the face amount of all Letters of Credit issued hereunder (or letters of credit issued by any other Person with respect to which Borrower is directly or contingently liable), or the maximum amount of all of Borrower's Accommodation Obligations, for the benefit of such Foreign Affiliates, (iii) the aggregate Acquisition Prices of all Foreign Properties owned by Borrower or its Wholly-Owned Subsidiaries, and (iv) the Borrower's Share of the aggregate Acquisition Prices of all Foreign Properties owned by Subsidiaries of Borrower that are not Wholly-Owned Subsidiaries.

"Foreign Properties" means all Properties which are not located within the boundaries of the United States.

"Funding Date" means, with respect to any Loan made after the Closing Date, the date of the funding of such Loan.

"<u>Funds From Operations</u>" means, for any period, the Borrower's Funds From Operations determined in accordance with the definition approved by the National Association of Real Estate Investment Trusts.

"GAAP" means generally accepted accounting principles set forth in the opinions and pronouncements of the Accounting Principles Board and the American Institute of Certified Public Accountants and statements and pronouncements of the Financial Accounting Standards Board, or in such other statements by such other entity as may be in general use by significant segments of the accounting profession, which are applicable to the circumstances as of the date of determination.

"<u>Governmental Authority</u>" means any nation or government, any federal, state, local, municipal or other political subdivision thereof or any entity exercising executive, legislative, judicial, regulatory or administrative functions of or pertaining to government.

"<u>Guarantor Subsidiary</u>" means a Wholly-Owned Subsidiary which executes and delivers a guaranty of the Obligations in favor of the Agent and the Lenders, which guaranty shall be substantially in the form of the Guaranty from certain Guarantor Subsidiaries of even date herewith, and shall be accompanied by certificates and a legal opinion reasonably acceptable to the Agent. Guarantor Subsidiaries as of the date hereof are listed on <u>Schedule 5.5</u>.

"Guaranty" means the Guaranty of even date herewith executed by the REIT in favor of the Agent and the Lenders.

"Indebtedness", as applied to any Person (determined on a consolidated basis and without duplication), means the sum of (i) all indebtedness, obligations or other liabilities of such Person for borrowed money, (ii) all indebtedness, obligations or other liabilities of such Person evidenced by Securities or other similar instruments, (iii) all reimbursement obligations and other liabilities of such Person with respect to letters of credit or banker's acceptances issued for such Person's account, (iv) all obligations of such Person to pay the deferred purchase price of Property or services or to reimburse tenants for the costs of improvements constructed by such tenants on the Property of such Person, (v) all obligations in respect of Capital Leases of such Person, (vi) all Accommodation Obligations of such Person, (vii) all indebtedness, obligations or other liabilities of such Person or others secured by a Lien on any asset of such Person, whether or not such indebtedness, obligations or liabilities are assumed by, or are a personal liability of, such Person (including, without limitation, the principal amount of any assessment or similar indebtedness encumbering any property), (viii) all indebtedness, obligations or other liabilities (other than interest expense liability) in respect of Interest Rate Contracts and foreign currency exchange agreements, and (ix) ERISA obligations currently due and payable. Indebtedness shall not include accrued ordinary operating expenses payable on a current basis.

"Interest Expense" means, for any period, total interest expense, whether paid, accrued or capitalized (including the interest component of Capital Leases) in respect of Borrower Debt, including, without limitation, amortization of loan acquisition costs, all commissions, discounts and other fees and charges owed with respect to letters of credit, and net costs under Interest Rate Contracts.

"Interest Period" means, relative to any LIBOR Loans comprising part of the same Borrowing, the period beginning on (and including) the date on which such LIBOR Loans are made as, or converted into, LIBOR Loans, and ending on (but excluding) the day which numerically corresponds to such date seven (7), thirty (30), sixty (60), ninety (90) or one hundred eighty (180) days thereafter, in either case as Borrower may select in its relevant Notice of Borrowing pursuant to Section 2.1.2; provided, however, that:

- (a) if such Interest Period would otherwise end on a day which is not a Business Day, such Interest Period shall end on the next following Business Day; and
  - (b) no Interest Period may end later than the Termination Date.

"<u>Interest Rate Contracts</u>" means, collectively, interest rate swap, collar, cap or similar agreements providing interest rate protection.

"Internal Revenue Code" means the Internal Revenue Code of 1986, as amended from time to time hereafter, and any successor statute.

"Internet Affiliates" means Affiliates of Borrower which directly or indirectly develop, own or finance internet online stores and other technology-based e-commerce platforms and services. Internet Affiliates existing on the date hereof are listed on Schedule 5.5.

"Internet Investments" means the sum, without duplication, of the following: (i) the aggregate amount of all Investments by Borrower in Borrower's Internet Affiliates, and (ii) the face amount of all Letters of Credit issued hereunder (or letters of credit issued by any other Person with respect to which Borrower is directly or contingently liable), or the maximum amount of all of Borrower's Accommodation Obligations, for the benefit of such Internet Affiliates.

"Investment" means, as applied to any Person, (i) any direct or indirect purchase or other acquisition by that Person of Securities, or of a beneficial interest in Securities, of any other Person, (ii) any direct or indirect loan, advance (other than deposits with financial institutions available for withdrawal on demand, prepaid expenses, advances to employees and similar items made or incurred in the ordinary course of business), or capital contribution by such Person to any other Person, including all Indebtedness and accounts owed by that other Person which are not current assets or did not arise from sales of goods or services to that Person in the ordinary course of business, and (iii) any liabilities of such Person (contingent or otherwise) under guarantees or other Accommodation Obligations with respect to the outstanding Indebtedness of that other Person. The amount of any Investment described in clause (iii) of this definition shall be deemed to be in the maximum amount of such Person's liability thereunder.

"<u>Investment Mortgages</u>" mean notes receivable or other indebtedness secured by mortgages or other security interests directly or indirectly owned by Borrower or any Subsidiary of Borrower, including certificates of interest in real estate mortgage investment conduits.

"IRS" means the Internal Revenue Service and any Person succeeding to the functions thereof.

"Issuance Date" means the date of issuance of any Letters of Credit.

"Land" means unimproved real estate, including future phases of a partially completed project, owned by Borrower or any Subsidiary of Borrower for the purpose of future development of improvements. For purposes of the foregoing definition, "unimproved" shall mean Land on which the construction of building improvements has not commenced or land on which construction has been discontinued for a continuous period longer than sixty (60) days prior to completion.

"<u>Lease</u>" means a lease or license between Borrower and a tenant or licensee with respect to premises located within a Portfolio Property.

"Lender Taxes" has the meaning given to such term in Section 2.4.7.

"Lenders" means Fleet and any other bank, finance company, insurance or other financial institution which is or becomes a party to this Agreement by execution of a counterpart signature page hereto or an Assignment and Assumption, as assignee. With respect to matters requiring the consent to or approval of all Lenders at any given time, all then existing Defaulting Lenders will be disregarded and excluded, and, for voting purposes only, "all Lenders" shall be deemed to mean "all Lenders other than Defaulting Lenders".

"Letter of Credit" means a letter of credit issued by the Agent for the account of Borrower pursuant to Section 2.9.

"<u>Liabilities and Costs</u>" means all claims, judgments, liabilities, obligations, responsibilities, losses, damages (including lost profits), punitive or treble damages, costs, disbursements and expenses (including, without limitation, reasonable attorneys, experts' and consulting fees and costs of investigation and feasibility studies), fines, penalties and monetary sanctions, interest, direct or indirect, known or unknown, absolute or contingent, past, present or future.

"LIBOR" means, relative to any Interest Period for any LIBOR Loan included in any Borrowing, the rate per annum as determined on the basis of the offered rates for deposits in U.S. Dollars, for a period of time comparable to the Interest Period applicable to such LIBOR Loan which appears on the Telerate page 3750 as of 11:00 a.m. London time on the day that is two Business Days preceding the first day of such LIBOR Loan; provided, however, if the rate described above does not appear on the Telerate System on any applicable interest determination date, the LIBOR rate shall be the rate (rounded upward, if necessary, to the nearest one hundred-thousandth of a percentage point), determined on the basis of the offered rates for deposits in U.S. dollars for a period of time comparable to the Interest Period applicable to such LIBOR Loan which are offered by four major banks in the London interbank market at approximately 11:00 a.m. London time, on the day that is two (2) Business Days preceding the first day of such LIBOR Loan as selected by the Agent. The principal London office of each of the four major London banks will be requested to provide a quotation of its U.S. Dollar deposit offered rate. If at least two such quotations are provided, the rate for that date will be the arithmetic mean of the quotations. If fewer than two quotations are provided as requested, the rate for that date will be determined on the basis of the rates quoted for loans in U.S. dollars to leading European banks for a period of time comparable to the Interest Period applicable to such LIBOR Loan offered by major banks in New York City at approximately 11:00 a.m. New York City time, on the day that is two Business Days preceding the first day of such Interest Period applicable to such LIBOR Loan. In the event that the Agent is unable to obtain any such quotation as provided above, it will be deemed that LIBOR pursuant to a LIBOR Loan cannot be determined. In the event that the Board of Governors of the Federal Reserve System shall impose a LIBOR Reserve Percentage with respect to LIBOR deposits of the Agent, then for any period during which such LIBOR Reserve Percentage shall apply, LIBOR shall be equal to the amount determined above divided by an amount equal to 1 minus such LIBOR Reserve Percentage.

"<u>LIBOR Loan</u>" means a Loan bearing interest, at all times during an Interest Period applicable to such Loan, at a fixed rate of interest determined by reference to LIBOR.

"<u>LIBOR Office</u>" means, relative to any Lender, the office of such Lender designated as such on the counterpart signature pages hereto or such other office of a Lender as designated from time to time by notice from such Lender to Agent, whether or not outside the United States, which shall be making or maintaining LIBOR Loans of such Lender.

"LIBOR Prepayment Fee" has the meaning given to such term in Section 2.4.8(c).

"<u>LIBOR Reserve Percentage</u>" means the minimum aggregate reserve requirement (including all basic, supplemental, marginal and other reserves) which is imposed on member banks of the Federal Reserve System against "Euro-currency Liabilities" as defined in Regulation D.

"<u>Lien</u>" means any mortgage, deed of trust, pledge, hypothecation, assignment, deposit arrangement, security interest, encumbrance (including, but not limited to, easements, rights-of-way, zoning restrictions and the like), lien (statutory or other), preference, priority or other security agreement or preferential arrangement of any kind or nature whatsoever, including without limitation any conditional sale or other title retention agreement, any financing lease having substantially the same economic effect as any of the foregoing, and the filing of any financing statement or document having similar effect (other than a financing statement filed by a "true" lessor pursuant to Section 9408 of the Uniform Commercial Code) naming the owner of the asset to which such Lien relates as debtor, under the Uniform Commercial Code or other comparable law of any jurisdiction.

"Loan Account" has the meaning given to such term in Section 2.3.

"<u>Loan Documents</u>" means this Agreement, the Loan Notes, the Guaranty, the Subsidiary Guaranty and all other agreements, instruments and documents (together with amendments and supplements thereto and replacements thereof) now or hereafter executed by the Borrower, which evidences or relates to the Obligations.

"Loan Notes" means the promissory notes evidencing the Loans in the aggregate original principal amount of Two Hundred Million Dollars (\$200,000,000) executed by Borrower in favor of Lenders, as they may be amended, supplemented, replaced or modified from time to time. The Loan Notes and any replacements thereof shall be substantially in the form of Exhibit  $\underline{C}$ .

"Loans" means the Loans made pursuant to the Facility.

"<u>Long Term Unsecured Indebtedness</u>" means all Unsecured Indebtedness which at the time of determination has a maturity of not less than five (5) years.

"Material Adverse Effect" means, with respect to a Person or Property, a material adverse effect upon the condition (financial or otherwise), operations, performance or properties of such Person or Property. The phrase "has a Material Adverse Effect" or "will result in a Material Adverse Effect" or words substantially similar thereto shall in all cases be intended to mean "has resulted, or will or could reasonably be anticipated to result, in a Material Adverse Effect", and the phrase "has no (or does not have a) Material Adverse Effect" or "will not result in a Material Adverse Effect" or words substantially similar thereto shall in all cases be intended to mean "does not or will not or could not reasonably be anticipated to result in a Material Adverse Effect".

"<u>Materially Defaulted Leases</u>" means Leases under which the tenant has failed to make any payment of base rent, percentage rent or additional rent when due and such failure has continued for more than ninety (90) days after the due date of the applicable payment or any Lease which the Borrower has terminated based on any default by the Tenant thereunder.

"Maturity Date" means March 31, 2005, as such date may be extended in accordance with the terms of Section 2.10.

"<u>Maximum Loan Amount</u>" means, at any time, the amount of the Facility from time to time less the aggregate face amount of Letters of Credit outstanding hereunder provided, however, that after any Event of Default has occurred and until the same shall have been remedied or waived pursuant to Section 12.4, the Maximum Loan Amount shall be zero.

"<u>Multiemployer Plan</u>" means an employee benefit plan defined in Section 4001(a)(3) of ERISA which is, or within the immediately preceding six (6) years was, contributed to by a Person or an ERISA Affiliate of such Person.

"Net Operating Income" means with respect to any Fiscal Quarter of the Borrower and with respect to any one or more of its Properties, (i) the total rental and other operating income from the operation of such Properties after deducting all expenses and other proper charges incurred by the Borrower in connection with the operation of such Properties during such Fiscal Quarter, including, without limitation, property operating expenses, real estate taxes and bad debt expenses, but before payment or provision for debt service, income taxes, and depreciation, amortization, and other non-cash expenses, all as determined in accordance with generally accepted accounting principles, minus (ii) percentage rent income of such Properties for such Fiscal Quarter, plus (iii) twenty-five percent (25%) of the total percentage rent income of such Properties during such Fiscal Quarter and the three immediately preceding Fiscal Quarters, minus (iv) the Replacement Reserve Amount for such Properties. With respect to Properties located outside of the United States, Net Operating Income shall be converted from the currency in which the applicable income and expenses are paid to Dollars using the currency exchange rates in effect as of the end of the applicable Fiscal Quarter.

"Net Offering Proceeds" means all cash proceeds received by the REIT as a result of the sale of common, preferred or other classes of stock in the REIT (if and only to the extent reflected in stockholders' equity on the consolidated balance sheet of the REIT prepared in accordance with GAAP) less customary costs and discounts of issuance paid by the REIT, all of which proceeds shall have been concurrently contributed by the REIT to Borrower as additional capital.

"Non Pro Rata Loan" means a Loan with respect to which fewer than all Lenders have funded their respective Pro Rata Shares of such Loans and the failure of the non-funding Lender or Lenders to fund its or their respective Pro Rata Shares of such Loan constitutes a breach of this Agreement.

"Non-Retail Properties" means Portfolio Properties which are not intended to be used as or in connection with a retail Property including residential or other Properties in the vicinity of Construction Projects acquired to facilitate the obtaining of governmental permits or the resolution of zoning or land use issues related to such Construction Projects.

"Nonrecourse Indebtedness" means Indebtedness with respect to which recourse for payment is contractually limited to specific assets encumbered by a Lien securing such Indebtedness.

"Notice of Borrowing" means, with respect to a proposed Borrowing pursuant to Section 2.1.2, a notice substantially in the form of Exhibit D.

"Obligations" means, from time to time, all Indebtedness of Borrower owing to Agent, any Lender or any Person entitled to indemnification pursuant to Section 12.2, or any of their respective successors, transferees or assigns, of every type and description, whether or not evidenced by any note, guaranty or other instrument, arising under or in connection with this Agreement or any other Loan Document, whether or not for the payment of money, whether direct or indirect (including those acquired by assignment), absolute or contingent, due or to become due, now existing or hereafter arising and however acquired. The term includes, without limitation, all interest, charges, expenses, fees, reasonable attorneys' fees and disbursements, reasonable fees and disbursements of expert witnesses and other consultants, and any other sum now or hereinafter chargeable to Borrower under or in connection with this Agreement or any other Loan Document.

"Officer's Certificate" means a certificate signed by a specified officer of a Person certifying as to the matters set forth therein.

"<u>Operating Cash Flow</u>" means, at any time, for the most recent Fiscal Quarter, EBITDA <u>minus</u> cash income taxes paid during such Fiscal Quarter and not deducted on the Financial Statements in determining earnings for such Fiscal Quarter or any prior period <u>minus</u> the Replacement Reserve Amount for the Portfolio Properties.

"Other Properties" means Properties other than Premium Centers or Other Retail Centers that are so classified in Borrower's filings with the Commission.

"Other Properties Net Operating Income" means, with respect to any Fiscal Quarter of the Borrower, the sum of (i) the Net Operating Income of all Other Properties owned by Borrower or a Wholly-Owned Subsidiary and (ii) the Borrower's Share of the Net Operating Income of all Other Properties owned by Affiliates of Borrower that are not Wholly-Owned Subsidiaries.

"Other Retail Centers" means Properties other than Premium Centers or Other Properties that are so classified in Borrower's filings with the Commission.

"Other Retail Centers Net Operating Income" means, with respect to any Fiscal Quarter of the Borrower, the sum of (i) the Net Operating Income of all Other Retail Centers owned by Borrower or a Wholly-Owned Subsidiary and (ii) the Borrower's Share of the Net Operating Income of all Other Retail Centers owned by Affiliates of Borrower that are not Wholly-Owned Subsidiaries.

"PBGC" means the Pension Benefit Guaranty Corporation or any Person succeeding to the functions thereof.

"<u>Permit</u>" means any permit, approval, authorization, license, variance or permission required from a Governmental Authority under an applicable Requirement of Law.

"<u>Permitted Investments</u>" means Investments in Land, Non-Retail Properties, Investment Mortgages and the categories of Investments set forth in <u>Section 9.10</u>.

#### "Permitted Liens" means:

- (a) Liens (other than Environmental Liens and any Lien imposed under ERISA) for taxes, assessments or charges of any Governmental Authority or claims not yet due or not yet required to be paid pursuant to Section 7.4;
- (b) Liens (other than any Lien imposed under ERISA) incurred or deposits made in the ordinary course of business (including without limitation surety bonds and appeal bonds) in connection with workers' compensation, unemployment insurance and other types of social security benefits or to secure the performance of tenders, bids, leases, contracts (other than for the repayment of Indebtedness), statutory obligations;
- (c) any laws, ordinances, easements, rights of way, restrictions, exemptions, reservations, conditions, defects or irregularities in title, limitations, covenants or other matters that, in the aggregate, do not in the reasonable opinion of Borrower (i) materially interfere with the occupation, use and enjoyment of the Property or other assets encumbered thereby, by the Person owning such Property or other assets, in the normal course of its business or (ii) materially impair the value of the Property subject thereto;
- (d) Liens imposed by laws, such as mechanics' liens and other similar liens arising in the ordinary course of business which either (i) have been in existence for less than 120 days from the date of filing or (ii) have been in existence for 120 days or longer so long as the aggregate amount of all such Liens is less than

\$100,000 for each Construction Project and the Borrower is in good faith contesting the validity or amount thereof by appropriate proceedings, provided however that any Lien permitted under this paragraph must be discharged prior to foreclosure thereof;

- (e) Leases to tenants which are not Affiliates of Borrower existing on the date hereof or subsequently entered into in the ordinary course of business; and
- (f) Liens securing judgments or awards that have been in force for less than the applicable period for taking an appeal so long as execution is not levied thereunder or in respect of which the Borrower shall at the time in good faith be prosecuting an appeal or proceedings for review and in respect of which a stay of execution shall have been obtained pending such appeal or review.

"<u>Person</u>" means any natural person, employee, corporation, limited partnership, general partnership, joint stock company, limited liability company, joint venture, association, company, trust, bank, trust company, land trust, business trust or other organization, whether or not a legal entity, or any other non-governmental entity, or any Governmental Authority.

"<u>Plan</u>" means an employee benefit plan defined in Section 3(3) of ERISA (other than a Multiemployer Plan) in respect of which Borrower or an ERISA Affiliate, as applicable, is an "employer" as defined in Section 3(5) of ERISA.

"<u>Portfolio Properties</u>" means real property improved with one or more completed buildings in which more than a fifty percent (50%) interest is owned, directly or indirectly, by Borrower or any Subsidiary of Borrower, including the Unencumbered Properties and the Properties listed on <u>Schedule 1.1</u>, as such schedule may be updated from time to time to reflect the acquisition or disposition of Portfolio Properties. <u>Schedule 1.1</u> also indicates which of the Portfolio Properties are currently classified as Premium Centers and which are currently classified as Other Retail Centers or Other Properties.

"<u>Premium Centers</u>" means Properties that satisfy Borrower's criteria for classification as part of its Premium Portfolio, that are publicly advertised as "Premium Outlets" or a similar term and that are classified as Premium Centers in Borrower's filings with the Commission.

"Premium Centers Net Operating Income" means, with respect to any Fiscal Quarter of the Borrower, the sum of (i) the Net Operating Income of all Premium Centers owned by Borrower or a Wholly-Owned Subsidiary and (ii) the Borrower's Share of the Net Operating Income of all Premium Centers owned by Affiliates of Borrower that are not Wholly-Owned Subsidiaries.

"Prepayment Date" has the meaning given to such term in Section 2.4.8(c).

"Pro Forma Unsecured Debt Service Charges" means, for any Fiscal Quarter of the Borrower, the sum of (a) an amount determined by the Borrower (and approved by the Agent in its sole discretion) based on a twenty-five (25) year mortgage style amortization schedule, calculated on the outstanding principal amount of all Unsecured Indebtedness as of the end of such Fiscal Quarter excluding Long Term Unsecured Indebtedness and an interest rate equal to the greater of (i) the weighted average annual interest rate actually applicable to all Unsecured Indebtedness excluding Long Term Unsecured Indebtedness during such Fiscal Quarter or (ii) the then current ten (10) year U.S. Treasury bill yield plus one and three-quarters percent (1.75%) plus (b) one-quarter of the actual debt service charges due during the current fiscal year pursuant to the Long Term Unsecured Indebtedness.

"<u>Pro Rata Share</u>" means, with respect to any Lender, a fraction (expressed as a percentage), the numerator of which shall be the amount of such Lender's Commitment and the denominator of which shall be the aggregate amount of all of the Lenders' Commitments.

"<u>Proceedings</u>" means, collectively, all actions, suits and proceedings before, and investigations commenced or threatened by or before, any court or Governmental Authority with respect to a Person.

"Property" means, as to any Person, all real or personal property (including, without limitation, buildings, facilities, structures, equipment and other assets, tangible or intangible) owned by such Person.

"Rating Agency" means either of (i) Standard & Poor's, a division of The McGraw-Hill Companies, Inc. or (ii) Moody's Investors Services, Inc.

"Regulations T, U and X" mean such Regulations of the Federal Reserve Board as in effect from time to time.

"<u>REIT</u>" means Chelsea Property Group, Inc., a Maryland corporation that was formerly known as Chelsea GCA Realty, Inc.

"Release" means the release, spill, emission, leaking, pumping, injection, deposit, disposal, discharge, dispersal, leaching or migration into the indoor or outdoor environment or into or out of any property, including the movement of Contaminants through or in the air, soil, surface water, groundwater or property.

"Remedial Action" means any action required by applicable Environmental Laws to (i) clean up, remove, treat or in any other way address Contaminants in the indoor or outdoor environment; (ii) prevent the Release or threat of Release or minimize the further Release of Contaminants so they do not migrate or endanger or threaten to endanger public health or welfare

or the indoor or outdoor environment; or (iii) perform preremedial studies and investigations and post-remedial monitoring and care

"Rent Stabilization Date" means, with respect to each Construction Project, the date which shall be (i) the first day of a Fiscal Quarter (ii) not more than six (6) months after the date on which the first certificate of occupancy (or the equivalent) has been issued for any portion of such Construction Project and (iii) set forth in a notice from Borrower to Agent given prior to such Rent Stabilization Date.

"Replacement Reserve Amount" means, with respect to any Property or group of Properties for any Fiscal Quarter, a reserve for replacement reserves, leasing costs and recurring capital expenditures equal to the product of \$0.075 times the gross leasable area of such Property or group of Properties excluding the gross leasable area of any Construction Project thereon prior to the Rent Stabilization Date with respect to such Construction Project. The Replacement Reserve Amount for the Portfolio Properties will be based on the gross leasable area of the Properties owned by Borrower or one of its Subsidiaries and the Borrower's Share of the gross leasable area of the Properties owned by the Unconsolidated Affiliates.

"Reportable Event" means any of the events described in Section 4043(b) of ERISA, other than an event for which the thirty (30) day notice requirement is waived by regulations.

"Requirements of Law" mean, as to any Person, the charter and by-laws, partnership agreement or other organizational or governing documents of such Person, and any law, rule or regulation, Permit, or determination of an arbitrator or a court or other Governmental Authority, in each case applicable to or binding upon such Person or any of its property or to which such Person or any of its property is subject, including without limitation, the Securities Act, the Securities Exchange Act, Regulations T, U and X, FIRREA and any certificate of occupancy, zoning ordinance, building, environmental or land use requirement or Permit or occupational safety or health law, rule or regulation.

"Requisite Lenders" mean, collectively, Lenders whose Pro Rata Shares, in the aggregate, are at least sixty-six and two-thirds percent (66-2/3%), provided that, in determining such percentage at any given time, all then existing Defaulting Lenders will be disregarded and excluded and the Pro Rata Shares of Lenders shall be redetermined, for voting purposes only, to exclude the Pro Rata Shares of such Defaulting Lenders and provided, further, that the Agent must always be among the Requisite Lenders except that after an Event of Default described in Section 10.1.1 decisions by the Requisite Lenders to accelerate and/or exercise remedies pursuant to Section 10.2.1 shall be made without regard to whether the Agent is among the Requisite Lenders.

"Revolving Loans" means the Revolving Loans made pursuant to the Facility under Section 2.1.1.

"Secured Borrower Debt" means all Borrower Debt that is secured by a Lien on any Property.

"Securities" means any stock, shares, voting trust certificates, bonds, debentures, notes or other evidences of indebtedness, secured or unsecured, convertible, subordinated or otherwise, or in general any instruments commonly known as "securities", or any certificates of interest, shares, or participations in temporary or interim certificates for the purchase or acquisition of, or any right to subscribe to, purchase or acquire any of the foregoing, but shall not include any evidence of the Obligations, provided that Securities shall not include Cash Equivalents, Investment Mortgages or interests in Unconsolidated Affiliates.

"Securities Act" means the Securities Act of 1933, as amended to the date hereof and from time to time hereafter, and any successor statute.

"Securities Exchange Act" means the Securities Exchange Act of 1934, as amended to the date hereof and from time to time hereafter, and any successor statute.

"Senior Loans" has the meaning given to such term in Section 11.4.2.

"Solvent" means, as to any Person at the time of determination, that such Person (i) owns property the value of which (both at fair valuation and at present fair saleable value) is greater than the amount required to pay all of such Person's liabilities (including contingent liabilities and debts); (ii) is able to pay all of its debts as such debts mature; and (iii) has capital sufficient to carry on its business and transactions and all business and transactions in which it is about to engage.

"Subsidiary" of a Person means any corporation, limited liability company, partnership, trust or other entity of which a majority of the stock (or equivalent ownership or controlling interest) having voting power to elect a majority of the Board of Directors (if a corporation) or to select the trustee or equivalent controlling interest, shall, at the time such reference becomes operative, be directly or indirectly owned or controlled by such Person.

"Subsidiary Guaranty" means the Guaranty of even date herewith in favor of the Agent and the Lenders executed by the Guarantor Subsidiaries listed on Schedule 5.5 and each Guaranty in substantially the same form that is hereafter executed and delivered by other Wholly-Owned Subsidiaries.

"Taxes" means all federal, state, local and foreign income and gross receipts taxes.

"Termination Date" has the meaning given to such term in Section 2.1.4.

"Termination Event" means (i) any Reportable Event, (ii) the withdrawal of a Person, or an ERISA Affiliate from a Benefit Plan during a plan year in which it was a "substantial employer" as defined in Section 4001(a)(2) of ERISA, (iii) the occurrence of an obligation arising under Section 4041 of ERISA of a Person or an ERISA Affiliate to provide affected parties with a written notice of an intent to terminate a Benefit Plan in a distress termination described in Section 4041(c) of ERISA, (iv) the institution by the PBGC of proceedings to terminate any Benefit Plan under Section 4042 of ERISA, (v) any event or condition which constitutes grounds under Section 4042 of ERISA for the appointment of a trustee to administer a Benefit Plan, (vi) the partial or complete withdrawal of such Person or any ERISA Affiliate from a Multiemployer Plan, or (vii) the adoption of an amendment by any Person or any ERISA Affiliate to terminate any Benefit Plan.

"<u>Total Liabilities</u>" means without duplication (i) all Indebtedness of Borrower and its Subsidiaries, whether or not such Indebtedness would be included as a liability on the balance sheet of Borrower in accordance with GAAP, plus (ii) all other liabilities of every nature and kind of Borrower and its Subsidiaries that would be included as liabilities on the balance sheet of Borrower in accordance with GAAP, plus (iii) the Borrower's Share of all Indebtedness of all Unconsolidated Affiliates.

"<u>Unconsolidated Affiliate</u>" means any Affiliate in which Borrower or any Subsidiary of Borrower has an ownership interest, whose financial results are not consolidated under GAAP in the Financial Statements.

"<u>Unencumbered Net Operating Income</u>" means, with respect to any Fiscal Quarter of the Borrower, the sum of the Net Operating Income of all of its Properties which were Unencumbered Properties hereunder during such Fiscal Quarter.

"<u>Unencumbered Property</u>" means a Property which at the date of determination, (i) is owned by Borrower, by a Guarantor Subsidiary or by an Eligible Affiliate, (ii) is classified as either a Premium Center or an Other Retail Center in Borrower's filings with the Commission; (iii) is not directly or indirectly subject to any Lien (other than Permitted Liens) or to any negative pledge agreement or other agreement (other than this Agreement) that prohibits the creation of any Lien thereon; (iv) is a Property with respect to which each of the representations contained in <u>Section 5.22</u> and in <u>Section 5.23</u> hereof is true and accurate as of such date of determination; (v) may be legally conveyed separately from any other Real Estate without the need to obtain any subdivision approval, zoning variance or other consent or approval from an unrelated Person; (vi) is located in the United States, and (vii) to the extent requested by the Agent, the Borrower has delivered to the Agent historical operating and leasing information relating to such Unencumbered Property, in form and substance satisfactory to the Agent.

"<u>Unencumbered Property Value</u>" means, with respect to any Unencumbered Property at any time, an amount computed as follows: (a) the Net Operating Income of such Unencumbered Property for the most recent Fiscal Quarter for which financial statements have been delivered to the Agent pursuant to <u>Section 6.1</u>; (b) then multiplying by four (4); and (c) dividing such product by either 0.09 (if such Unencumbered Property is classified as a Premium Center) or 0.12 (if such Unencumbered Property is classified as an Other Retail Center). With respect to any Unencumbered Property which, during the applicable Fiscal Quarter, has been acquired by Borrower or has had the building or buildings being constructed thereon completed and occupied by tenants, Borrower may compute the Unencumbered Property Value for such Unencumbered Property based on a pro forma Net Operating Income for such Fiscal Quarter, which computation must be approved by the Agent.

"<u>Unmatured Event of Default</u>" means an event which, with the giving of notice or the lapse of time, or both, would constitute an Event of Default.

"<u>Unsecured Indebtedness</u>" means all Indebtedness of Borrower or of any of its Subsidiaries which is not secured by a Lien on any Properties including, without limitation, the Loans, the Borrower's reimbursement obligations relating to the Letters of Credit, the Fleet Term Loan, the Unsecured Term Notes and any Indebtedness evidenced by any bonds, debentures, notes or other debt securities which may be hereafter issued by Borrower or by the REIT. Unsecured Indebtedness shall not include accrued ordinary operating expenses payable on a current basis.

"<u>Unsecured Term Note Indenture</u>" means the Indenture dated as of January 23, 1996 among the Borrower, the REIT and State Street Bank and Trust Company, as trustee, as supplemented through the date hereof, and as hereafter amended or further supplemented.

"<u>Unsecured Term Note Secured Debt Limitation</u>" means the provision contained in the Unsecured Term Note Indenture which limits the Borrower's "Secured Debt" to not more than 40% of its "Adjusted Total Assets" (as such quoted terms are defined in said Indenture). If the Unsecured Term Note Indenture is amended to reduce the permitted amount of Secured Borrower Debt this defined term shall automatically be deemed to incorporate such amendment, but if the Unsecured Term Note Indenture is amended to increase the permitted amount of Secured Borrower Debt this definition shall not be deemed to incorporate such amendment until the same has been approved by the Requisite Lenders.

"<u>Unsecured Term Notes</u>" means, collectively, all notes from time to time outstanding under the Unsecured Term Note Indenture.

"<u>Value of All Other Properties</u>" means, when determined as of the end of a Fiscal Quarter, an amount computed as follows: (a) Other Properties Net Operating Income; (b) then multiplying by four (4); and (c) dividing such product by 0.12.

"Value of All Other Retail Centers" means, when determined as of the end of a Fiscal Quarter, an amount computed as follows: (a) Other Retail Centers Net Operating Income; (b) then multiplying by four (4); and (c) dividing such product by 0.12.

"<u>Value of All Premium Centers</u>" means, when determined as of the end of a Fiscal Quarter, an amount computed as follows: (a) Premium Centers Net Operating Income; (b) then multiplying by four (4); and (c) dividing such product by 0.09.

"Value of All Unencumbered Properties" means, when determined as of the end of a Fiscal Quarter, an amount equal to the sum of the Unencumbered Property Value of each Unencumbered Property owned by Borrower or by a Guarantor Subsidiary for such Fiscal Quarter, plus the Borrower's Share of the Unencumbered Property Value of each Unencumbered Property owned by an Eligible Affiliate, provided, however, that (i) to the extent that more than ten percent (10%) of such number is attributable to Unencumbered Properties that are owned by Eligible Affiliates, the amount in excess of said 10% shall be excluded from the Value of All Unencumbered Properties, and (ii) to the extent that more than five percent (5%) of such number is attributable to Unencumbered Properties that are not on land owned in fee by Borrower, a Guarantor Subsidiary or an Eligible Affiliate or that are subject to a Capital Lease or a ground lease, the amount in excess of said 5% shall be excluded from the Value of All Unencumbered Properties. When determined as of a date which is during a Fiscal Quarter based on an updated list of Unencumbered Properties attached to the applicable Compliance Certificate, the Value of All Unencumbered Properties most recently computed as provided in the preceding sentence of this definition will be adjusted by subtracting the Unencumbered Property Value of the previous Unencumbered Properties which have been deleted from such list and by adding the Unencumbered Property Value of the Unencumbered Properties which have been added to such list.

"<u>Wholly-Owned Subsidiary</u>" means a Subsidiary which is 100% owned directly or indirectly by Borrower or a Wholly-Owned Subsidiary of Borrower.

1.2 <u>Computation of Time Periods</u>. In this Agreement, in the computation of periods of time from a specified date to a later specified date, the word "from" means "from and including" and the words "to" and "until" each mean "to and including". Periods of days referred to in this Agreement shall be counted in calendar days unless Business Days are expressly prescribed.

#### 1.3 Terms.

- 1.3.1 Any accounting terms used in this Agreement which are not specifically defined shall have the meanings customarily given them in accordance with GAAP. All references herein to Borrower, the REIT or any other Person, in connection with any financial or related covenant, representation or calculation, shall be understood to mean and refer to Borrower, the REIT and such other Person on a consolidated basis in accordance with GAAP, unless otherwise specifically provided and subject in all events to any adjustments herein set forth.
- 1.3.2 Any time the phrase "to the best of Borrower's knowledge" or a phrase similar thereto is used herein, it means: "to the actual knowledge of the then executive or senior officers of Borrower and the REIT, after reasonable inquiry of those agents, employees or contractors of the REIT or Borrower who could reasonably be anticipated to have knowledge with respect to the subject matter or circumstances in question and after review of those documents or instruments which could reasonably be anticipated to be relevant to the subject matter or circumstances in question provided that such reasonable inquiry need not be undertaken at the time of each Compliance Certificate."
- 1.3.3 In each case where the consent or approval of Agent, all Lenders and/or Requisite Lenders is required, or their non-obligatory action is requested by Borrower, such consent, approval or action shall be in the sole and absolute discretion of Agent and, as applicable, each Lender, unless otherwise specifically indicated.
- 1.3.4 Any time the word "or" is used herein, unless the context otherwise clearly requires, it has the inclusive meaning represented by the phrase "and/or". The words "hereof", "herein", "hereby", "hereunder" and similar terms refer to this Agreement as a whole and not to any particular provision of this Agreement. Article, section, subsection, clause, exhibit and schedule references are to this Agreement unless otherwise specified. Any reference in this Agreement to this Agreement or to any other Loan Document includes any and all amendments, modifications, supplements, renewals or restatements thereto or thereof, as applicable.

#### **ARTICLE II**

#### LOANS AND LETTERS OF CREDIT

#### 2.1 <u>Loan Advances and Repayment</u>.

2.1.1 <u>Maximum Loan Amount</u>. Subject to the terms and conditions set forth in this Agreement, Lenders hereby agree to make Loans to Borrower from time to time during the period from the Closing Date to the Business Day next preceding the Termination Date, in an aggregate outstanding principal amount which shall not exceed the Maximum Loan Amount at any time. All Loans under this Agreement shall be made by Lenders simultaneously and proportionately to their respective Pro Rata Shares, it being understood that no Lender shall be responsible for any failure by any other Lender to perform its obligation to make a Loan hereunder and that the Commitment of any Lender shall not be increased or decreased as a result of the failure by any other Lender to perform its obligation to make a Loan. Loans may be voluntarily prepaid pursuant to <u>Section 2.6.1</u> and, subject to the provisions of this Agreement, any amounts so prepaid may be reborrowed under this <u>Section 2.1.1</u>. The principal balance of the Loans shall be payable in full on the Termination Date. The Loans will be evidenced by the Loan Notes.

#### 2.1.2 <u>Notice of Borrowing.</u>

(a) (i) Whenever Borrower desires to borrow under this <u>Section 2.1</u>, but in no event more

than three (3) times during any one (1) calendar month, Borrower shall give Agent, at 115 Perimeter Center Place, N.E., Suite 500, Atlanta, GA 30346, Attn: Lori Y. Litow (Fax No. (770)390-8434) or such other address as Agent shall designate, an original or facsimile Notice of Borrowing no later than 9:00 A.M. (Eastern time), not less than three (3) nor more than five (5) Business Days prior to the proposed Funding Date of each Loan. The Agent shall notify each Lender by telephone or facsimile with regard to each Notice of Borrowing not later than 11:00 A.M. (Eastern Time) on the second Business Day preceding the proposed Funding Date.

- (ii) Notwithstanding the foregoing or any other provision hereof to the contrary a Notice of Borrowing may be given not less than two (2) Business Days prior to the proposed Funding Date of a Loan if the additional Borrowing shall be requested as a Base Rate Loan.
- (iii) Each Notice of Borrowing shall specify (1) the Funding Date (which shall be a Business Day) in respect of the Loan, (2) the amount of the proposed Loan, <u>provided</u> that the aggregate amount of such proposed Loan shall equal Four Million Dollars (\$4,000,000) or integral multiples of One Million Dollars (\$1,000,000) in excess thereof, (3) whether the Loan to be made thereunder will be a Base Rate Loan or a LIBOR Loan and, if a LIBOR Loan, the Interest Period, (4) to which account of Borrower the funds are to be directed, and (5) the proposed use of such Loan. Any Notice of Borrowing pursuant to this <u>Section 2.1.2</u> shall be irrevocable.
- (b) Borrower may elect (i) to convert LIBOR Loans or any portion thereof into Base Rate Loans, (ii) to convert Base Rate Loans or any portion thereof to LIBOR Loans, or (iii) to continue any LIBOR Loans or any portion thereof for an additional Interest Period, provided, however, that the aggregate amount of the Loans being converted into or continued as LIBOR Loans shall equal Four Million Dollars (\$4,000,000) or an integral multiple of One Million Dollars (\$1,000,000) in excess thereof. The applicable Interest Period for the continuation of any LIBOR Loan shall commence on the day on which the next preceding Interest Period expires. The conversion of a LIBOR Loan to a Base Rate Loan shall only occur on the last Business Day of the Interest Period relating to such LIBOR Loan; such conversion shall occur automatically in the absence of an election under clause (iii) above. Each election under clause (iii) above shall be made by Borrower giving Agent an original or facsimile Notice of Borrowing no later than 9:00 A.M. (Eastern time), not less than three (3) nor more than five (5) Business Days prior to the date of a conversion to or continuation of a LIBOR Loan, specifying, in each case (1) the amount of the conversion or continuation, (2) the Interest Period therefor, and (3) the date of the conversion or continuation (which date shall be a Business Day).
- (c) Upon receipt of a Notice of Borrowing in proper form requesting LIBOR Loans under <u>subparagraph</u> (a) or (b) above, Agent shall determine the LIBOR applicable to the Interest Period for such LIBOR Loans, and shall, prior to the beginning of such Interest Period, give (by facsimile) a <u>Fixed Rate Notice</u> in respect thereof to Borrower and Lenders; <u>provided</u>, <u>however</u>, that failure to give such notice to Borrower shall not affect the validity of such rate. Each determination by Agent of the LIBOR shall be conclusive and binding upon the parties hereto in the absence of manifest error.
- 2.1.3 <u>Making of Loans</u>. Subject to <u>Section 11.3</u>, Agent shall make the proceeds of Loans available to Borrower on such Funding Date and shall disburse such funds in Dollars in immediately available funds to Borrower's commercial demand account at Fleet or such other account specified in the Notice of Borrowing acceptable to Agent.
- 2.1.4. <u>Term.</u> The outstanding balance of the Loans shall be payable in full on the earliest to occur of (i) the Maturity Date, or, if the Maturity Date has been extended pursuant to <u>Section 2.10</u>, the Maturity Date as so extended, (ii) the acceleration of the Loans pursuant to <u>Section 10.2.1</u>, or (iii) Borrower's written notice to Agent (pursuant to <u>Section 2.6.1</u>) of Borrower's election to prepay all accrued obligations and terminate all Commitments (the "<u>Termination Date</u>").
- 2.2 <u>Authorization to Obtain Loans</u>. Each of Borrower's Chief Executive Officer, President, Chief Financial Officer, and Treasurer are hereby authorized by Borrower to sign Notices of Borrowing and Letter of Credit Requests. Borrower may provide Agent with documentation satisfactory to Agent indicating the names of other officers or employees of Borrower authorized by Borrower to sign Notices of Borrowing and Letter of Credit Requests, and Agent and Lenders shall be entitled to rely on such documentation until notified in writing by Borrower of any change(s) of the persons so authorized. Agent shall be entitled to act on the instructions of anyone identifying himself or herself as one of the Persons authorized to execute a Notice of Borrowing or Letter of Credit Request, and Borrower shall be bound thereby in the same manner as if such Person were actually so authorized. Borrower agrees to indemnify, defend and hold Lenders and Agent harmless from and against any and all Liabilities and Costs which may arise or be created by the acceptance of instructions in any Notice of Borrowing or Letter of Credit Request, unless caused by the gross negligence of the Person to be indemnified.
- 2.3 <u>Lenders' Accounting</u>. Agent shall maintain a loan account (the "<u>Loan Account</u>") on its books in which shall be recorded (i) the names and addresses and the Commitments of Lenders, and principal amount of Loans owing to each Lender from time to time, and (ii) all advances and repayments of principal and payments of accrued interest under the Loans, as well as payments of the Facility Fee and the Letter of Credit fee, as provided in this Agreement.

#### 2.4 <u>Interest on the Loans</u>.

2.4.1 <u>Base Rate Loans</u>. Subject to <u>Section 2.4.4</u>, all Base Rate Loans shall bear interest on the average daily unpaid principal amount thereof from the date made until paid in full at a fluctuating rate per annum equal to the Base Rate. Base Rate Loans shall be made in minimum amounts of Four Million Dollars (\$4,000,000) or an integral multiple of One Million (\$1,000,000) in excess thereof.

- 2.4.2 <u>LIBOR Loans</u>. Subject to <u>Sections 2.4.4</u> and <u>2.4.8</u>, all LIBOR Loans shall bear interest on the unpaid principal amount thereof during the Interest Period applicable thereto at a rate per annum equal to the sum of LIBOR for such Interest Period <u>plus</u> the Applicable LIBOR Rate Margin. LIBOR Loans shall be in tranches of Four Million Dollars (\$4,000,000) or One Million Dollar (\$1,000,000) increments in excess thereof. No more than eight (8) LIBOR Loan tranches shall be outstanding at any one time, of which no more than two (2) LIBOR Loan tranches with an Interest Period of seven (7) days shall be outstanding at any one time. Notwithstanding anything to the contrary contained herein and subject to the Default Interest provisions contained in <u>Section 2.4.4</u>, if an Event of Default occurs and as a result thereof the Commitments are terminated, all LIBOR Loans will convert to Base Rate Loans upon the expiration of the applicable Interest Periods therefor or the date all Loans become due, whichever occurs first.
- 2.4.3 <u>Interest Payments</u>. Subject to <u>Section 2.4.4</u>, interest accrued on all Loans shall be payable by Borrower, in the manner provided in <u>Section 2.6.2</u>, in arrears on the first Business Day of the first calendar month following the Closing Date, the first Business Day of each succeeding calendar month thereafter, and on the Termination Date.
- 2.4.4 <u>Default Interest</u>. Notwithstanding the rates of interest specified in <u>Sections 2.4.1</u> and <u>2.4.2</u> and the payment dates specified in <u>Section 2.4.3</u>, effective immediately upon the occurrence and during the continuance of any Event of Default, the principal balance of all Loans then outstanding and, to the extent permitted by applicable law, any interest payments on the Loans not paid when due shall bear interest payable upon demand at a rate which is four percent (4%) per annum in excess of the Base Rate. All other amounts due Agent or Lenders (whether directly or for reimbursement) under this Agreement or any of the other Loan Documents if not paid when due, or if no time period is expressed, if not paid within thirty (30) days after demand, shall bear interest from and after demand at the rate set forth in this <u>Section 2.4.4</u>.
- 2.4.5 <u>Late Fee.</u> Borrower acknowledges that late payment to Agent will cause Agent and Lenders to incur costs not contemplated by this Agreement. Such costs include, without limitation, processing and accounting charges. Therefore, if Borrower fails timely to pay any sum due and payable hereunder through the Termination Date, unless waived by Agent pursuant to <u>Section 11.11.1</u> or Requisite Lenders, a late charge of four cents (\$.04) for each dollar of any principal payment, interest or other charge due hereon and which is not paid within ten (10) days after such payment is due, shall be charged by Agent (for the benefit of Lenders) and paid by Borrower for the purpose of defraying the expense incident to handling such delinquent payment; <u>provided</u>, <u>however</u>, that no late charges shall be assessed with respect to any period during which Borrower is obligated to pay interest at the rate specified in <u>Section 2.4.4</u>, or in respect of any failure to pay all Obligations on the Termination Date. Borrower and Agent agree that this late charge represents a reasonable sum considering all of the circumstances existing on the date hereof and represents a fair and reasonable estimate of the costs that Agent and Lenders will incur by reason of late payment. Borrower and Agent further agree that proof of actual damages would be costly and inconvenient. Acceptance of any late charge shall not constitute a waiver of the default with respect to the overdue installment, and shall not prevent Agent from exercising any of the other rights available hereunder or any other Loan Document. Such late charge shall be paid without prejudice to any other rights of Agent.
- 2.4.6 <u>Computation of Interest</u>. Interest shall be computed on the basis of the actual number of days elapsed in the period during which interest or fees accrue and a year of three hundred sixty (360) days. In computing interest on any Loan, the date of the making of the Loan shall be included and the date of payment shall be excluded; <u>provided</u>, <u>however</u>, that if a Loan is repaid on the same day on which it is made, one (1) day's interest shall be paid on that Loan. Changes in the Applicable LIBOR Rate Margin shall take effect as of the date on which the condition set forth in the relevant clause of the definition of each such term is satisfied. If, at any time, the rate of interest, together with all amounts which constitute interest and which are reserved, charged or taken by the Lenders as compensation for fees, services or expenses incidental to the making, negotiating or collection of the Loans or the other Obligations, shall he deemed by any competent court of law, governmental agency or tribunal to exceed the maximum rate of interest permitted to be charged by any Lender to Borrower under applicable law, then, during such time as such rate of interest would be deemed excessive, that portion of each sum paid attributable to that portion of such interest rate that exceeds the maximum rate of interest so permitted shall be deemed a voluntary prepayment of principal. As used herein, the term "applicable law" shall mean the law in effect as of the date hereof, provided, however, that in the event there is a change in the law which results in a higher permissible rate of interest, then this Agreement and the Notes shall be governed by such new law as of its effective date.
- 2.4.7 <u>Changes; Legal Restrictions</u>. In the event that after the Closing Date (i) the adoption of or any change in any law, treaty, rule, regulation, guideline or determination of a court or Governmental Authority or any change in the interpretation or application thereof by a court or Governmental Authority, or (ii) compliance by Agent or any Lender with any request or directive made or issued after the Closing Date from any central bank or other Governmental Authority or quasi-governmental authority:
  - (a) subjects Agent or any Lender to any tax, duty or other charge of any kind with respect to the Facility, this Agreement or any of the other Loan Documents or the Loans, or changes the basis of taxation of payments to Agent or such Lender of principal, fees, interest or any other amount payable hereunder, except for net income, gross receipts, gross profits or franchise taxes imposed by any jurisdiction and not specifically based upon loan transactions (all such non-excepted taxes, duties and other charges being hereinafter referred to as "Lender Taxes");
  - (b) imposes, modifies or holds applicable, in the determination of Agent or any Lender, any reserve, special deposit, compulsory loan, FDIC insurance, capital allocation or similar requirement (other than a requirement of the type described in <u>Section 2.7</u>) against assets held by, or deposits or other liabilities in or for the account of, advances or loans by, or other credit extended by, or any other acquisition of funds by, Agent or such Lender or any applicable lending office; or

(c) imposes on Agent or any Lender any other condition (<u>other than one described in Section 2.7</u>) materially more burdensome in nature, extent or consequence than those in existence as of the Closing Date,

and the result of any of the foregoing is to increase the cost to Agent or any Lender of making, renewing, maintaining or participating in the Loans or to reduce any amount receivable thereunder; then, in any such case, Borrower shall promptly pay to Agent or such Lender, as applicable, upon demand, such amount or amounts (based upon a reasonable allocation thereof by Agent or such Lender to the financing transactions contemplated by this Agreement and affected by this Section 2.4.7) as may be necessary to compensate Agent or such Lender for any such additional cost incurred or reduced amounts received; provided, however, that if the payment of such compensation may not be legally made whether by modification of the applicable interest rate or otherwise, then Lenders shall have no further obligation to make Loans that cause Agent or any Lender to incur such increased cost, and all affected Loans shall become immediately due and payable by Borrower. Agent or such Lender shall deliver to Borrower and in the case of a delivery by Lender, such Lender shall also deliver to Agent, a written statement of the claimed additional costs incurred or reduced amounts received and the basis therefor as soon as reasonably practicable after such Lender obtains knowledge thereof. If Agent or any Lender subsequently recovers any amount of Lender Taxes previously paid by Borrower pursuant to this Section 2.4.7, whether before or after termination of this Agreement, then, upon receipt of good funds with respect to such recovery, Agent or such Lender will refund such amount to Borrower if no Event of Default or Unmatured Event of Default then exists, such amount will be credited to the Obligations in the manner determined by Agent or such Lender.

### 2.4.8 <u>Certain Provisions Regarding LIBOR Loans.</u>

- (a) <u>LIBOR Lending Unlawful</u>. If any Lender shall determine (which determination shall, upon notice thereof to Borrower and Agent, be conclusive and binding on the parties hereto) that the introduction of or any change in or in the interpretation of any law makes it unlawful, or any central bank or other Governmental Authority asserts that it is unlawful, for such Lender to make or maintain any Loan as a LIBOR Loan, (i) the obligations of such Lenders to make or maintain any Loans as LIBOR Loans shall, upon such determination, forthwith be suspended until such Lender shall notify Agent that the circumstances causing such suspension no longer exist, and (ii) if required by such law or assertion, the LIBOR Loans of such Lender shall automatically convert into Base Rate Loans in which case no LIBOR Prepayment Fee shall be due upon such conversion.
- (b) <u>Deposits Unavailable</u>. If Agent shall have determined in good faith that adequate means do not exist for ascertaining the interest rate applicable hereunder to LIBOR Loans, then, upon notice from Agent to Borrower the obligations of all Lenders to make or maintain Loans as LIBOR Loans shall forthwith be suspended until Agent shall notify Borrower that the circumstances causing such suspension no longer exist. Agent will give such notice when it determines, in good faith, that such circumstances no longer exist; <u>provided</u>, <u>however</u>, that Agent shall not have any liability to any Person with respect to any delay in giving such notice.
- (c) LIBOR Prepayment Fee . Borrower acknowledges that prepayment or acceleration of a LIBOR Loan during an Interest Period shall result in Lenders incurring additional costs, expenses and/or liabilities and that it is extremely difficult and impractical to ascertain the extent of such costs, expenses and/or liabilities. (For all purposes of this subparagraph (c), any Loan not being made as a LIBOR Loan in accordance with the Notice of Borrowing therefor, as a result of Borrower's cancellation thereof, shall be treated as if such LIBOR Loan had been prepaid.) Therefore, on the date a LIBOR Loan is prepaid or the date all sums payable hereunder become due and payable, by acceleration or otherwise ("Prepayment Date"), Borrower will pay to Agent, for the account of each Lender, (in addition to all other sums then owing), an amount ("LIBOR Prepayment Fee") determined by the Agent as follows: The current rate for United States Treasury securities (bills on a discounted basis shall be converted to a bond equivalent) with a maturity date closest to the end of the Interest Period as to which prepayment is made, shall be subtracted from the interest rate applicable to the LIBOR Loan being prepaid. If the result is zero or a negative number, there shall be no LIBOR Prepayment Fee. If the result is a positive number, then the resulting percentage shall be multiplied by the amount of the prepayment of the LIBOR Loan. The resulting amount shall be divided by 360 and multiplied by the number of days remaining in the Interest Period as to which the prepayment is being made. The resulting amount shall be the LIBOR Prepayment Fee.
- (d) Upon the written notice to Borrower from Agent, Borrower shall immediately pay to Agent, for the account of Lenders, the LIBOR Prepayment Fee. Such written notice (which shall include calculations in reasonable detail) shall, in the absence of manifest error, be conclusive and binding on the parties hereto.
- (e) Borrower understands, agrees and acknowledges the following: (i) no Lender has any obligation to purchase, sell and/or match funds in connection with the use of LIBOR as a basis for calculating the rate of interest on a LIBOR Loan; (ii) LIBOR is used merely as a reference in determining such rate; and (iii) Borrower has accepted LIBOR as a reasonable and fair basis for calculating such rate and a LIBOR Prepayment Fee. Borrower further agrees to pay the LIBOR Prepayment Fee and Lender Taxes, if any, whether or not a Lender elects to purchase, sell and/or match funds.

Withholding Tax Exemption. At least five (5) Business Days prior to the first day on which interest or fees are payable hereunder for the account of any Lender, each Lender that is not incorporated under the laws of the United States of America, or a state thereof, agrees that it will deliver to Agent and Borrower two (2) duly completed copies of United States Internal Revenue Service Form 1001 or Form 4224, certifying in either case that such Lender is entitled to receive payments under this Agreement without deduction or withholding of any United States federal income taxes. Each Lender which so delivers a Form 1001 or Form 4224 further undertakes to deliver to Agent and Borrower two (2) additional copies of such form (or any applicable successor form) on or before the date that such form expires (currently, three (3) successive calendar years for Form 1001 and one (1) calendar year for Form 4224) or becomes obsolete or after the occurrence of any event requiring a change in the most recent forms so delivered by it, and such amendments thereto or extensions or renewals thereof as may be reasonably requested by Agent or Borrower, in each case certifying that such Lender is entitled to receive payments under this Agreement without deduction or withholding of any United States federal income taxes, unless an event (including without limitation any change in treaty, law or regulation) has occurred prior to the date on which any such delivery would otherwise be required which renders all such forms inapplicable or which would prevent such Lender from duly completing and delivering any such form with respect to it and such Lender advises Agent that it is not capable of receiving payments without any deduction or withholding of United States federal income taxes. If any Lender cannot deliver such form, then Borrower may withhold from such payments such amounts as are required by the Internal Revenue Code.

#### 2.5 <u>Fees</u>.

- 2.5.1 <u>Facility Fee</u>. From and after the Closing Date and until the Obligations are paid in full and this Agreement is terminated or, if sooner, the date the Commitments terminate, and subject to <u>Section 11.4.2</u>, Borrower shall pay to Agent, for the account of each Lender, a fee calculated at the rate of 12.5 basis points per year of the amount of the Facility. The aforesaid fee (the "<u>Facility Fee</u>") shall be payable, in the manner provided in <u>Section 2.6.2</u>, annually in advance on the Closing Date and on each April 1 thereafter and shall be prorated for the period between the Closing Date and March 31, 2003. The Facility Fee shall be deemed to be fully earned when paid and no portion thereof shall be refunded should the Facility be terminated prior to the expiration of the applicable annual period.
- 2.5.2 <u>Commitment Fees</u>. On the Closing Date, Borrower shall pay to Agent, for the account of each Lender, a commitment fee calculated at the rate of (i) 25 basis points of the amount of such Lender's Commitment under the Facility that corresponds to a commitment provided by such Lender under the 1998 Facility and (ii) 75 basis points of the amount of such Lender's Commitment under the Facility that does not correspond to a commitment provided by such Lender under the 1998 Facility and such 75 basis point commitment fee shall be prorated to reflect the fact that the period between the Closing Date and the Maturity Date is less than three (3) full years.
- 2.5.3 <u>Agency Fees</u>. Borrower shall pay Agent such fees as are provided for in a fee letter agreed to between Agent and Borrower, as in existence from time to time.
- 2.5.4 <u>Extension Fees</u>. In the event that Borrower exercises its right to extend the Maturity Date pursuant to <u>Section 2.10</u>, Borrower shall pay to Agent, for the account of each Lender, extension fees as provided in <u>Section 2.10</u>.
- 2.5.5 Payment of Fees. The fees described in this Section 2.5 represent compensation for services rendered and to be rendered separate and apart from the lending of money or the provision of credit and do not constitute compensation for the use, detention or forbearance of money, and the obligation of Borrower to pay the fees described herein shall be in addition to, and not in lieu of, the obligation of Borrower to pay interest, other fees and expenses otherwise described in this Agreement. All fees shall be payable when due in immediately available funds and shall be non-refundable when paid. If Borrower fails to make any payment of fees or expenses specified or referred to in this Agreement due to Agent or Lenders, including without limitation those referred to in this Section 2.5, in Section 12.1, or otherwise under this Agreement or any separate fee agreement between Borrower and Agent or any Lender relating to this Agreement, when due, the amount due shall bear interest until paid at the Base Rate and, after ten (10) days at the rate specified in Section 2.4.4 (but not to exceed the maximum rate permitted by applicable law), and shall constitute part of the Obligations.

## 2.6 Payments.

- 2.6.1 <u>Voluntary Prepayments</u>. Borrower may, upon not less than three (3) Business Days prior written notice to Agent not later than 11:00 A.M. (Eastern time) on the date given, at any time and from time to time, prepay any Loans in whole or in part. Any notice of prepayment given to Agent under this <u>Section 2.6.1</u> shall specify the date of prepayment and the principal amount of the prepayment. In the event of a prepayment of LIBOR Loans, Borrower shall concurrently pay any LIBOR Prepayment Fee payable in respect thereof. Agent shall provide to each Lender a confirming copy of such notice on the same Business Day such notice is received.
- 2.6.2 <u>Manner and Time of Payment</u>. All payments shall be made by Borrower to Agent at Agent's main office or such other place as Agent may from time to time specify in writing in lawful currency of the United States of America in immediately available funds, without counterclaim or setoff and free and clear of, and without any deduction or withholding for, any taxes or other payments. All payments of principal, interest and fees hereunder payable to Agent or the Lenders shall be made not later than 11:00 A.M. (Eastern time) on the date due; and funds received by Agent after that time and date shall be deemed to have been paid on the next succeeding Business Day.
- 2.6.3 <u>Payments on Non-Business Days</u>. Whenever any payment to be made by Borrower hereunder shall be stated to be due on a day which is not a Business Day, payments shall be made on the next succeeding Business Day and such extension of

time shall be included in the computation of the payment of interest hereunder and of any of the fees specified in <u>Section 2.5</u>, as the case may be.

- 2.7 <u>Increased Capital</u>. If either (i) the introduction of or any change in or in the interpretation of any law or regulation or (ii) compliance by Agent or any Lender with any guideline or request from any central bank or other Governmental Authority made or issued after the Closing Date affects or would affect the amount of capital required or expected to be maintained by Agent or such Lender or any corporation controlling Agent or such Lender, and Agent or such Lender determines that the amount of such capital is increased by or based upon the existence of Agent's obligations hereunder or such Lender's Commitment, then, upon demand by Agent or such Lender, Borrower shall immediately pay to Agent or such Lender, from time to time as specified by Agent or such Lender, additional amounts sufficient to compensate Agent or such Lender in the light of such circumstances, to the extent that Agent or such Lender determines such increase in capital to be allocable to the existence of Agent's obligations hereunder or such Lender's Commitment. A certificate as to such amounts submitted to Borrower by Agent or such Lender shall, in the absence of manifest error, be conclusive and binding for all purposes.
- 2.8 <u>Notice of Increased Cost</u>s. Each Lender agrees that, as promptly as reasonably practicable after it becomes aware of the occurrence of an event or the existence of a condition which would cause it to be affected by any of the events or conditions described in <u>Section 2.4.7</u> or <u>2.4.8</u> or <u>Section 2.7</u>, it will notify Borrower, and provide a copy of such notice to Agent, of such event and the possible effects thereof, <u>provided</u> that the failure to provide such notice shall not affect Lender's rights to reimbursement provided for herein.

#### 2.9 Letters of Credit

- 2.9.1 Terms of Letters of Credit. Up to \$50,000,000 of the Facility may be used by Borrower for the issuance of Letters of Credit by the Agent for the account of the Borrower subject to the terms and conditions set forth herein. Each Letter of Credit shall be denominated in dollars and shall be a standby letter of credit issued to support the obligations of Borrower (i) in connection with any purpose for which Loan proceeds may be used hereunder, or (ii) to advance loans to Foreign Affiliates, or to make other forms of Foreign Investment, provided that upon the issuance of a Letter of Credit for the purpose set forth in this clause (ii) the face amount thereof shall be added to the amount of Foreign Investments as of the time of issuance when determining compliance with Section 9.10. Each Letter of Credit shall have an initial term of not more than one (1) year, and shall expire no later than five (5) Business Days prior to the Termination Date. Although the Agent shall be the issuing bank of the Letter of Credit, each Lender hereby accepts for its own account and risk an undivided interest equal to its Pro Rata Share in the Agent's obligations and rights under each Letter of Credit issued hereunder. Each Lender unconditionally and irrevocably agrees with the Agent that, if a draft is paid under any Letter of Credit, such Lender shall promptly pay to the Agent an amount equal to such Lender's Pro Rata Share of the amount of such draft or any part thereof. Upon the issuance of each Letter of Credit hereunder, there shall be reserved from each Lender's Commitment an amount equal to such Lender's Pro Rata Share of the face amount of the Letter of Credit. Such reserved amounts shall remain in place and shall be unavailable for borrowing under Section 2.1 until the date that the Letter of Credit expires, is fully drawn or is terminated.
- 2.9.2 Letter of Credit Request. The Borrower shall give to the Agent a written notice in the form of Exhibit F hereto of each Letter of Credit requested hereunder (a "Letter of Credit Request") no less than six (6) Business Days prior to the proposed Issuance Date of the requested Letter of Credit, provided that Agent may, in its sole discretion, reduce said 6 Business Day period to not less than 3 Business Days. Each Letter of Credit Request shall specify (i) the name and address of the beneficiary of the requested Letter of Credit, (ii) the face amount of the requested Letter of Credit, (iii) the proposed Issuance Date and expiration date of the requested Letter of Credit, (iv) the proposed form of the requested Letter of Credit, and (v) the permitted purpose for which the Letter of Credit will be used. The Agent may also require that the Borrower complete its standard letter of credit application form and submit the same together with the Letter of Credit Request. The Agent shall provide a copy of the Letter of Credit Request to the Lenders at least two (2) Business Days before the Issuance Date of the Letter of Credit. If the issuance of the requested Letter of Credit will not cause the outstanding principal of the Loans to exceed the Maximum Loan Amount and the Agent determines, in its discretion, that it is willing to issue the requested Letter of Credit, and that it is satisfied with the proposed form thereof, the Letter of Credit shall be issued by the Agent and each of the Lenders shall then be obligated to the Agent with respect to its Pro Rata Share of the Letter of Credit as provided above in Section 2.9.1.
- 2.9.3 Letter of Credit Fees. On or before the Issuance Date of any requested Letters of Credit, the Borrower shall pay to the Agent for its own account an issuance fee equal to fifteen basis points (.15%) of the face amount of the Letter of Credit. On or before the date of any renewal or extension of a Letter of Credit, the Borrower shall pay to the Agent for its own account a renewal fee equal to seven and one half basis points (.075%) of the face amount of the Letter of Credit. The Borrower shall pay to the Agent for the account of the Lenders a Letter of Credit fee equal to the then prevailing Applicable LIBOR Rate Margin per annum of the face amount of the Letter of Credit, which Letter of Credit fee shall be due and payable on the Issuance Date of the Letter of Credit and on the date of each renewal or extension thereof, and shall be prorated for any partial year based on a 360-day year and paid for the actual number of days elapsed. Promptly after its receipt thereof the Agent shall distribute such Letter of Credit fee to the Lenders in accordance with their respective Pro Rata Shares. Such fees shall be nonrefundable and shall not be further prorated in the event that the Letter of Credit terminates prior to its scheduled expiration date. The Borrower also agrees to reimburse the Agent for all reasonable fees, costs, expenses and disbursements of the Agent in issuing, effecting payment under, amending or otherwise administering any Letter of Credit.
- 2.9.4 <u>Drawing</u>. Promptly after each Drawing Date the Agent shall notify the Lenders and the Borrower of the amount of the draft paid by the Agent on such Drawing Date. The payment of a draft under a Letter of Credit shall constitute an advance of a Loan which shall bear interest as a Base Rate Loan from the Drawing Date. On the Drawing Date each Lender shall pay to the

Agent its Pro Rata Share of the amount of the draft under the Letter of Credit so paid. If the Agent receives such payment from any Lender on a date after the Drawing Date, such Lender shall pay to the Agent on demand interest on said amount at the Federal Funds Rate. Each Lender's obligation to pay its Pro Rata Share of each draft under a Letter of Credit shall not be subject to the satisfaction of any conditions set forth in this Agreement and shall not depend on whether there may then be an Event of Default or an Unmatured Event of Default. Within three (3) Business Days after each Drawing Date, the Borrower shall deliver to the Agent a written explanation of the facts and circumstances relating to such drawing and a Compliance Certificate and any other information requested by the Agent for the purpose of allowing the Lenders to determine whether the drawing or related events have resulted in an Event of Default. The Agent shall promptly provide copies of such explanation and information to the Lenders. In the event that there is a legal impediment to treating the payment of a draft under a Letter of Credit as constituting an advance of a Loan as provided in this paragraph, upon written demand by the Agent, each other Lender shall purchase from the Agent a participating interest in the unreimbursed draft paid pursuant to such Letter of Credit in an amount equal to such other Lender's Pro Rata Share of such unreimbursed draft within one (1) Business Day after such demand therefor is made by the Agent and on such date of purchase, an amount equal to such Lender's Pro Rata Share of the such unreimbursed draft to be purchased by such Lender shall be advanced to the Agent.

- 2.9.5 <u>Borrower's Obligations Regarding Letters of Credit</u>. The Borrower's obligations under this <u>Section 2.9</u> shall be absolute and unconditional under any and all circumstances and irrespective of any set-off, counterclaim or defense to payment which the Borrower may have or have had against the Agent, any Lender or any beneficiary of a Letter of Credit. The Borrower also agrees that the Agent shall not be responsible for, and the Borrower's reimbursement obligations hereunder shall not be affected by, among other things, (i) the validity or genuineness of documents or of any endorsements thereon, even though such documents shall in fact prove to be invalid, fraudulent or forged, or (ii) any dispute between or among the Borrower and any beneficiary of any Letter of Credit or any other party to which such Letter of Credit may be transferred or (iii) any claims whatsoever of the Borrower against any beneficiary of such Letter of Credit or any such transferee. The Agent shall not be liable for any error, omission, interruption or delay in transmission, dispatch or delivery of any message or advice, however transmitted, in connection with any Letter of Credit, except for errors, omissions, interruptions or delays caused by the Agent's gross negligence or willful misconduct. The Borrower agrees that any action taken or omitted by the Agent under or in connection with any Letter of Credit or the related drafts or documents, if done in the absence of gross negligence or willful misconduct and in accordance with the standards of care specified in the Uniform Customs and Practices for Documentary Credits as the same may be amended from time to time, shall be binding on the Borrower and shall not result in any liability of the Agent to the Borrower.
- 2.10 <u>Extension of Maturity Date</u>. Borrower shall have the right to extend the Maturity Date for one (1) additional period of twelve (12) consecutive months (the "<u>Extension Period</u>"), provided that Borrower strictly complies with all of the following conditions precedent for such Extension Period:
  - (a) Borrower shall have provided a written notice of its intent to so extend (an "Extension Notice") to Agent at least forty-five (45) days prior to the initial Maturity Date;
  - (b) There shall be no existing Event of Default on the date the Extension Notice is delivered or on the initial Maturity Date; and
  - (c) Borrower shall have paid, prior to the commencement of the Extension Period, to Agent, for the account of each Lender, an extension fee equal to one-quarter of one percent (0.25%) of the amount of the Facility.
- 2.11 Reallocation of Loans on Closing Date. On the Effective Date the outstanding principal balance of the Loans under the 1998 Facility shall be reallocated among the Lenders such that after the Closing Date the outstanding principal amount of Loans owed to each Lender shall be equal to such Lender's Pro Rata Share of the outstanding amount of all Loans. Not later than 11:00 a.m. (Boston time) on the Closing Date each of Wachovia Bank and Bayerische Landesbank will make available to the Agent in immediately available funds, the amount required to effect such reallocation and such amount will be distributed to the other Lenders in order to achieve such reallocation. On the Closing Date, interest and unused commitment fees accrued under the 1998 Credit Agreement through the Closing Date will be paid by Borrower and distributed by the Agent to the Lenders based on the Pro Rata Shares applicable under the 1998 Credit Agreement prior to the Closing Date.

#### ARTICLE III

## **UNENCUMBERED PROPERTIES AND PORTFOLIO PROPERTIES**

- 3.1. <u>Listing of Unencumbered Properties</u>. The Borrower represents and warrants that each of the Properties listed on <u>Schedule 1</u> will on the Closing Date satisfy all of the conditions set forth in the definition of Unencumbered Property. From time to time during the term of this Agreement additional Properties may become Unencumbered Properties and certain Properties which previously satisfied the conditions set forth in the definition of Unencumbered Property may cease to be Unencumbered Properties by virtue of property dispositions, creation of Liens or other reasons. There shall be attached to each Compliance Certificate delivered pursuant hereto an updated listing of the Unencumbered Properties relied upon by the Borrower in computing the Value of All Unencumbered Properties and the Unencumbered Net Operating Income stated in such Compliance Certificate.
- 3.2. <u>Waivers by Requisite Lenders</u>. If any Property fails to satisfy any of the requirements contained in the definition of Unencumbered Property then the applicable Property may nevertheless be deemed to be Unencumbered Property hereunder if the

Requisite Lenders grant the necessary waivers and vote to accept such Property as an Unencumbered Property.

- 3.3. <u>Rejection of Unencumbered Properties</u>. If at any time the Agent determines that any Property listed as an Unencumbered Property by the Borrower does not satisfy all of the requirements of the definition of Unencumbered Property (to the extent not waived by the Requisite Lenders pursuant to <u>Section 3.2</u>) it may reject an Unencumbered Property by notice to the Borrower and if the Agent so requests the Borrower shall revise the applicable Compliance Certificate to reflect the resulting change in the Value of All Unencumbered Properties and the Unencumbered Net Operating Income.
- 3.4 <u>Updated Lists of Unencumbered Properties and Portfolio Properties.</u> <u>Schedule 1</u> contains a list of the Unencumbered Properties and <u>Schedule 1.1</u> sets forth a list of the Portfolio Properties (other than the Unencumbered Properties) each as of the date hereof. <u>Schedule 1.1</u> includes a statement of the reason that each Property listed thereon does not qualify as an Unencumbered Property. Promptly upon the acquisition or disposition of any of the Portfolio Properties and promptly upon the creation of any Lien or other event which causes any of the Portfolio Properties which previously qualified as an Unencumbered Property to no longer satisfy the definition of Unencumbered Property, Borrower shall deliver to Agent an updated <u>Schedule 1</u> and/or <u>Schedule 1.1</u> and any other information as may be reasonably requested by Agent relating to such change in the list of Unencumbered Properties and/or Portfolio Properties, including a Compliance Certificate.

#### **ARTICLE IV**

### **CONDITIONS TO LOANS**

- 4.1 <u>Conditions to effectiveness of amendment and restatement</u>. This Amended and Restated Credit Agreement shall become effective upon satisfaction or waiver of each of the following conditions precedent (provided that the Agent may, in its sole discretion, extend the time for delivery of evidence of qualification and good standing pursuant to paragraphs (f) and (h) of <u>Section 4.1.1</u> and paragraph (e) of <u>Section 4.1.2</u> with respect to states other than the state of organization of the applicable entity), and until such conditions are satisfied or waived, the 1998 Credit Agreement shall continue in effect:
- 4.1.1 <u>Borrower Documents</u>. Borrower shall have executed and/or delivered to Agent each of the following, in form and substance acceptable to Agent:
  - (a) this Agreement;
  - (b) the Loan Notes;
  - (c) the Subsidiary Guaranty;
  - (d) Certified copy of Borrower's Limited Partnership Agreement, as amended;
  - (e) Certified copy of Borrower's Certificate of Limited Partnership from the Delaware Secretary of State;
  - (f) Evidence of qualification and good standing of Borrower in Delaware and in each state where Borrower owns or manages any Unencumbered Property;
  - (g) Certified copies of the organizational documents of each Guarantor Subsidiary;
  - (h) Evidence of qualification and good standing of each Guarantor Subsidiary in the state of its organization and in each state where it owns any Unencumbered Property.
- 4.1.2 <u>REIT Documents</u>. The REIT shall have executed and/or delivered to Agent each of the following, in form and substance acceptable to Agent:
  - (a) The Guaranty;
  - (b) Articles of Incorporation, as amended, of the REIT, as certified by the Secretary of State of Maryland;
  - (c) By-laws of the REIT as certified by the Secretary of the REIT;
  - (d) Good Standing Certificate for the REIT from the Secretary of State of Maryland;
  - (e) Evidence of qualification and good standing of the REIT in each state where Borrower owns or manages any Unencumbered Property;
  - (f) Certificate of Secretary regarding corporate resolutions of the REIT, and the incumbency of its officers as certified by the Secretary of the REIT.
- 4.1.3 <u>Compliance Certificate</u>. Borrower shall have delivered to Agent a Compliance Certificate demonstrating compliance with the financial covenants in <u>Article IX</u> on the Closing Date.

- 4.1.4 <u>Material Adverse Changes</u>. No change, as determined by Agent shall have occurred, during the period commencing December 31, 2001, and ending on the Closing Date, which has a Material Adverse Effect on Borrower or the REIT.
- 4.1.5 <u>Litigation Proceedings</u>. There shall not have been instituted or threatened, during the period commencing December 31, 2001, and ending on the Closing Date, any litigation or proceeding in any court or Governmental Authority affecting or threatening to affect Borrower or the REIT which has a Material Adverse Effect, as reasonably determined by Agent.
- 4.1.6 <u>No Event of Default; Satisfaction of Financial Covenants</u>. On the Closing Date, no Event of Default or Unmatured Event of Default shall exist and all of the financial covenants contained in <u>Article IX</u> shall be satisfied.
- 4.1.7 <u>Payments on Closing Date</u>. Agent shall have received interest and unused commitment fees accrued under the 1998 Credit Agreement through the Closing Date, certain fees as provided in <u>Section 2.5</u>, and all expenses of Agent incurred prior to such Closing Date in connection with this Agreement (including without limitation all attorneys' fees and costs), shall have been paid by Borrower. The Agent shall pay to each Lender its portion of the facility fees and commitment fees as described in said <u>Section 2.5</u>.
- 4.1.8 <u>Opinion of Counsel</u>. Agent shall have received, on behalf of Agent and Lenders, favorable opinions of counsel for Borrower and the REIT dated as of the Closing Date, in form and substance satisfactory to Agent.
- 4.1.9 <u>Representations and Warranties</u>. All representations and warranties contained in this Agreement and the other Loan Documents shall be true and correct in all material respects.
- 4.2 <u>Conditions Precedent to All Loans and Letters of Credit</u>. The obligation of each Lender to make any Loan requested to be made by it and the obligation of the Agent to issue any Letter of Credit requested by Borrower, on any date, is subject to satisfaction of the following conditions precedent as of such date:
- 4.2.1 <u>Documents</u>. With respect to a request for a Loan, Agent shall have received, on or before the Funding Date and in accordance with the provisions of <u>Section 2.1.2</u>, an original and duly executed Notice of Borrowing. With respect to a request for a Letter of Credit, Agent shall have received, on or before the Issuance Date and in accordance with the provisions of <u>Section 2.9.2</u> an original and duly executed Letter of Credit Request.
- 4.2.2 <u>Additional Matters</u>. As of the Funding Date for any Loan and after giving effect to the Loans being requested and as of the Issuance Date of any Letter of Credit after giving effect to the issuance of the requested Letter of Credit:
  - (a) Representations and Warranties. All of the representations and warranties contained in this Agreement and in any other Loan Document (other than representations and warranties which expressly speak only as of a different date and other than for changes permitted or contemplated by this Agreement) shall be true and correct in all material respects on and as of such Funding Date or such Issuance Date, as though made on and as of such date;
  - (b) <u>No Default</u>. No Event of Default or Unmatured Event of Default shall have occurred and be continuing or would result from the making of the requested Loan or the issuance of the requested Letter of Credit, and all of the financial covenants contained in <u>Article IX</u> shall be satisfied; and
  - (c) <u>No Material Adverse Change</u>. Since the Closing Date, no change shall have occurred which shall have a Material Adverse Effect on Borrower or REIT, as determined by Agent, other than any such change the occurrence of which has been waived by Requisite Lenders in connection with any prior Borrowing.

Each submission by Borrower to Agent of a Notice of Borrowing with respect to a Loan and the acceptance by Borrower of the proceeds of each such Loan made hereunder and each submission by Borrower to Agent of a Letter of Credit Request shall constitute a representation and warranty by Borrower as of the Funding Date in respect of such Loan that all the conditions contained in this Section 4.2.2 have been satisfied.

#### ARTICLE V

#### REPRESENTATIONS AND WARRANTIES

In order to induce Lenders to make the Loans and issue the Letters of Credit, Borrower hereby represents and warrants to Lenders as follows:

5.1 <u>Borrower Organization; Partnership Powers.</u> Borrower (i) is a limited partnership duly organized, validly existing and in good standing under the laws of the State of Delaware, (ii) is duly qualified to do business as a foreign limited partnership and in good standing under the laws of each jurisdiction in which Borrower owns or leases real property or in which the nature of its business requires it to be so qualified, except for those jurisdictions where failure to so qualify and be in good standing would not have a Material Adverse Effect on Borrower, and (iii) has all requisite partnership power and authority to own and operate its property and assets and to conduct its business as presently conducted.

- 5.2 <u>Borrower Authority</u>. Borrower has the requisite partnership power and authority to execute, deliver and perform each of the Loan Documents to which it is or will be a party. The execution, delivery and performance thereof, and the consummation of the transactions contemplated thereby, have been duly approved by the general partner of Borrower, and no other partnership proceedings or authorizations on the part of Borrower or its general or limited partners are necessary to consummate such transactions. Each of the Loan Documents to which Borrower is a party has been duly executed and delivered by Borrower and constitutes its legal, valid and binding obligation, enforceable against it in accordance with its terms, subject to bankruptcy, insolvency and other laws affecting creditors' rights generally.
- 5.3 <u>REIT Organization; Corporate Powers</u>. The REIT (i) is a corporation duly organized, validly existing and in good standing under the laws of the State of Maryland, (ii) is duly qualified to do business as a foreign corporation and in good standing under the laws of each jurisdiction in which Borrower or the REIT owns or leases real property or in which the nature of its business requires it to be so qualified, except for those jurisdictions where failure to so qualify and be in good standing would not have a Material Adverse Effect on the REIT, and (iii) has all requisite corporate power and authority to own and operate its property and assets, to perform its duties as general partner of Borrower and to conduct its business as presently conducted.
- 5.4 <u>REIT Authority</u>. The REIT has the requisite corporate power and authority to execute, deliver and perform the Guaranty and, in its capacity as general partner of the Borrower each of the other Loan Documents. The execution, delivery and performance thereof, and the consummation of the transactions contemplated thereby, have been duly approved by the Board of Directors of the REIT, and no other corporate proceedings on the part of the REIT are necessary to consummate such transactions. Each of the Loan Documents to which the REIT is a party has been duly executed and delivered by the REIT and constitutes its legal, valid and binding obligation, enforceable against it in accordance with its terms, subject to bankruptcy, insolvency and other laws affecting creditors' rights generally.
- 5.5 Ownership of Borrower, each Subsidiary and Unconsolidated Affiliate. Schedule 5.5 sets forth the general partners and limited partners (or other holders of ownership interests) of each Subsidiary or Unconsolidated Affiliate and their respective ownership percentages and there are no other partnership (or other ownership) interests outstanding. Schedule 5.5 also classifies the Unconsolidated Affiliates as Eligible Affiliates, Foreign Affiliates, Internet Affiliates or other Unconsolidated Affiliates. Promptly following any material change in the information set forth on Schedule 5.5, Borrower shall deliver to the Agent a notice describing such change. All of the partnership (or other ownership) interests in Borrower and each Unconsolidated Affiliate and all of the stock of each Subsidiary have been issued in compliance with all applicable Requirements of Law.
- 5.6 No Conflict. The execution, delivery and performance by Borrower of the Loan Documents to which it is or will be a party, and each of the transactions contemplated thereby, do not and will not (i) conflict with or violate Borrower's limited partnership agreement or certificate of limited partnership or other organizational documents or the REIT's articles of incorporation, by-laws or other organizational documents, as the case may be, or (ii) conflict with, result in a breach of or constitute (with or without notice or lapse of time or both) a default under any Requirement of Law, Contractual Obligation or Court Order of or binding upon Borrower or the REIT, or (iii) require termination of any Contractual Obligation, or (iv) result in or require the creation or imposition of any Lien whatsoever upon any of the Portfolio Properties or assets of Borrower, other than Permitted Liens or Liens created by the Loan Documents.
- 5.7 <u>Consents and Authorizations</u>. Each of Borrower and the REIT has obtained all consents and authorizations required pursuant to its Contractual Obligations with any other Person, and shall have obtained all consents and authorizations of, and effected all notices to and filings with, any Governmental Authority, as may be necessary to allow Borrower and the REIT to lawfully execute, deliver and perform the Loan Documents.
- 5.8 <u>Governmental Regulation</u>. Neither Borrower, the REIT nor any Subsidiary or Unconsolidated Affiliate is subject to regulation under the Public Utility Holding Company Act of 1935, the Federal Power Act, the Interstate Commerce Act, the Investment Company Act of 1940 or any other federal or state statute or regulation such that its ability to incur indebtedness is limited or its ability to consummate the transactions contemplated by the Loan Documents is materially impaired.
- 5.9 <u>Prior Financials</u>. The December 31, 2001 Consolidated Balance Sheet, Statement of operations and Statement of Cash Flows of (i) the REIT contained in the REIT's Form 10K and (ii) the Borrower contained in the Borrower's Form 10K (the "<u>December 31, 2001 Financials</u>") delivered to Agent prior to the date hereof were prepared in accordance with GAAP and fairly present the assets, liabilities and financial condition of the REIT on a consolidated basis, at such date and the results of its operations and its cash flows, on a consolidated basis, for the period then ended.
- 5.10 <u>Projections and Forecasts</u>. Each of the projections and forecasts delivered to Agent since January 1, 2001 (1) has been prepared by Borrower in light of the past business and performance of Borrower on a consolidated basis and (2) represent as of the date thereof, the reasonable good faith estimates of Borrower's financial personnel.
- 5.11 <u>Prior Operating Statements</u>. Each of the consolidating operating statements pertaining to the Portfolio Properties delivered to Agent since January 1, 2001 was prepared in accordance with GAAP in effect on the date such operating statement of each Portfolio Property was prepared and fairly presents the results of operations of such Portfolio Property for the period then ended.
- 5.12 <u>Rent Rolls</u>. The Rent Rolls for the Portfolio Properties as of December 31, 2001 previously delivered to the Agent pursuant to the 1998 Facility (i) have been prepared in accordance with the books and records of the Portfolio Properties, and (ii) fairly present the leasing status of the Portfolio Properties as of the date thereof. Since the date of said Rent Rolls there has been no substantial adverse change to the leasing status of the Portfolio Properties.

- 5.13 Litigation; Adverse Effects.
  - There is no action, suit, Proceeding, governmental investigation or arbitration, at law or in equity, or before or by any Governmental Authority, pending or, to the best of Borrower's knowledge, threatened against Borrower, the REIT or any Portfolio Property, which, if adversely determined, would (i) result in a Material Adverse Effect on Borrower or the REIT, (ii) materially and adversely affect the ability of any party to any of the Loan Documents to perform its obligations thereunder, or (iii) materially and adversely affect the ability of Borrower to perform its obligations contemplated in the Loan Documents.
  - Neither Borrower nor the REIT is (i) in violation of any applicable law, which violation has a Material Adverse Effect on Borrower or the REIT, or (ii) in default with respect to any Court Order.
- 5.14 No Material Adverse Change. Since December 31, 2001, there has occurred no event which has a Material Adverse Effect on Borrower or the REIT, and no material adverse change in Borrower's ability to perform its obligations under the Loan Documents to which it is a party or the transactions contemplated thereby.
- Payment of Taxes. All material tax returns and reports to be filed by Borrower and the REIT have been timely filed, and all taxes, assessments, fees and other governmental charges shown on such returns or otherwise payable by Borrower have been paid when due and payable (other than real property taxes, which may be paid prior to delinquency so long as no penalty or interest shall attach thereto), except such taxes, if any, as are reserved against in accordance with GAAP and are being contested in good faith by appropriate proceedings or such taxes, the failure to make payment of which when due and payable will not have, in the aggregate, a Material Adverse Effect on Borrower or the REIT. Borrower has no knowledge of any proposed tax assessment against Borrower or the REIT that will have a Material Adverse Effect on Borrower or the REIT.
- Material Adverse Agreements. Neither the Borrower nor the REIT is a party to or subject to any Contractual Obligation or other restriction contained in the Borrower's limited partnership agreement or certificate of limited partnership, the REIT's Articles of Incorporation or bylaws or similar governing documents which has a Material Adverse Effect on Borrower or the ability of Borrower to perform its obligations under the Loan Documents.
- <u>Performance</u>. Neither Borrower nor the REIT is in default in the performance, observance or fulfillment of any of the obligations, covenants or conditions contained in any Contractual Obligation applicable to it, and no condition exists which, with the giving of notice or the lapse of time or both, would constitute a default under such Contractual Obligation in each case, except where the consequences, direct or indirect, of such default or defaults, if any, will not have a Material Adverse Effect on Borrower or the REIT.
- <u>Federal Reserve Regulations</u>. No part of the proceeds of the Loan hereunder will be used to purchase or carry any "margin security" as defined in Regulation T or for the purpose of reducing or retiring any indebtedness which was originally incurred to purchase or carry any margin security or for any other purpose which might constitute this transaction a "purpose credit" within the meaning of said Regulation T. Neither Borrower nor the REIT is engaged primarily in the business of extending credit for the purpose of purchasing or carrying any "margin stock" as defined in Regulation U. No part of the proceeds of the Loan hereunder will be used for any purpose that violates, or which is inconsistent with, the provisions of Regulation X or any other regulation of the Federal Reserve Board.
- <u>Unsecured Term Notes</u>. The Unsecured Term Notes were issued in compliance with all applicable Requirements of Law and there is no existing "Event of Default" as defined in the Unsecured Term Note Indenture.
- Requirements of Law. Borrower and the REIT are in compliance with all Requirements of Law (including without limitation the Securities Act and the Securities Exchange Act, and the applicable rules and regulations thereunder, state securities law and "Blue Sky" laws) applicable to it and its respective businesses, in each case, where the failure to so comply will have a Material Adverse Effect on any such Person. The REIT has made all filings with and obtained all consents of the Commission required under the Securities Act and the Securities Exchange Act in connection with the execution, delivery and performance by the REIT of the Loan Documents, except that this Credit Agreement will be filed with the Commission after the closing date.
- Patents, Trademarks, Permits, Etc. Borrower and the REIT own, are licensed or otherwise have the lawful right to use, or have all permits and other governmental approvals, patents, trademarks, trade names, copyrights, technology, know-how and processes used in or necessary for the conduct of each such Person's business as currently conducted, the absence of which would have a Material Adverse Effect upon such Person. The use of such permits and other governmental approvals, patents, trademarks, trade names, copyrights, technology, know-how and processes by each such Person does not infringe on the rights of any Person, subject to such claims and infringements as do not, in the aggregate, give rise to any liability on the part of any such Person which would have a Material Adverse Effect on any such Person.
- Environmental Matters. Except as set forth on Schedule 5.22, to the best of Borrower's knowledge, (i) the operations of Borrower and its Subsidiaries and Unconsolidated Affiliates comply in all material respects with all applicable local, state and federal environmental, health and safety Requirements of Law ("Environmental Laws"); (ii) none of the Portfolio Properties are subject to any Remedial Action or other Liabilities and Costs arising from the Release or threatened Release of a Contaminant into the environment in violation of any Environmental Laws; (iii) neither Borrower, the REIT nor any Unconsolidated Affiliate or Subsidiary has filed any notice under applicable Environmental Laws reporting a Release of a Contaminant into the environment in violation of any Environmental Laws, except as the same may have been heretofore remedied; (iv) there is not now on or in any of

the Portfolio Properties (except in compliance in all material respects with all applicable Environmental Laws): (A) any underground storage tanks, (B) any asbestos-containing material, or (C) any polychlorinated biphenyls (PCB's) used in hydraulic oils, electrical transformers or other equipment owned by such Person; and (v) neither Borrower, the REIT nor any Unconsolidated Affiliate or Subsidiary has received any notice or claim to the effect that it is or may be liable to any Person as a result of the Release or threatened Release of a Contaminant into the environment.

- 5.23 <u>Unencumbered Properties</u>. Each of the Properties listed on <u>Schedule 1</u> (i) qualifies as an Unencumbered Property and (ii) is free of any material defects in the roof, foundation, structural elements and masonry walls of the buildings thereon or their hyac, electrical, sprinkler or plumbing systems.
- 5.24 <u>Solvency</u>. Borrower is and will be Solvent after giving effect to the disbursements of the Loans and the payment and accrual of all fees then payable.
- 5.25 <u>Title to Assets; No Liens</u>. Borrower has good, indefeasible and merchantable title to all Properties owned or leased by it, and each of the Unencumbered Properties is free and clear of all Liens, except Permitted Liens.
- 5.26 <u>Use of Proceeds</u>. Borrower's use of the proceeds of the Loans are, and will continue to be, legal and proper uses (and to the extent necessary, duly authorized by Borrower's partners) and such uses are consistent with all applicable laws and statutes and Section 7.9.
- 5.27 <u>REIT Capitalization</u>. All of the capital stock of the REIT has been issued in compliance with all applicable Requirements of Law.
- 5.28 <u>ERISA</u>. Neither the REIT nor any ERISA Affiliate thereof (including, for all purposes under this <u>Section 5.28</u>, Borrower) has in the past five (5) years maintained or contributed to or currently maintains or contributes to any Benefit Plan. No Unconsolidated Affiliate has or is likely to incur any liability with respect to any Benefit Plan maintained or contributed to by such Unconsolidated Affiliate or its ERISA Affiliates, which would have a Material Adverse Effect on Borrower. Neither the REIT nor any ERISA Affiliate thereof has during the past five (5) years maintained or contributed to or currently maintains or contributes to any employee welfare benefit plan within the meaning of Section 3(1) of ERISA which provides benefits to retirees. Neither the REIT nor any ERISA Affiliate thereof is now contributing nor has it ever contributed to or been obligated to contribute to any Multiemployer Plan, no employees or former employees of the REIT, or such ERISA Affiliate have been covered by any Multiemployer Plan in respect of their employment by the REIT, and no ERISA Affiliate of the REIT has or is likely to incur any withdrawal liability with respect to any Multiemployer Plan which would have a Material Adverse Effect on the REIT.
- 5.29 <u>Status as a REIT</u>. The REIT (i) is a real estate investment trust as defined in Section 856 of the Internal Revenue Code (or any successor provision thereto), (ii) has not revoked its election to be a real estate investment trust, (iii) has not engaged in any "prohibited transactions" as defined in Section 856(b)(6)(iii) of the Internal Revenue Code (or any successor provision thereto), and (iv) for its current "tax year" (as defined in the Internal Revenue Code) is and for all prior tax years subsequent to its election to be a real estate investment trust has been entitled to a dividends paid deduction which meets the requirements of Section 857 of the Internal Revenue Code.
- 5.30 <u>Ownership</u>. The REIT does not own or have any interest in any other Person, other than its general partnership interests in Borrower and in the ATC Partnership and its interest as a Member of Chelsea GCA Realty LLC and S/C Orlando Development LLC.
- 5.31 <u>NYSE Listing</u>. The common stock of the REIT is and will continue to be listed for trading on either the New York Stock Exchange or the American Stock Exchange.
- 5.32 <u>Current Construction Projects</u>. <u>Schedule 5.32</u> sets forth a description of all of the Current Construction Projects and all of the Construction Projects presently planned for 2002 including with respect to each project: its location, gross leasable area, total budgeted cost, construction commencement date and expected Rent Stabilization Date which information shall be deemed to be updated to reflect information set forth in Compliance Certificates delivered to Agent.

#### ARTICLE VI

## **REPORTING COVENANTS**

Borrower covenants and agrees that, on and after the date hereof, until payment in full of all of the Obligations, the expiration of the Commitments and termination of this Agreement:

- 6.1 <u>Financial Statements and Other Financial and Operating Information</u>. Borrower shall maintain or cause to be maintained a system of accounting established and administered in accordance with sound business practices and consistent with past practice to permit preparation of quarterly and annual financial statements in conformity with GAAP, and each of the financial statements described below shall be prepared on a consolidated basis for the REIT and for the Borrower from such system and records. Borrower shall deliver or cause to be delivered to Agent (with copies sufficient for each Lender):
- 6.1.1 Rent Rolls. As soon as practicable following the request from any Lender, rent rolls (on Borrower's detailed form of rent roll) for each Portfolio Property (to the extent so requested) dated as of the last day of the most recently completed

Fiscal Quarter, in form and substance satisfactory to Agent, certified by the REIT's chief executive officer, president, chief financial officer, or treasurer. Copies of the rent rolls will be provided only to those Lenders which expressly request copies thereof.

- 6.1.2 <u>Quarterly Financial Statements Certified by Officer</u>. As soon as practicable, and in any event within fifty (50) days after the end of each of the first three Fiscal Quarters, consolidated balance sheets, statements of operations and statements of cash flow for the REIT and the Borrower, which may be in the form provided to the Commission on the REIT's Form 10Q and the Borrower's Form 10Q (unless the Borrower is not required to file a Form 10Q), and certified by the REIT's chief executive officer, president, chief financial officer, or treasurer.
- 6.1.3 Annual Financial Statements. Within ninety (90) days after the close of each Fiscal Year, annual Financial Statements of the REIT and of the Borrower, on a consolidated basis (in the form provided to the Commission on the REIT's Form 10K and the Borrower's Form 10K), audited and certified without qualification by the Accountants provided, however, that at such time as the Borrower is no longer required to file a Form 10K with the Commission, Borrower may deliver only the REIT's audited Financial Statement accompanied by a letter from the Accountants stating that with the exception of the minority interest line there is no material difference between the Financial Statements of Borrower and such audited Financial Statements of the REIT. To the extent Agent desires additional details or supporting information with respect to Unconsolidated Affiliates or individual Portfolio Properties not contained in the REIT's or Borrower's Form 10K, Borrower shall provide Agent with such details or supporting information as Agent requests which is reasonably available to Borrower.
- Officer's Certificate. (i) Together with each delivery of any Financial Statement pursuant to Sections 6.1.2 and 6.1.3, (A) an Officer's Certificate of the REIT stating that each of the Financial Statements delivered to Agent therewith (i) has been prepared in accordance with the books and records of the REIT and Borrower on a consolidated basis, and (ii) fairly presents the financial condition of the REIT and Borrower on a consolidated basis, at the dates thereof (and, if applicable, subject to normal year-end adjustments) and the results of its operations and cash flows, on a consolidated basis, for the period then ended; (B) an Officer's Certificate of the REIT, stating that the executive officer who is the signatory thereto (which officer shall be the chief executive officer, the chief operating officer, the chief financial officer, any senior vice president or the treasurer of the REIT) has reviewed, or caused under his supervision to be reviewed, the terms of this Agreement and the other principal Loan Documents, and has made, or caused to be made under his supervision, a review in reasonable detail of the transactions and condition of Borrower and the REIT, during the accounting period covered by such Financial Statements, and that such review has not disclosed the existence during or at the end of such accounting period, and that the signers do not have knowledge of the existence as of the date of the Officer's Certificate, of any condition or event which constitutes an Event of Default or Unmatured Event of Default, or, if any such condition or event existed or exists, specifying the nature and period of existence thereof and what action has been taken, is being taken and is proposed to be taken with respect thereto; and (C) a Compliance Certificate demonstrating in reasonable detail (which detail shall include actual calculation and supporting information) compliance during and at the end of such accounting periods with the financial covenants contained in Article IX.
- 6.1.5 <u>Borrowing Projections</u>. At least ten (10) days prior to the end of each Fiscal Year, projections of Borrower, on a consolidated basis, detailing expected borrowing and repayment of the Loans for the following Fiscal Year together with an Officer's Certificate of the REIT stating that such projections (1) have been prepared by Borrower in light of the past business and performance of Borrower on a consolidated basis and (2) represent, as of the date thereof, the reasonable good faith estimates of Borrower's financial personnel. Borrower shall also provide such additional supporting details as Agent may reasonably request.
- 6.1.6 <u>Compliance with Unencumbered Property Requirements</u>. Promptly upon becoming aware of any condition or event which causes any of the Properties listed as Unencumbered Properties on the most recent Compliance Certificate to no longer comply with the requirements set forth in the definition of Unencumbered Properties, an Officer's Certificate specifying the relevant information and a revised Compliance Certificate.
- 6.1.7 Knowledge of Event of Default. Promptly upon Borrower obtaining knowledge (i) of any condition or event which constitutes an Event of Default or Unmatured Event of Default, or becoming aware that any Lender has given notice or taken any other action with respect to a claimed Event of Default or Unmatured Event of Default or (ii) of any condition or event which has a Material Adverse Effect on Borrower or the REIT, an Officer's Certificate specifying the nature and period of existence of any such condition or event, or specifying the notice given or action taken by such Lender and the nature of such claimed Event of Default, Unmatured Event of Default, event or condition, and what action Borrower and/or the REIT has taken, is taking and proposes to take with respect thereto.
- 6.1.8 <u>Litigation, Arbitration or Government Investigation</u>. Promptly upon Borrower or the REIT obtaining knowledge of (i) the institution of, or threat of, any material action, suit, proceeding, governmental investigation or arbitration against or affecting Borrower or the REIT not previously disclosed in writing by Borrower to Agent pursuant to this <u>Section 6.1.8</u>, or (ii) any material development in any action, suit, proceeding, governmental investigation or arbitration already disclosed, which, in either case, has a Material Adverse Effect on Borrower or the REIT, a notice thereof to Agent and such other information as may be reasonably available to it to enable Agent, Lenders and their counsel to evaluate such matters.
- 6.1.9 <u>Establishment of Benefit Plan and Increase in Contributions to the Benefit Plan</u>. Not less than ten (10) days prior to the effective date thereof, a notice to Agent of the establishment of a Benefit Plan (or the incurrence of any obligation to contribute to a Multiemployer Plan) by Borrower, the REIT or any ERISA Affiliate. Within thirty (30) days after the first to occur of an amendment of any then existing Benefit Plan of Borrower, the REIT or any ERISA Affiliate which will result in an increase in the benefits under such Benefit Plan or a notification of any such increase, or the establishment of any new Benefit Plan by Borrower, the REIT or any ERISA Affiliate or the commencement of contributions to any Benefit Plan to which Borrower, the REIT or any ERISA Affiliate was not previously contributing, a copy of said amendment, notification or Benefit Plan. For so long

as any such Benefit Plan exists, prompt notice of any Termination Event, prohibited transaction, funding waiver request, unfavorable determination letter or withdrawal liability under a Multiemployer Plan.

- 6.1.10 Failure of the REIT to Qualify as Real Estate Investment Trust. Promptly upon, and in any event within forty-eight (48) hours after Borrower first has actual knowledge of (i) the REIT failing to continue to qualify as a real estate investment trust as defined in Section 856 of the Internal Revenue Code (or any successor provision thereof), (ii) any act by the REIT causing its election to be taxed as a real estate investment trust to be terminated, (iii) any act causing the REIT to be subject to the taxes imposed by Section 857(b)(6) of the Internal Revenue Code (or any successor provision thereto), or (iv) the REIT failing to be entitled to a dividends paid deduction which meets the requirements of Section 857 of the Internal Revenue Code, a notice of any such occurrence or circumstance.
- 6.1.11 Asset Acquisitions and Dispositions, Indebtedness, Etc. Without limiting Article VIII or any other restriction in the Loan Documents, and in all events not later than the same Business Day on which there is public disclosure of any material Investments (other than in Cash Equivalents), acquisitions, dispositions, disposals, divestitures or similar transactions involving Property, the creation of Liens on any of the Portfolio Properties, other than Non-Retail Properties, the execution of any long term leases of real estate in which the Borrower or its Subsidiary is the lessee, the raising of additional equity or the incurring or repayment of material Indebtedness, by or with Borrower, any Unconsolidated Affiliate or Subsidiary or the REIT, telephonic or facsimile notice thereof to Lori Y. Litow or such other person(s) as Agent may designate from time to time, and, if requested by Agent, promptly upon consummation of such transaction, a Compliance Certificate demonstrating in reasonable detail (which detail shall include actual calculations) compliance, after giving effect to such proposed transaction(s), with the covenants contained in Article IX.
- 6.1.12 Other Information. Such other information, reports, contracts, schedules, lists, documents, agreements and instruments in the possession of the REIT or Borrower with respect to (i) the Portfolio Properties, (ii) any material change in the REIT's investment, finance or operating policies, or (iii) Borrower's or the REIT's business, condition (financial or otherwise), operations, performance, properties or prospects as Agent may from time to time reasonably request, including, without limitation, annual information with respect to cash flow projections, budgets, operating statements (current year and immediately preceding year), rent rolls, lease expiration reports, leasing status reports, tenant sales reports (to the extent available), note payable summaries, equity funding requirements, contingent liability summaries, line of credit summaries, tenant improvement allowance summaries, note receivable summaries, schedules of outstanding letters of credit, summaries of cash and Cash Equivalents, projections of management and leasing fees and overhead budgets. Provided that Agent gives Borrower reasonable prior notice and an opportunity to participate, Borrower hereby authorizes Agent to communicate with the Accountants and authorizes the Accountants to disclose to Agent any and all financial statements and other information of any kind, including copies of any management letter or the substance of any oral information, that such accountants may have with respect to Borrower's or the REIT's condition (financial or otherwise), operations, properties, performance and prospects. At Agent's request, Borrower shall deliver a letter addressed to the Accountants instructing them to disclose such information in compliance with this Section 6.1.12.
- 6.1.13 <u>Press Releases; SEC Filings and Financial Statements</u>. Telephonic or telecopy notice to Agent concurrent with or prior to issuance of any material press release concerning the REIT or Borrower and, as soon as practicable after filing with the Commission, all reports and notices, proxy statements, registration statements and prospectuses of the REIT. All materials sent or made available generally by the REIT to the holders of its publicly-held Securities or to a trustee under any indenture or filed with the Commission, including all periodic reports required to be filed with the commission, will be delivered to Agent and Lenders as soon as available.
- 6.1.14 <u>Accountant Reports</u>. Copies of all reports prepared by the Accountants and submitted to Borrower or the REIT in connection with each annual, interim or special audit or review of the financial statements or practices of Borrower or the REIT, including the comment letter submitted by the Accountants in connection with their annual audit.
- 6.1.15 <u>Termination or Modification of Earthquake Coverage</u>. Promptly upon, and in any event within thirty (30) days after Borrower first has knowledge of the termination or modification (with respect to the amount of either the coverage provided or the applicable deductible) of the coverage provided by the blanket property insurance rider regarding earthquake insurance for Portfolio Properties located in "Zone 1" maintained by Borrower as of the date of this Agreement, a notice of such termination or modification.
- 6.2 Environmental Notices. Borrower shall notify Agent, in writing, as soon as practicable, and in any event within ten (10) days after Borrower's or the REIT's learning thereof, of any: (i) written notice or claim to the effect that Borrower or the REIT is or may be liable to any Person as a result of any material Release or threatened Release of any Contaminant into the environment; (ii) written notice that Borrower or the REIT is subject to investigation by any Governmental Authority evaluating whether any Remedial Action is needed to respond to the Release or threatened Release of any Contaminant into the environment; (iii) written notice that any Portfolio Property is subject to an Environmental Lien; (iv) written notice that Borrower, any Subsidiary or Unconsolidated Affiliate or the REIT has received a notice of violation of any Environmental Laws by Borrower, any Subsidiary or Unconsolidated Affiliate or the REIT; (v) commencement or written threat of any judicial or administrative proceeding alleging a violation of any Environmental Laws; (vi) written notice from a Governmental Authority of any changes to any existing Environmental Laws that will have a Material Adverse Effect on the operations of Borrower or the REIT; or (vii) any proposed acquisition of stock, assets, real estate or leasing of property, or any other action by Borrower that, to the best of Borrower's knowledge, could subject Borrower, any Subsidiary or Unconsolidated Affiliate or the REIT to environmental, health or safety Liabilities and Costs that will have a Material Adverse Effect on Borrower or the REIT.

6.3 <u>Confidentiality</u>. Confidential Information obtained by Agent or Lenders pursuant to this Agreement or in connection with the Facility shall not be disseminated by Agent or Lenders and shall not be disclosed to third parties except to regulators, taxing authorities and other governmental agencies having jurisdiction over Agent or such Lender or otherwise in response to Requirements of Law, to their respective auditors and legal counsel and in connection with regulatory, administrative and judicial proceedings as necessary or relevant including enforcement proceedings relating to the Loan Documents, and to any prospective assignee of or participant in a Lender's interest under this Agreement or any prospective purchaser of the assets or a controlling interest in any Lender, <u>provided</u> that such prospective assignee, participant or purchaser first agrees to be bound by the provisions of this <u>Section 6.3</u>. For purposes hereof, "<u>Confidential Information</u>" shall mean all nonpublic information obtained by Agent or Lenders, unless and until such information becomes publicly known, other than as a result of unauthorized disclosure by Agent or Lenders of such information.

#### ARTICLE VII

## **AFFIRMATIVE COVENANTS**

Borrower covenants and agrees that, on and after the date hereof, until payment in full of all of the Obligations, the expiration of the Commitments and termination of this Agreement:

- 7.1 <u>Existence</u>. Each of Borrower and the REIT shall at all times maintain its existence and preserve and keep in full force and effect its rights and franchises. Borrower shall remain a Delaware limited partnership with the REIT as its sole general partner.
- 7.2 <u>Qualification, Name</u>. Each of Borrower and the REIT shall qualify and remain qualified to do business in each jurisdiction in which the nature of its business requires it to be so qualified except for those jurisdictions where failure to so qualify would not have a material Adverse Effect on Borrower. Borrower will transact business solely in its own name or in the commonly known name of one of the Portfolio Properties.
- 7.3 <u>Compliance with Laws, Etc.</u> Each of Borrower and REIT shall (i) comply with all Requirements of Law, and all restrictive covenants affecting it or its properties, performance, prospects, assets or operations, and (ii) obtain as needed all Permits necessary for its operations and maintain such in good standing, except where the failure to do so will not have a Material Adverse Effect on Borrower.
- 7.4 Payment of Taxes and Claims. Each of Borrower and the REIT shall pay (i) all taxes, assessments and other governmental charges imposed upon it or on any of its properties or assets or in respect of any of its franchises, business, income or property before any penalty or interest accrues thereon, and (ii) all claims (including, without limitation, claims for labor, services, materials and supplies) for sums which have become due and payable and which by law have or may become a Lien other than a judgment lien upon any of Borrower's properties or assets, prior to the time when any penalty or fine shall be incurred with respect thereto, provided, however that the payment of such taxes, assessments, charges and claims may be deferred so long as the validity or amount thereof shall currently be contested in good faith by appropriate proceedings and if Borrower shall have set aside in its books adequate reserves specifically with respect thereto, but Borrower shall pay such matters prior to the foreclosure of a Lien which may have attached as security therefor.
- 7.5 <u>Maintenance of Properties; Insurance</u>. Borrower shall maintain in good repair, working order and condition, excepting ordinary wear and tear, all of the Portfolio Properties and will make or cause to be made all appropriate repairs, renewals and replacements thereof. Borrower shall maintain commercially reasonable and appropriate amounts of "all risk" property and liability insurance, which insurance shall include in any event:
  - (a) with respect to each Property: (i) property and casualty insurance (including coverage for flood and water damage for any Portfolio Property located within a 100-year flood plain) in an amount not less than the replacement costs of the improvements thereon (subject to reasonable deductibles and, in the case of flood insurance, subject to the maximum coverages available under the National Flood Insurance Program), and (ii) loss of rental insurance income in an amount not less than one year's gross revenues of such Portfolio Property; and
  - (b) comprehensive general liability insurance in an amount not less than \$20,000,000 per occurrence, including all insurance pursuant to umbrella and excess liability policies.

At the request of Agent, Borrower shall provide evidence of insurance, including certificates of insurance and binders.

7.6 <u>Inspection of Property; Books and Records; Discussions.</u> Borrower shall permit, and shall cause the REIT and each Subsidiary or Unconsolidated Affiliate to permit, any authorized representative(s) designated by any Lender to visit and inspect any of its properties, to inspect financial and accounting records and leases, and to make copies and take extracts therefrom, all at such times after reasonable advance notice during normal business hours and as often as any Lender may reasonably request. In connection therewith, Borrower shall pay all expenses of the Agent (but not of the other Lenders) of the types described in <u>Section 12.1</u> subject to the limitation that prior to an Event of Default the Borrower shall not be required to reimburse expenses for inspections of Properties made more frequently than annually. Borrower will keep proper books of record and account in which entries, in conformity with GAAP and as otherwise required by this Agreement and applicable Requirements of Law, shall be made of all dealings and transactions in relation to its businesses and activities and as otherwise required under <u>Section 6.1</u>.

- 7.7 <u>Maintenance of Permits, Etc.</u> Each of Borrower and the REIT will maintain in full force and effect all Permits, franchises, patents, trademarks, trade names, copyrights, authorizations or other rights necessary for the operation of its business, except where the failure to obtain any of the foregoing would not have a Material Adverse Effect on Borrower; and notify Agent in writing, promptly after learning thereof, of the suspension, cancellation, revocation or discontinuance of or of any pending or threatened action or proceeding seeking to suspend, cancel, revoke or discontinue any material Permit, patent, trademark, trade name, copyright, governmental approval, franchise authorization or right.
- 7.8 <u>Conduct of Business</u>. Subject to the limits on Permitted Investments pursuant to <u>Section 9.10</u>, and Investments in cash and Cash Equivalents, Borrower shall engage only in the business of direct or indirect ownership, operation, development, redevelopment and management of real estate and businesses or activities which are substantially similar, related or incidental thereto.
- 7.9 <u>Use of Proceeds</u>. Borrower shall use the proceeds of the Loans only for predevelopment costs, development costs, acquisitions, working capital, equity Investments and repayment of Indebtedness, including required interest and/or principal payments thereon.
- 7.10 <u>Securities Law Compliance</u>. Each of the Borrower and the REIT shall comply in all material respects with all rules and regulations of the Commission and file all reports required by the Commission relating to the Borrower's or the REIT's publicly-held Securities.
- 7.11 <u>Continued Status as a REIT; Prohibited Transactions</u>. The REIT (i) will continue to be a real estate investment trust as defined in Section 856 of the Internal Revenue Code (or any successor provision thereto), (ii) will not revoke its election to be a real estate investment trust, (iii) will not engage in any "prohibited transactions" as defined in Section 856(b)(6)(iii) of the Internal Revenue Code (or any successor provision thereto), and (iv) will continue to be entitled to a dividend paid deduction meeting the requirements of Section 857 of the Internal Revenue Code.
- 7.12 <u>NYSE Listed Company</u>. The common stock of the REIT shall at all times be listed for trading on the New York Stock Exchange.
- 7.13 <u>Property Management</u>. All Portfolio Properties (other than Non-Retail Properties) in which the direct or indirect ownership interest of Borrower exceeds fifty percent (50%) shall be directly managed by Borrower or an Affiliate of Borrower.
- 7.14 <u>Interest Rate Contracts</u>. At all times when LIBOR for 30 day Interest Periods has, for 30 consecutive days, exceeded eight and one-quarter percent (8.25%), to the extent that the aggregate amount of variable interest rate Indebtedness of Borrower (including the Facility and the Fleet Term Loan) then outstanding exceeds forty percent (40%) of all outstanding Indebtedness of Borrower, Borrower shall maintain in effect Interest Rate Contracts which are satisfactory to the Agent. In determining whether such Interest Rate Contracts are satisfactory, the Agent shall not require that the terms thereof extend beyond the Termination Date.

#### **ARTICLE VIII**

#### **NEGATIVE COVENANTS**

Borrower covenants and agrees that, on and after the date hereof, until payment in full of all of the Obligations, the expiration of the Commitments and termination of this Agreement:

- 8.1 <u>Restrictions on Fundamental Changes</u>. Neither Borrower nor the REIT shall, without the prior written consent of the Requisite Lenders:
  - (a) Liquidate, wind-up or dissolve (or suffer any liquidation or dissolution);
  - (b) Change its Fiscal Year;
  - (c) Enter into any merger or consolidation in which any entity other than the Borrower or the REIT is the surviving entity or which results in the management of Borrower or the REIT by persons other than the Executive Officers, other than as permitted in Section 8.8; or
  - (d) Enter into any merger, consolidation, acquisition, joint venture or disposition which has the effect of changing Adjusted Asset Value by more than 35% of the Adjusted Asset Value (as most recently computed prior to such transaction);
  - (e) Enter into any merger or consolidation which has the effect of changing Adjusted Asset Value by more than 5% (but not more than 35%) of the Adjusted Asset Value (as most recently computed prior to such transaction)unless prior to the completion of such transaction the Borrower shall provide the Agent with a notice describing the terms of such transaction and a Compliance Certificate with updated calculations reflecting such transaction and showing that no financial covenants will be violated as a result thereof.

- 8.2 <u>ERISA</u>. Neither the Borrower nor the REIT shall permit any ERISA Affiliates to do any of the following to the extent that such act or failure to act would result in the aggregate, after taking into account any other such acts or failure to act, in a Material Adverse Effect on Borrower or the REIT:
  - (a) Engage, or knowingly permit an ERISA Affiliate to engage, in any prohibited transaction described in Section 406 of the ERISA or Section 4975 of the Internal Revenue Code which is not exempt under Section 407 or 408 of ERISA or Section 4975(d) of the Internal Revenue Code for which a class exemption is not available or a private exemption has not been previously obtained from the DOL;
  - (b) Permit to exist any accumulated funding deficiency (as defined in Section 302 of ERISA and Section 412 of the Internal Revenue Code), whether or not waived;
  - (c) Fail, or permit an ERISA Affiliate to fail, to pay timely required contributions or annual installments due with respect to any waived funding deficiency to any Plan if such failure could result in the imposition of a Lien or otherwise would have a Material Adverse Effect on Borrower or the REIT;
  - (d) Terminate, or permit an ERISA Affiliate to terminate, any Benefit Plan which would result in any liability of Borrower or an ERISA Affiliate under Title IV of ERISA or the REIT; or
  - (e) Fail, or permit any ERISA Affiliate to fail, to pay any required installment under section (m) of Section 412 of the Internal Revenue Code or any other payment required under Section 412 of the Internal Revenue Code on or before the due date for such installment or other payment, if such failure could result in the imposition of a Lien or otherwise would have a Material Adverse Effect on Borrower or the REIT.
- 8.3 Amendment of Constituent Documents. Except for any such amendment that is required (i) under any Requirement of Law imposed by any Governmental Authority or (ii) in order to maintain compliance with Section 7.11: (1) the Borrower shall not amend its partnership agreement (including, without limitation, as to the admission of any new partner, directly or indirectly), and (2) the REIT shall not amend its articles of incorporation or by-laws; in any such case, except amendments which do not materially affect the ability of the Borrower or the REIT to perform its obligations under the Loan Documents or other amendments which have received the prior written consent of the Agent.
- 8.4 <u>Margin Regulations</u>. No portion of the proceeds of any Loans shall be used in any manner which might cause the extension of credit or the application of such proceeds to violate Regulation T, U or X or any other regulation of the Federal Reserve Board or to violate the Securities Exchange Act or the Securities Act, in each case as in effect on the applicable Funding Date.

#### 8.5 With Respect to the REIT:

- 8.5.1 The REIT shall not own any material assets (other than its interest in the ATC Partnership, Chelsea GCA Realty, LLC and S/C Orlando Development LLC) or engage in any line of business other than owning partnership interests in Borrower.
- 8.5.2 The REIT shall not directly or indirectly create, incur, assume or otherwise become or remain directly or indirectly liable with respect to, any Indebtedness, except the Obligations and other Borrower Debt.
- 8.5.3 The REIT shall not directly or indirectly create, incur, assume or permit to exist any Lien (other than Permitted Liens) on or with respect to any of its Property or assets.
- 8.5.4 The REIT will not directly or indirectly convey, sell, transfer, assign, pledge or otherwise encumber or dispose of any of its partnership interests in Borrower so as to reduce its interest in Borrower to less than 60%.
- 8.6 <u>Restrictions on Indebtedness</u>. Except with the prior written consent of Requisite Lenders, the Borrower will not create, incur, assume, guarantee or become or remain liable, contingently or otherwise, with respect to any Indebtedness described in any one or more of the following paragraphs:
  - (a) Indebtedness that creates a violation of Sections 9.4, 9.5, 9.6 or 9.7;
  - (b) Indebtedness that, when counted as an Investment pursuant to clause (iii) of the definition of Investment, creates a violation of Section 9.10;
    - (c) Indebtedness that is prohibited by the terms of the Unsecured Term Notes; or
  - (d) If the terms of such Indebtedness include financial covenants, such covenants are determined by the Agent, in its sole discretion, to be more stringent than the covenants set forth in <u>Article IX</u>.
- 8.7 <u>Construction Projects</u>. Borrower shall not commence construction of any Construction Project if the addition of the budgeted project costs for such project to the CIP Budget Amount would result in a violation of <u>Section 9.10</u>.

8.8 <u>Discontinuity in Management</u>. In the event that during any period of twelve (12) consecutive months any three of the five Executive Officers shall cease to be active on a full time, continuous basis in the senior management of Borrower and the REIT ("<u>Discontinuity in Management</u>"), Borrower shall have up to one hundred eighty (180) days to obtain the approval of Requisite Lenders to additional executives, such that the remaining and new management executives, as a group, have substantial and sufficient knowledge, experience and capabilities in the management of a publicly-held company engaged in the operation of a multi-asset real estate business of the type engaged in by Borrower. In the event Borrower shall fail to obtain approval of Requisite Lenders as aforesaid within said 180-day period, then Borrower shall, at the election and upon the demand of Requisite Lenders, pay in full all Obligations under the Loan Documents not later than sixty (60) days after the end of such 180-day period, whereupon this Agreement and all Commitments hereunder shall be terminated. Any Lender which abstains or otherwise fails to respond to any request by Borrower for approval under this <u>Section 8.8</u> within ten (10) Business Days shall be counted among the Requisite Lenders approving the proposed additional executives.

#### **ARTICLE IX**

#### FINANCIAL COVENANTS

Borrower covenants and agrees that, on and after the date of this Agreement and until payment in full of all the Obligations, the expiration of all Commitments, the termination of all Letters of Credit and the termination of this Agreement:

- 9.1. <u>Value of All Unencumbered Properties</u>. The Borrower will not at any time permit the Value of All Unencumbered Properties to be less than one hundred seventy five percent (175%) of the outstanding balance of Unsecured Indebtedness.
- 9.2. <u>Minimum Debt Service Coverage</u>. The Borrower will not at any time permit the outstanding principal amount of the Unsecured Indebtedness to exceed an amount such that: (a) the Unencumbered Net Operating Income, divided by (b) Pro Forma Unsecured Debt Service Charges would be less than 1.5 for any Fiscal Quarter.
- 9.3 <u>Minimum Fair Market Net Worth</u>. Borrower will maintain a Fair Market Net Worth of not less than Four Hundred Thirty-One Million Dollars (\$431,000,000) plus seventy-five percent (75%) of Net Offering Proceeds received by Borrower after the Closing Date.
- 9.4 <u>Total Liabilities to Adjusted Asset Value Ratio</u>. Borrower will not at any time permit its Total Liabilities to exceed fifty-five percent (55%) of Adjusted Asset Value.
- 9.5 <u>Maximum Secured Borrower Debt</u>. The Secured Borrower Debt shall not exceed thirty-five percent (35%) of Adjusted Asset Value.
- 9.6 <u>Operating Cash Flow to Debt Service Ratio</u>. The ratio of Operating Cash Flow to Debt Service shall not be less than 2.0:1 for any Fiscal Quarter.
- 9.7 <u>EBITDA to Fixed Charges Ratio</u>. The ratio of EBITDA to Fixed Charges shall not be less than 1.75:1 for any Fiscal Quarter.
- 9.8 <u>Aggregate Occupancy Rate</u>. The Aggregate Occupancy Rate of the Unencumbered Properties shall not be less than ninety percent (90%).

## 9.9 <u>Distributions</u>.

- 9.9.1 Subject to Section 9.9.2, aggregate distributions to common shareholders of the REIT and all partners of Borrower holding common units other than the REIT shall not exceed the lesser of (i) ninety percent (90%) of Funds From Operations for any Fiscal Year or (ii) one hundred percent (100%) of Funds From Operations for more than two (2) consecutive Fiscal Quarters. For purposes of this Section 9.9, the term "distributions" shall mean and include all dividends and other distributions to, and the repurchase of stock or partnership interests from, the holder of any equity interests in Borrower or the REIT.
- 9.9.2 Aggregate distributions during the continuance of any Event of Default shall not exceed the lesser of (i) the aggregate amount permitted to be made during the continuance thereof under Section 9.9.1, and (ii) the minimum amount that the REIT must distribute to its shareholders in order to maintain compliance with Section 7.11. If the Loans are not paid in full on the Termination Date, no distributions shall be made thereafter except to the extent expressly authorized in advance by Agent.
- 9.10 <u>Permitted Investments</u>. In addition to Investments in cash and Cash Equivalents, Premium Centers and Other Retail Centers, Borrower may make Permitted Investments, so long as (i) the aggregate amount of all Permitted Investments does not exceed, at any time, thirty percent (30%) of Adjusted Asset Value, and (ii) the aggregate amount of each of the following categories of Permitted Investments does not exceed the specified percentage of Adjusted Asset Value, in each case as of the date made:

Maximum Percentage of <u>Adjusted Asset Value</u> Foreign Investments: 15%

Internet Investments: 10%

9.11 <u>Repurchase of Equity Interests</u>. The aggregate payments to repurchase, redeem or retire any common or preferred shares of the REIT, any common or preferred partnership units of Borrower or any other type of equity interests in Borrower or the REIT shall not exceed One Hundred Million Dollars (\$100,000,000) during any Fiscal Year and no such payments shall be made during the continuance of any Event of Default.

9.12 <u>Calculation</u>. Each of the foregoing ratios and financial requirements shall be calculated as of the last day of each Fiscal Quarter, provided, however that each of the foregoing ratios and financial requirements which is affected by an increase in the outstanding balance of the Loans, by the issuance of a Letter of Credit or by the sale or encumbrance of a Portfolio Property shall also be recalculated as of the time of such event. For purposes of determining compliance with <u>Section 9.6</u>, the period covered thereby shall be the immediately preceding Fiscal Quarter.

#### ARTICLE X

#### **EVENTS OF DEFAULT; RIGHTS AND REMEDIES**

- 10.1 <u>Events of Default</u>. Each of the following occurrences shall constitute an Event of Default under this Agreement:
- 10.1.1 <u>Failure to Make Payments When Due</u>. Borrower shall fail to pay (i) any amount due on the Termination Date, (ii) any principal when due, or (iii) any interest on any Loan, or any fee or other amount payable under any Loan Documents, within five (5) days after the same becomes due.
- 10.1.2 <u>REIT and Financial Covenants</u>. Borrower or the REIT shall breach any covenant set forth in <u>Section 7.11</u> or in <u>Article IX</u> (excluding <u>Section 9.8</u>).
- 10.1.3 <u>Aggregate Occupancy Rate</u>. Borrower shall fail to satisfy the financial covenant regarding the Aggregate Occupancy Rate set forth in <u>Section 9.8</u> and such failure shall continue for sixty (60) days.
- 10.1.4 Other Defaults. Borrower or the REIT shall fail duly and punctually to perform or observe any agreement, covenant or obligation binding on Borrower or the REIT under this Agreement or under any of the other Loan Documents (other than as described in any other provision of this Section 10.1), and with respect to agreements, covenants or obligations for which no time period for performance is otherwise provided, such failure shall continue for fifteen (15) days after Borrower or the REIT knew of such failure (or such lesser period of time as is mandated by applicable Requirements of Law); provided, however, if such failure is capable of cure but is not capable of cure within such fifteen (15) day period, then if Borrower promptly undertakes action to cure such failure and thereafter diligently prosecutes such cure to completion within forty-five (45) days after Borrower or the REIT knew of such failure, then Borrower shall not be in default hereunder.
- 10.1.5 <u>Breach of Representation or Warranty</u>. Any representation or warranty made or deemed made by Borrower or the REIT to Agent or any Lender herein or in any of the other Loan Documents or in any statement, certificate or financial statements at any time given by Borrower pursuant to any of the Loan Documents shall be false or misleading in any material respect on the date as of which made.
- 10.1.6 <u>Default as to Other Indebtedness</u>. (i) Borrower, the REIT or any Unconsolidated Affiliate (unless such occurrence with respect to such Unconsolidated Affiliate will not result in a Material Adverse Effect on Borrower) shall have (A) failed to pay when due (beyond any applicable grace period), any amount in respect of any Indebtedness (other than Nonrecourse Indebtedness) of such party other than the Obligations if the aggregate amount of such other Indebtedness is Twenty Million Dollars (\$20,000,000) or more; or (B) otherwise defaulted (beyond any applicable grace period) under any Indebtedness of such party other than the Obligations if (1) the aggregate amount of such other Indebtedness is Twenty Million Dollars (\$20,000,000) or more, and (2) the holder of such Indebtedness has accelerated such Indebtedness; or (ii) any such other Indebtedness shall have otherwise become payable, or be required to be purchased or redeemed, prior to its scheduled maturity; or (iii) the holder(s) of any Lien, in any amount, commence foreclosure of such Lien upon any Property having an aggregate value in excess of Twenty Million Dollars (\$20,000,000); or (iv) any "Event of Default" shall exist under the Unsecured Term Note Indenture, or (v) any "Event of Default" shall exist under the Term Loan Agreement relating to the Fleet Term Loan.
  - 10.1.7 <u>Involuntary Bankruptcy; Appointment of Receiver, Etc.</u>
    - (a) An involuntary case shall be commenced against the REIT, Borrower or any Unconsolidated Affiliate (unless such occurrence with respect to such Unconsolidated Affiliate will not result in a Material Adverse Effect on Borrower), and the petition shall not be dismissed within sixty (60) days after commencement of the case, or a court having jurisdiction shall enter a decree or order for relief in respect of any such Person in an involuntary case, under any applicable bankruptcy, insolvency or other similar law now or hereinafter in effect; or any other similar relief shall be granted under any applicable federal, state or foreign law; or
      - (b) A decree or order of a court having jurisdiction in the premises for the appointment of a receiver,

liquidator, sequestrator, trustee, custodian or other officer having similar powers over the REIT, Borrower or any Unconsolidated Affiliate (unless such occurrence with respect to such Unconsolidated Affiliate will not result in a Material Adverse Effect on Borrower), or over all or a substantial part of the property of any such Person, shall be entered; or an interim receiver, trustee or other custodian of any such Person or of all or a substantial part of the property of any such Person, shall be appointed or a warrant of attachment, execution or similar process against any substantial part of the property of any such Person, shall be issued and any such event shall not be stayed, vacated, dismissed, bonded or discharged within sixty (60) days of entry, appointment or issuance.

- 10.1.8 <u>Voluntary Bankruptcy; Appointment of Receiver, Etc.</u> The REIT, Borrower, or any Unconsolidated Affiliate (unless such occurrence with respect to such Unconsolidated Affiliate will not result in a Material Adverse Effect on Borrower) shall have an order for relief entered with respect to it or commence a voluntary case under any applicable bankruptcy, insolvency or other similar law now or hereafter in effect, or shall consent to the entry of an order for relief in an involuntary case, or to the conversion of an involuntary case to a voluntary case, under any such law, or shall consent to the appointment of or taking of possession by a receiver, trustee or other custodian for all or a substantial part of its property; any such Person shall make any assignment for the benefit of creditors or shall be unable or fail, or admit in writing its inability, to pay its debts as such debts become due; or the general partner (or Person(s) serving in a similar capacity) of Borrower or the REIT's Board of Directors (or any committee thereof) adopts any resolution or otherwise authorizes any action to approve any of the foregoing.
- 10.1.9 <u>Judgments and Attachments</u>. (i) Any money judgment (other than a money judgment covered by insurance but only if the insurer has admitted liability with respect to such money judgment), writ or warrant of attachment, or similar process involving in any case an amount in excess of One Million Dollars (\$1,000,000) shall be entered or filed against the REIT, Borrower, or any Unconsolidated Affiliate (unless such occurrence with respect to such Unconsolidated Affiliate will not result in a Material Adverse Effect on Borrower) or their respective assets and shall remain undischarged, unvacated, unbonded or unstayed for a period of thirty (30) days, or (ii) any judgment or order of any court or administrative agency awarding material damages shall be entered against any such Person in any action under the Federal securities laws seeking rescission of the purchase or sale of, or for damages arising from the purchase or sale of, any Securities, such judgment or order shall have become final after exhaustion of all available appellate remedies and, in Agent's judgment, the payment of such judgment or order would have a Material Adverse Effect on such Person.
- 10.1.10 <u>Dissolution</u>. Any order, judgment or decree shall be entered against the REIT, Borrower, or any Unconsolidated Affiliate (unless such occurrence with respect to such Unconsolidated Affiliate will not result in a Material Adverse Effect on Borrower) decreeing its involuntary dissolution or split up and such order shall remain undischarged and unstayed for a period in excess of thirty (30) days; or the REIT or Borrower shall otherwise dissolve or cease to exist.
- 10.1.11 <u>Loan Documents; Failure of Subordination</u>. If for any reason (i) any Loan Document shall cease to be in full force and effect, or (ii) any Obligation shall be subordinated in right of payment to any other unsecured liability of the Borrower.
- 10.1.12 <u>ERISA Liabilities</u>. Any Termination Event occurs which will or is reasonably likely to subject Borrower, the REIT or any ERISA Affiliate to a liability which Agent reasonably determines will have a Material Adverse Effect on Borrower or the REIT, or the plan administrator of any Benefit Plan applies for approval under Section 412(d) of the Internal Revenue Code for a waiver of the minimum funding standards of Section 412(a) of the Internal Revenue Code and Agent reasonably determines that the business hardship upon which the Section 412(d) waiver was based will or would reasonably be anticipated to subject Borrower or the REIT to a liability which Agent determines will have a Material Adverse Effect on Borrower or the REIT.
- 10.1.13 <u>Environmental Liabilities</u>. Borrower, the REIT or any Subsidiary or Unconsolidated Affiliate (unless such occurrence with respect to such Unconsolidated Affiliate will not result in a Material Adverse Effect on Borrower) becomes subject to any Liabilities and Costs which Agent reasonably deems to have a Material Adverse Effect on such Person arising out of or related to the Release at any Property of any Contaminant into the environment, or any Remedial Action in response thereto, or any other violation of any Environmental Laws.
- 10.1.14 <u>Solvency; Material Adverse Change</u>. Borrower or the REIT shall cease to be Solvent, or there shall have occurred any material adverse change in the business, operations, properties, assets or condition (financial or otherwise) of Borrower or the REIT.

An Event of Default shall be deemed "continuing" until cured or waived in writing in accordance with <u>Section 12.4</u>.

## 10.2 <u>Rights and Remedies</u>.

10.2.1 Acceleration, Etc. Upon the occurrence of any Event of Default described in the foregoing Section 10.1.7 or 10.1.8 with respect to the REIT or Borrower, the Commitments shall automatically and immediately terminate and the unpaid principal amount of and any and all accrued interest on the Loans shall automatically become immediately due and payable, with all additional interest from time to time accrued thereon and without presentment, demand or protest or other requirements of any kind (including, without limitation, valuation and appraisement, diligence, presentment, notice of intent to demand or accelerate or notice of acceleration), all of which are hereby expressly waived by Borrower, and the obligations of Lenders to make any Loans hereunder shall thereupon terminate; and upon the occurrence and during the continuance of any other Event of Default, Agent shall, at the request, or may, with the consent of Requisite Lenders, by written notice to Borrower, (i) declare that the Commitments are terminated, whereupon the Commitments and the obligation of Lenders to make any Loan hereunder shall immediately

terminate, and/or (ii) declare the unpaid principal amount of, any and all accrued and unpaid interest on the Loans and all of the other Obligations to be, and the same shall thereupon be, immediately due and payable with all additional interest from time to time accrued thereon and without presentment, demand, or protest or other requirements of any kind (including without limitation, valuation and appraisement, diligence, presentment, notice of intent to demand or accelerate and of acceleration), all of which are hereby expressly waived by Borrower. Without limiting Agent's authority hereunder, on or after the Termination Date, Agent shall, at the request, or may, with the consent, of Requisite Lenders exercise any or all rights and remedies under the Loan Documents or applicable law. Upon the occurrence of and during the continuance of an Event of Default, Agent shall be entitled to request and receive, by or through Borrower or appropriate legal process, any and all information concerning the REIT, Borrower or any property of any of them, which is reasonably available to or obtainable by Borrower.

- 10.2.2 <u>Waiver of Demand</u>. Demand, presentment, protest and notice of nonpayment are hereby waived by Borrower. Borrower also waives, to the extent permitted by law, the benefit of all exemption laws.
- 10.2.3 <u>Waivers, Amendments and Remedies</u>. No delay or omission of Agent or Lenders to exercise any right under any Loan Document shall impair such right or be construed to be a waiver of any Event of Default or an acquiescence therein, and any single or partial exercise of any such right shall not preclude other or further exercise thereof or the exercise of any other right, and no waiver, amendment or other variation of the terms, conditions or provisions of the Loan Documents whatsoever shall be valid unless in writing signed by Agent after obtaining written approval thereof or the signature thereon of those Lenders required to approve such waiver, amendment or other variation, and then only to the extent in such writing specifically set forth. All remedies contained in the Loan Documents or by law afforded shall be cumulative and all shall be available to Agent and Lenders until the Obligations have been paid in full, the Commitments have expired or terminated and this Agreement has been terminated.
- 10.2.4 <u>Cash Collateral Account for Letters of Credit</u>. In the event that any Letters of Credit are in effect at the time of an acceleration of the maturity of the Loans, the amounts which shall thereupon become immediately due and payable by the Borrower shall include a sum equal to the aggregate face amount of such then effective Letters of Credit. Such sum shall be deposited in a cash collateral account to be opened by the Agent. Amounts held in such cash collateral account shall be applied by the Agent on each Drawing Date thereafter to pay any drafts presented pursuant to the Letters of Credit. After all Letters of Credit have been fully drawn upon, expired or otherwise terminated, any balance remaining in such cash collateral account shall be applied in the same manner as proceeds from the enforcement of the Loan Documents.
- 10.3 Rescission. If at any time after acceleration of the maturity of the Loans, Borrower shall pay all arrears of interest and all payments on account of principal of the Loans which shall have become due otherwise than by acceleration (with interest on principal and, to the extent permitted by law, on overdue interest, at the rates specified in this Agreement) and all Events of Default and Unmatured Events of Default (other than nonpayment of principal of and accrued interest on the Loans due and payable solely by virtue of acceleration) shall be remedied or waived pursuant to Section 12.4, then by written notice to Borrower, Requisite Lenders may elect, in their sole discretion, to rescind and annul the acceleration and its consequences; but such action shall not affect any subsequent Event of Default or Unmatured Event of Default or impair any right or remedy consequent thereon. The provisions of the preceding sentence are intended merely to bind Lenders to a decision which may be made at the election of Requisite Lenders; they are not intended to benefit Borrower and do not give Borrower the right to require Lenders to rescind or annul any acceleration hereunder, even if the conditions set forth herein are met.

## <u>ARTICLE XI</u>

### AGENCY PROVISIONS

## 11.1 Appointment.

- 11.1.1 Each Lender hereby (i) designates and appoints Fleet as Agent of such Lender under this Agreement and the Loan Documents, (ii) authorizes and directs Agent to enter into the Loan Documents other than this Agreement for the benefit of Lenders, and (iii) authorizes Agent to take such action on its behalf under the provisions of this Agreement and the Loan Documents and to exercise such powers as are set forth herein or therein, together with such other powers as are reasonably incidental thereto, subject to the limitations referred to in <a href="Sections 11.10.1">Sections 11.10.1</a> and <a href="11.10.2">11.10.2</a>. Agent agrees to act as such on the express conditions contained in this <a href="Article XI">Article XI</a>.
- 11.1.2 The provisions of this <u>Article XI</u> are solely for the benefit of Agent and Lenders, and Borrower shall not have any rights to rely on or enforce any of the provisions hereof (other than as expressly set forth in <u>Sections 11.3</u> and <u>11.9</u>, <u>provided</u>, <u>however</u>, that the foregoing shall in no way limit Borrower's obligations under this <u>Article XI</u>. In performing its functions and duties under this Agreement, Agent shall act solely as Agent of Lenders and does not assume and shall not be deemed to have assumed any obligation toward or relationship of agency or trust with or for Borrower or any other Person.
- 11.2 <u>Nature of Duties</u>. Agent shall not have any duties or responsibilities except those expressly set forth in this Agreement or in the Loan Documents. Subject to the provisions of <u>Sections 11.5</u> and <u>11.7</u>, Agent shall administer the Loans in the same manner as it administers its own loans. Promptly following the effectiveness of this Agreement, Agent shall send to each Lender its originally executed Note and the executed original, to the extent the same are available in sufficient numbers, of each other Loan Document other than the Notes in favor of other Lenders. Agent shall not have by reason of this Agreement a fiduciary relationship in respect of any Lender. Nothing in this Agreement or any of the Loan Documents, expressed or implied, is intended or shall be construed to impose upon Agent any obligation in respect of this Agreement or any of the Loan Documents except as expressly set forth herein or therein. Each Lender shall make its own independent investigation of the financial condition and affairs of the REIT, Borrower and each Portfolio Property in connection with the making and the continuance of the Loans

hereunder and shall make its own appraisal of the creditworthiness of the REIT and Borrower, and, except as specifically provided herein, Agent shall not have any duty or responsibility, either initially or on a continuing basis, to provide any Lender with any credit or other information with respect thereto, whether coming into its possession before the Closing Date or at any time or times thereafter.

#### 11.3 Loan Disbursements.

- Not later than 1:00 P.M. (Eastern time) on the next Business Day following receipt of a Notice of Borrowing. Agent shall send a copy thereof by facsimile to each other Lender and shall otherwise notify each Lender of the proposed Borrowing and the Funding Date. Each Lender shall make available to Agent (or the funding bank or entity designated by Agent), the amount of such Lender's Pro Rata Share of such Borrowing in immediately available funds not later than the times designated in <u>Section 11.3.2</u>. Unless Agent shall have been notified by any Lender prior to such time for funding in respect of any Borrowing that such Lender does not intend to make available to Agent such Lender's Pro Rata Share of such Borrowing, Agent may assume that such Lender has made such amount available to Agent. In any case where a Lender does not for any reason make available to Agent such Lender's Pro Rata Share of such Borrowing, Agent, in its sole discretion, may, but shall not be obligated to, fund to Borrower such Lender's Pro Rata Share of such Borrowing. If the amount so funded by Agent is not in fact made available to Agent by the responsible Lender, then Borrower agrees to repay to Agent such amount, together with interest thereon at the Base Rate for each day from the date such amount is made available to Borrower until the date such amount is repaid to Agent, not later than three (3) Business Days following Agent's demand to Borrower that such repayment be made. In addition, such Lender agrees to pay to Agent forthwith on demand such corresponding amount, together with interest thereon at the Federal Funds Rate. If such Lender shall pay to Agent such corresponding amount, such amount so paid shall constitute such Lender's Pro Rata Share of such Borrowing, and if both such Lender and Borrower shall have paid and repaid, respectively, such corresponding amount, Agent shall promptly return to Borrower such corresponding amount in same day funds; interest paid by Borrower in respect of such corresponding amount shall be prorated, as of the date of payment thereof by such Lender to Agent. In the event that Agent shall not have funded such Lender's Pro Rata Share under this Section 11.3.1, then Borrower shall not be obligated to accept a late funding of such Lender's Pro Rata Share if such funding is made more than two (2) Business Days following the applicable Funding Date. If Borrower declines to accept such delinquent funding, Agent shall promptly return to such Lender the amount of such funding. Nothing in this Section 11.3.1 shall alter the respective rights and obligations of the parties hereunder in respect of a Defaulting Lender or a Non-Pro Rata Loan.
- 11.3.2 Requests by Agent for funding by Lenders of Loans will be made by telecopy. Each Lender shall make the amount of its Loan available to Agent in Dollars and in immediately available funds, to such bank and account, as Agent may designate, not later than 1:00 P.M. (Eastern time) on the Funding Date designated in the Notice of Borrowing with respect to such Loan, provided that to the extent the Agent is late in providing any Lender with notice the applicable time in advance of any Funding Date specified in Section 2.1.2(a), such Lender shall not be obligated to make such amount available to the Agent until said time on the Business Day which is the same number of Business Days after the Funding Date as Agent was late in providing such notice.
- 11.3.3 Nothing in this Section 11.3 shall be deemed to relieve any Lender of its obligation hereunder to make its Pro Rata Share of Loans on any Funding Date, nor shall any Lender be responsible for the failure of any other Lender to perform its obligations to make any Loan hereunder, and the Commitment of any Lender shall not be increased or decreased as a result of the failure by any other Lender to perform its obligation to make a Loan.

### 11.4 <u>Distribution and Apportionment of Payments</u>.

- Subject to Section 11.4.2, payments actually received by Agent for the account of Lenders shall be paid to 11.4.1 them promptly after receipt thereof by Agent, but in any event within two (2) Business Days, provided that Agent shall pay to Lenders interest thereon, at the lesser of (i) Federal Funds Rate and (ii) the rate of interest applicable to such Loans, from the Business Day following receipt of such funds by Agent until such funds are paid in immediately available funds to Lenders. Subject to Section 11.4.2, all payments of principal and interest in respect of outstanding Loans, all payments of the fees described in this Agreement, and all payments in respect of any other Obligations shall be allocated among such of Lenders as are entitled thereto, in proportion to their respective Pro Rata Shares or otherwise as provided herein. Agent shall promptly distribute, but in any event within two (2) Business Days, to each Lender at its primary address set forth on the appropriate signature page hereof or on the Assignment and Assumption, or at such other address as a Lender may request in writing, such funds as it may be entitled to receive, provided that Agent shall in any event not be bound to inquire into or determine the validity, scope or priority of any interest or entitlement of any Lender and may suspend all payments and seek appropriate relief (including, without limitation, instructions from Requisite Lenders or all Lenders, as applicable, or an action in the nature of interpleader) in the event of any doubt or dispute as to any apportionment or distribution contemplated hereby. The order of priority herein is set forth solely to determine the rights and priorities of Lenders as among themselves and may at any time or from time to time be changed by Lenders as they may elect, in writing in accordance with <u>Section 12.4</u>, without necessity of notice to or consent of or approval by Borrower or any other Person. All payments or other sums received by Agent for the account of Lenders shall not constitute property or assets of Agent and shall be held by Agent, solely in its capacity as agent for itself and the other Lenders, subject to the Loan Documents.
- 11.4.2 Notwithstanding any provision hereof to the contrary, until such time as a Defaulting Lender has funded its Pro Rata Share of a Loan which was previously a Non Pro Rata Loan, or all other Lenders have received payment in full (whether by repayment or prepayment) of the principal and interest due in respect of such Non Pro Rata Loan, all of the Obligations owing to such Defaulting Lender hereunder shall be subordinated in right of payment, as provided in the following sentence, to the prior

payment in full of all principal, interest and fees in respect of all Non Pro Rata Loans in which the Defaulting Lender has not funded its Pro Rata Share (such principal, interest and fees being referred to as "Senior Loans"). All amounts paid by Borrower and otherwise due to be applied to the Obligations owing to the Defaulting Lender pursuant to the terms hereof shall be distributed by Agent to the other Lenders in accordance with their respective Pro Rata Shares (recalculated for purposes hereof to exclude the Defaulting Lender's Commitment), until all Senior Loans have been paid in full. This provision governs only the relationship among Agent, each Defaulting Lender, and the other Lenders; nothing hereunder shall limit the obligation of Borrower to repay all Loans in accordance with the terms of this Agreement. The provisions of this section shall apply and be effective regardless of whether an Event of Default occurs and is then continuing, and notwithstanding (i) any other provision of this Agreement to the contrary, (ii) any instruction of Borrower as to its desired application of payments or (iii) the suspension of such Defaulting Lender's right to vote on matters which are subject to the consent or approval of Requisite Lenders or all Lenders. No Facility Fee shall accrue in favor of, or be payable to, such Defaulting Lender from the date of any failure to fund Loans or reimburse Agent for any Liabilities and Costs as herein provided until such failure has been cured, and Agent shall be entitled to (1) withhold or setoff, and to apply to the payment of the defaulted amount and any related interest, any amounts to be paid to such Defaulting Lender under this Agreement, and (2) bring an action or suit against such Defaulting Lender in a court of competent jurisdiction to recover the defaulted amount and any related interest. In addition, the Defaulting Lender shall indemnify, defend and hold Agent and each of the other Lenders harmless from and against any and all Liabilities and Costs, plus interest thereon at the Default Rate, which they may sustain or incur by reason of or as a direct consequence of the Defaulting Lender's failure or refusal to abide by its obligations under this Agreement.

- 11.5 <u>Rights, Exculpation, Etc.</u> Neither Agent, any Affiliate of Agent, nor any of their respective officers, directors, employees, agents, attorneys or consultants, shall be liable to any Lender for any action taken or omitted by them hereunder or under any of the Loan Documents, or in connection herewith or therewith, except that Agent shall be liable for its gross negligence or willful misconduct. In the absence of gross negligence or willful misconduct, Agent shall not be liable for any apportionment or distribution of payments made by it in good faith pursuant to <u>Section 11.4</u>, and if any such apportionment or distribution is subsequently determined to have been made in error the sole recourse of any Person to whom payment was due, but not made, shall be to recover from the recipients of such payments any payment in excess of the amount to which they are determined to have been entitled. Agent shall not be responsible to any Lender for any recitals, statements, representations or warranties herein or for the execution, effectiveness, genuineness, validity, enforceability, collectibility or sufficiency of this Agreement or any of the other Loan Documents, or any of the transactions contemplated hereby and thereby; or for the financial condition of the REIT, Borrower or any of their Affiliates. Agent shall not be required to make any inquiry concerning either the performance or observance of any of the terms, provisions or conditions of this Agreement or any of the Loan Documents or the financial condition of the REIT, Borrower or any of their Affiliates, or the existence or possible existence of any Unmatured Event of Default or Event of Default.
- 11.6 <u>Reliance</u>. Agent shall be entitled to rely upon any written notices, statements, certificates, orders or other documents, telecopies or any telephone message believed by it in good faith to be genuine and correct and to have been signed, sent or made by the proper Person, and with respect to all matters pertaining to this Agreement or any of the Loan Documents and its duties hereunder or thereunder, upon advice of legal counsel (including counsel for Borrower), independent public accountant and other experts selected by it.
- <u>Indemnification</u>. To the extent that Agent is not reimbursed and indemnified by Borrower, Lenders will reimburse, within ten (10) Business Days after notice from Agent, and indemnify and defend Agent for and against any and all Liabilities and Costs (other than losses in the collection of principal and interest on the Loans which losses shall be shared among all Lenders including the Agent as provided in Sections 11.4 and 11.13) which may be imposed on, incurred by, or asserted against it in any way relating to or arising out of this ----- Agreement or any of the other Loan Documents or any action taken or omitted by Agent or under this Agreement or any of the other Loan Documents, in proportion to each Lender's Pro Rata Share; provided that no Lender shall be liable for any portion of such Liabilities and Costs resulting from Agent's gross negligence or willful misconduct or in respect of normal administrative costs and expenses incurred by Agent (prior to any Event of Default or any Unmatured Event of Default) in connection with its performance of administrative duties under this Agreement and the other Loan Documents. The obligations of Lenders under this Section 11.7 shall survive the payment in full of all Obligations and the termination of this Agreement. In the event that after payment and distribution of any amount by Agent to Lenders, any Lender or third party, including Borrower, any creditor of Borrower or a trustee in bankruptcy, recovers from Agent any amount found to have been wrongfully paid to Agent or disbursed by Agent to Lenders, then Lenders, in proportion to their respective Pro Rata Shares, shall reimburse Agent for all such amounts. Notwithstanding the foregoing, Agent shall not be obligated to advance Liabilities and Costs and may require the deposit by each Lender of its Pro Rata Share of any material Liabilities and Costs anticipated by Agent before they are incurred or made payable.
- 11.8 Agent Individually. With respect to its Pro Rata Share of the Commitments hereunder and the Loans made by it, Agent shall have and may exercise the same rights and powers hereunder and is subject to the same obligations and liabilities as and to the extent set forth herein for any other Lender. The terms "Lenders", "Requisite Lenders" or any similar terms may include Agent in its individual capacity as a Lender or one of the Requisite Lenders. Agent and any Lender and its Affiliates may accept deposits from, lend money to, and generally engage in any kind of banking, trust or other business with Borrower or any of its Affiliates as if it were not acting as Agent or Lender pursuant hereto.

## 11.9 Successor Agent; Resignation of Agent; Removal of Agent.

11.9.1 Agent may resign from the performance of all its functions and duties hereunder at any time by giving at least thirty (30) Business Days, prior written notice to Lenders and Borrower, and shall automatically cease to be Agent hereunder in the event a petition in bankruptcy shall be filed by or against Agent or the Federal Deposit Insurance Corporation or any other

Governmental Authority shall assume control of Agent or Agent's interests under the Facility. Further, Lenders whose aggregate Commitments constitute at least sixty-six and two-thirds percent (66-2/3%) of the Commitments of all Lenders excluding the Agent may remove Agent for cause at any time by giving at least thirty (30) Business Days' prior written notice to Agent, Borrower and all other Lenders. If Agent enters into one or more assignments pursuant to Section 11.12 having the effect of reducing its Commitment to less than \$20,000,000 then any Lender whose Commitment exceeds that of Agent may remove Agent by notice given within thirty (30) days after such Lender receives notice of the assignments which reduce the Agent's Commitment below such level. Such resignation or removal shall take effect upon the acceptance by a successor Agent appointed pursuant to Section 11.9.2 or 11.9.3.

- 11.9.2 Upon any such notice of resignation by or removal of Agent, Requisite Lenders shall appoint a successor Agent with the consent of Borrower, which consent shall not be unreasonably withheld or delayed <u>and</u> which consent shall not be required if there shall then exist any Event of Default. Any successor Agent must be a bank (i) the senior debt obligations of which (or such bank's parent's senior unsecured debt obligations) are rated not less than BBB by one of the Rating Agencies and (ii) which has total assets in excess of Ten Billion Dollars (\$10,000,000,000). Such successor Agent shall separately confirm in writing with Borrower the fee to be paid to such Agent pursuant to Section 2.5.2.
- 11.9.3 If a successor Agent shall not have been so appointed within said thirty (30) Business Day period, the retiring or removed Agent, shall then appoint a successor Agent who shall meet the requirements described in <u>Section 11.9.2</u> and who shall serve as Agent until such time, if any, as Requisite Lenders, appoint a successor Agent as provided above.
- 11.9.4 Bank of America, N.A. and Commerzbank AG, New York Branch are hereby appointed as Co-Syndication Agents for the Facility and Wachovia Bank National Association and PNC Bank, National Association are hereby appointed as Co-Documentation Agents for the Facility. Each of said Lenders shall continue to hold its respective title hereunder until the earlier to occur of (i) the date that it enters into one or more assignments pursuant to Section 11.12 having the effect of reducing its Commitment to less than \$20,000,000, or (ii) the date that it resigns its title by notice to the Agent and the Borrower.

## 11.10 Consent and Approvals.

- 11.10.1 Each of the following shall require the approval or consent of Requisite Lenders:
  - (a) Consent to Fundamental Changes (Section 8.1);
  - (b) Consent to Indebtedness that is otherwise prohibited (Section 8.6);
  - (c) Approval of additional executives upon a Discontinuity in Management (Section 8.8);
  - (d) Acceleration following an Event of Default (Section 10.2.1) or rescission of such acceleration (Section 10.3);
  - (e) Approval of the exercise of rights and remedies under the Loan Documents following an Event of Default (Section 10.2.1);
    - (f) Appointment of a successor Agent (Section 11.9);
  - (g) Except as referred to in <u>Section 11.10.2</u> or <u>11.11.1</u>, approval of any amendment, modification or termination of this Agreement, or waiver of any provision herein (<u>Section 12.4</u>).
- 11.10.2 Each amendment, modification or waiver specifically enumerated in <u>Section 12.4.1</u> shall require the consent of all Lenders.
- 11.10.3 In addition to the required consents or approvals referred to in Section 11.10.1, Agent may at any time request instructions from Requisite Lenders with respect to any actions or approvals which, by the terms of this Agreement or of any of the Loan Documents, Agent is permitted or required to take or to grant without instructions from any Lenders, and if such instructions are promptly requested, Agent shall be absolutely entitled to refrain from taking any action or to withhold any approval and shall not be under any liability whatsoever to any Person for refraining from taking any action or withholding any approval under any of the Loan Documents until it shall have received such instructions from Requisite Lenders. Without limiting the foregoing, no Lender shall have any right of action whatsoever against Agent as a result of Agent acting or refraining from acting under this Agreement or any of the other Loan Documents in accordance with the instructions of Requisite Lenders or, where applicable, all Lenders. Agent shall promptly notify each Lender at any time that the Requisite Lenders have instructed Agent to act or refrain from acting pursuant to this Section 11.10.3.
- 11.10.4 Each Lender agrees that any action taken by Agent at the direction or with the consent of Requisite Lenders in accordance with the provisions of this Agreement or any Loan Document, and the exercise by Agent at the direction or with the consent of Requisite Lenders of the powers set forth herein or therein, together with such other powers as are reasonably incidental thereto, shall be authorized and binding upon all Lenders, except for actions specifically requiring the approval of all Lenders. All communications from Agent to Lenders requesting Lenders' determination, consent, approval or disapproval (i) shall be given in the form of a written notice to each Lender, (ii) shall be accompanied by a description of the matter or thing as to which such determination, approval, consent or disapproval is requested, or shall advise each Lender where such matter or thing may be

inspected, or shall otherwise describe the matter or issue to be resolved, (iii) shall include, if reasonably requested by a Lender and to the extent not previously provided to such Lender, written materials and a summary of all oral information provided to Agent by Borrower in respect of the matter or issue to be resolved, and (iv) may include Agent's recommended course of action or determination in respect thereof. Each Lender shall reply promptly, but in any event within ten (10) Business Days (the "Lender Reply Period"). Unless a Lender shall give written notice to Agent that it objects to the recommendation or determination of Agent (together with a written explanation of the reasons behind such objection) within the Lender Reply Period, such Lender shall be deemed to have approved of or consented to such recommendation or determination. With respect to decisions requiring the approval of Requisite Lenders or all Lenders, Agent shall submit its recommendation or determination for approval of or consent to such recommendation or determination to all Lenders and upon receiving the required approval or consent shall follow the course of action or determination recommended to Lenders by Agent or such other course of action recommended by Requisite Lenders, and each non-responding Lender shall be deemed to have concurred with such recommended course of action.

### 11.11 <u>Agency Provisions Relating to Certain Enforcement Actions.</u>

- 11.11.1 Agent is hereby authorized on behalf of all Lenders, without the necessity of any notice to or further consent from any Lender, to waive the imposition of late fees provided for in <u>Section 2.4.5</u> up to a maximum of three (3) times during the term of this Agreement.
- 11.11.2 Should Agent (i) employ counsel for advice or other representation (whether or not any suit has been or shall be filed) with respect to any of the Loan Documents, or (ii) commence any proceeding or in any way seek to enforce its rights or remedies under the Loan Documents, each Lender, upon demand therefor from time to time, shall contribute its share (based on its Pro Rata Share) of the reasonable costs and/or expenses of any such advice or other representation, enforcement or acquisition, including, but not limited to, fees of receivers, court costs and fees and expenses of attorneys to the extent not otherwise reimbursed by Borrower; <u>provided</u> that Agent shall not be entitled to reimbursement of its attorneys' fees and expenses incurred in connection with the resolution of disputes between Agent and other Lenders unless Agent shall be the prevailing party in any such dispute. Any loss of principal and interest resulting from any Event of Default shall be shared by Lenders in accordance with their respective Pro Rata Shares.

### 11.12 Assignments and Participations.

- 11.12.1 Each Lender may assign, to one or more Eligible Assignees, all or a portion of its rights and obligations under this Agreement (including without limitation all or a portion of its Commitment and the Loans owing to it) and other Loan Documents; provided, however, that (i) each such assignment shall be of a constant, and not a varying, percentage of the assigning Lender's rights and obligations under this Agreement and other Loan Documents, and the assignment shall cover the same percentage of such Lender's Commitment and Loans, (ii) unless Agent and Borrower otherwise consent (except that after an Event of Default only the consent of Agent shall be required), the aggregate amount of the Commitment of the assigning Lender being assigned pursuant to each such assignment (determined as of the date of the Assignment and Assumption with respect to such assignment) shall in no event be less than Ten Million Dollars (\$10,000,000) and shall be an integral multiple of One Million Dollars (\$1,000,000), (iii) after giving effect to such assignment, the aggregate amount of the Commitment retained by the assigning Lender shall in no event be less than Twelve Million Dollars (\$12,000,000), (iv) the parties to each such assignment shall execute and deliver to Agent, for its approval and acceptance, an Assignment and Assumption, and (v) Agent shall receive from the assignor a processing fee of Three Thousand Dollars (\$3,000). Upon such execution, delivery, approval and acceptance, and upon the effective date specified in the applicable Assignment and Assumption, (X) the assignee thereunder shall be a party hereto and, to the extent that rights and obligations hereunder have been assigned to it pursuant to such Assignment and Assumption, have the rights and obligations of a Lender hereunder, and (Y) the assigning Lender thereunder shall, to the extent that rights and obligations hereunder have been assigned by it pursuant to such Assignment and Assumption, relinquish its rights and be released from its obligations under this Agreement.
- By executing and delivering an Assignment and Assumption, the assigning Lender thereunder and the assignee thereunder confirm to and agree with each other and the other parties hereto as follows: (i) other than as provided in such Assignment and Assumption, such assigning Lender makes no representation or warranty and assumes no responsibility with respect to any statements, warranties or representations made in or in connection with this Agreement or any other Loan Document or the execution, legality, validity, enforceability, genuineness, sufficiency or value of this Agreement or any other Loan Document or any other instrument or document furnished pursuant hereto; (ii) such assigning Lender makes no representation or warranty and assumes no responsibility with respect to the financial condition of the REIT or Borrower or the performance or observance by the REIT or Borrower of any of their respective obligations under any Loan Document or any other instrument or document furnished pursuant hereto; (iii) such assignee confirms that it has received a copy of this Agreement, together with copies of the financial statements referred to in <u>Article V</u> or delivered pursuant to <u>Article VI</u> to the date of such assignment and such other Loan Documents and other documents and information as it has deemed appropriate to make its own credit analysis and decision to enter into such Assignment and Assumption; (iv) such assignee will, independently and without reliance upon Agent, such assigning Lender or any other Lender and based on such documents and information as it shall deem appropriate at the time, continue to make its own credit decisions in taking or not taking action under this Agreement; (v) such assignee appoints and authorizes Agent to take such action as Agent on its behalf and to exercise such powers under this Agreement and the other Loan Documents as are delegated to Agent by the terms hereof and thereof, together with such powers as are reasonably incidental thereto; and (vi) such assignee agrees that it will perform in accordance with their terms all of the obligations which by the terms of this Agreement are required to be performed by it as a Lender.

- 11.12.3 Agent shall maintain, at its address referred to on the counterpart signature pages hereof, a copy of each Assignment and Assumption delivered to and accepted by it and shall record in the Loan Account the names and addresses of each Lender and the Commitment of, and principal amount of the Loans owing to, such Lender from time to time. Borrower, Agent and Lenders may treat each Person whose name is recorded in the Loan Account as a Lender hereunder for all purposes of this Agreement.
- 11.12.4 Upon its receipt of an Assignment and Assumption executed by an assigning Lender and an assignee, Agent shall, if such Assignment and Assumption has been properly completed and is in substantially the form of Exhibit A, (i) accept such Assignment and Assumption, (ii) record the information contained therein in the Loan Account, and (iii) give prompt notice thereof to Borrower. Upon request, Borrower will execute and deliver to Agent an appropriate replacement promissory note or replacement promissory notes in favor of each assignee (and assignor, if such assignor is retaining a portion of its Commitment and Loans) reflecting such assignee's (and assigner's) Pro Rata Share(s) of the Facility. Upon execution and delivery of such replacement promissory notes the original promissory note or notes evidencing all or a portion of the Commitments and Loans being assigned shall be cancelled and returned to Borrower.
- 11.12.5 Each Lender may sell participations to one or more banks or other entities in or to all or a portion of its rights and obligations under this Agreement (including without limitation all or a portion of its Commitment and the Loans owing to it) and other Loan Documents; provided, however, that (i) such Lender's obligations under this Agreement (including without limitation its Commitment to Borrower hereunder) shall remain unchanged, (ii) such Lender shall remain solely responsible to the other parties hereto for the performance of such obligations, (iii) Borrower, Agent and the other Lenders shall continue to deal solely and directly with such Lender in connection with such Lender's rights and obligations under this Agreement and with regard to any and all payments to be made under this Agreement, and (iv) the holder of any such participation shall not be entitled to voting rights under this Agreement except for voting rights with respect to (A) increases in the Facility; (B) extensions of the Termination Date; and (C) decreases in the interest rates described in this Agreement. No participant shall be entitled to vote on any matter until the Lender with which such participant is participating in the Facility and the Loans confirms such participant's status as a participant hereunder.
- 11.12.6 Borrower will use reasonable efforts to cooperate with Agent and Lenders in connection with the assignment of interests under this Agreement or the sale of participations herein.
- 11.12.7 Anything in this Agreement to the contrary notwithstanding, and without the need to comply with any of the formal or procedural requirements of this Agreement, including this <u>Section 11.12</u>, any Lender may at any time and from time to time pledge or assign all or any portion of its interest and rights under this Agreement (including all or any portion of its Note) to any of the twelve Federal Reserve Banks organized under Section 4 of the Federal Reserve Act, 12 U.S.C. §341. No such pledge or the enforcement thereof shall release the pledgor Lender from its obligations hereunder or under any of the other Loan Documents.
- 11.12.8 Anything in this Agreement to the contrary notwithstanding, any Lender may assign all or any portion of its rights and obligations under this Agreement to another branch or Affiliate of such Lender, <u>provided</u> that (i) at the time of such assignment such Lender is not a Defaulting Lender, (ii) such Lender gives Agent and Borrower at least fifteen (15) days' prior written notice of any such assignment, (iii) the parties to each such assignment execute and deliver to Agent an Assignment and Assumption, and (iv) Agent receives from assignor a processing fee of Three Thousand Dollars (\$3,000).
- 11.12.9 No assignee of any rights and obligations under this Agreement shall be permitted to subassign such rights and obligations.
- 11.12.10 No Lender shall be permitted to assign or sell all or any portion of its rights and obligations under this Agreement to Borrower or any Affiliate of Borrower.
- amounts received by them which are applicable to the payment of the Obligations, equitable adjustment will be made so that, in effect, all such amounts will be shared among them ratably in accordance with their Pro Rata Shares, whether received by voluntary payment, by counterclaim or cross action or by the enforcement of any or all of the Obligations, (ii) if any of them shall by voluntary payment or by the exercise of any right of counterclaim or otherwise, receive payment of a proportion of the aggregate amount of the Obligations held by it which is greater than its Pro Rata Share of the payments on account of the Obligations, the one receiving such excess payment shall purchase, without recourse or warranty, an undivided interest and participation (which it shall be deemed to have done simultaneously upon the receipt of such payment) in such Obligations owed to the others so that all such recoveries with respect to such Obligations shall be applied ratably in accordance with their Pro Rata Shares; provided, that if all or part of such excess payment received by the purchasing party is thereafter recovered from it, those purchases shall be rescinded and the purchase prices paid for such participations shall be returned to that party to the extent necessary to adjust for such recovery, but without interest except to the extent the purchasing party is required to pay interest in connection with such recovery. Borrower agrees that any Lender so purchasing a participation from another Lender pursuant to this Section 11.13 may, to the fullest extent permitted by law, exercise all its rights of payment with respect to such participation as fully as if such Lender were the direct creditor of Borrower in the amount of such participation.
- 11.14 <u>Delivery of Documents</u>. Agent shall as soon as reasonably practicable distribute to each Lender at its primary address set forth on the appropriate counterpart signature page hereof, or at such other address as a Lender may request in writing, (i) all documents to which such Lender is a party or of which such Lender is a beneficiary set forth in <u>Section 4.1</u>, (ii) all documents of which Agent receives copies from Borrower pursuant to <u>Section 6.1</u> (except as provided in <u>Section 6.1.1</u>) and <u>Section 12.6</u>, (iii) all other documents or information which Agent is required to send to Lenders pursuant to the terms of this Agreement;

(iv) other information or documents received by Agent at the request of any Lender, and (v) all notices received by Agent pursuant to <u>Section 6.2</u>. In addition, within fifteen (15) Business Days after receipt of a request in writing from a Lender for written information or documents provided by or prepared by Borrower or the REIT, Agent shall deliver such written information or documents to such requesting Lender if Agent has possession of such written information or documents in its capacity as Agent or as a Lender.

11.15 Notice of Events of Default. Agent shall not be deemed to have knowledge or notice of the occurrence of any Unmatured Event of Default or Event of Default (other than nonpayment of principal of or interest on the Loans) unless Agent has received notice in writing from a Lender or Borrower referring to this Agreement or the other Loan Documents, describing such event or condition and expressly stating that such notice is a notice of an Unmatured Event of Default or Event of Default. Should Agent receive such notice of the occurrence of an Unmatured Event of Default or Event of Default, or should Agent send Borrower a notice of Unmatured Event of Default or Event of Default, Agent shall promptly give notice thereof to each Lender.

#### <u>ARTICLE XII</u>

#### MISCELLANEOUS

### 12.1 Expenses.

- 12.1.1 <u>Generally</u>. Borrower agrees upon demand to pay, or reimburse Agent for, all of Agent's external audit, legal and investigation expenses and for all other reasonable out-of-pocket costs and expenses of every type and nature (including, without limitation, the reasonable fees, expenses and disbursements of Agent's internal appraisers, environmental advisors or legal counsel) incurred by Agent at any time (whether prior to, on or after the date of this Agreement) in connection with (i) its own audit and investigation of Borrower and the Portfolio Properties provided that prior to an Event of Default Borrower shall not be required to reimburse expenses for any inspections of the Portfolio Properties by Agent's loan officers or other employees which are made more frequently than annually; (ii) the negotiation, preparation and execution of this Agreement (including, without limitation, the satisfaction or attempted satisfaction of any of the conditions set forth in <u>Article IV</u>) and the other Loan Documents and the making of the Loans; (iii) administration of this Agreement, the other Loan Documents and the Loans, including, without limitation, consultation with attorneys in connection therewith; and (iv) the protection, collection or enforcement of any of the Obligations.
- 12.1.2 After Event of Default. Borrower further agrees to pay, or reimburse Agent and Lenders, for all reasonable out-of-pocket costs and expenses, including without limitation reasonable attorneys' fees and disbursements incurred by Agent or Lenders after the occurrence of an Event of Default (i) in enforcing any Obligation or exercising or enforcing any other right or remedy available by reason of such Event of Default; (ii) in connection with any refinancing or restructuring of the credit arrangements provided under this Agreement in the nature of a "work-out" or in any insolvency or bankruptcy proceeding; (iii) in commencing, defending or intervening in any litigation or in filing a petition, complaint, answer, motion or other pleadings in any legal proceeding relating to Borrower, the REIT or any Affiliate and related to or arising out of the transactions contemplated hereby; or (iv) in taking any other action in or with respect to any suit or proceeding (whether in bankruptcy or otherwise).
- <u>Indemnity</u>. Borrower further agrees to defend, protect, indemnify and hold harmless Agent, each and all of the Lenders, each of their respective Affiliates and participants and each of the respective officers, directors, employees, agents, attorneys and consultants (including, without limitation, those retained in connection with the satisfaction or attempted satisfaction of any of the conditions set forth in Article IV) of each of the foregoing (collectively called the "Indemnitees") from and against any and all Liabilities and Costs imposed on, incurred by, or asserted against such Indemnitees (whether based on any federal or state laws or other statutory regulations, including, without limitation, securities and commercial laws and regulations, under common law or in equity, and based upon contract or otherwise, including any Liabilities and Costs arising as a result of a "prohibited transaction" under ERISA to the extent arising from or in connection with the past, present or future operations of the REIT or Borrower or their respective predecessors in interest) in any manner relating to or arising out of this Agreement or the other Loan Documents, or any act, event or transaction related or attendant thereto, the making of and participation in the Loans and the management of the Loans, or the use or intended use of the proceeds of the Loans (collectively, the "Indemnified Matters"); provided, however, that Borrower shall have no obligation to an Indemnitee hereunder with respect to (i) matters for which such Indemnitee has been compensated pursuant to or for which an exemption is provided in <u>Section 2.4.7</u> or any other provision of this Agreement, (ii) Indemnified Matters to the extent caused by or resulting from the willful misconduct or gross negligence of that Indemnitee, as determined by a court of competent jurisdiction, and (iii) Indemnified Matters arising from any dispute among the Lenders not attributable to the actions or omissions of Borrower or the REIT. To the extent that the undertaking to indemnify, pay and hold harmless set forth in the preceding sentence may be unenforceable because it is violative of any law or public policy, Borrower shall contribute the maximum portion which it is permitted to pay and satisfy under applicable law, to the payment and satisfaction of all Indemnified Matters incurred by the Indemnitees.
- 12.3 <u>Change in Accounting Principles</u>. Except as otherwise provided herein, if any changes in accounting principles from those used in the preparation of the most recent financial statements delivered to Agent pursuant to the terms hereof are hereinafter required or permitted by the rules, regulations, pronouncements and opinions of the Financial Accounting Standards Board or the American Institute of Certified Public Accountants (or successors thereto or agencies with similar functions) and are adopted by the REIT or Borrower with the agreement of its independent certified public accountants and such changes result in a change in the method of calculation of any of the financial covenants, standards or terms found herein, the parties hereto agree to enter into negotiations in order to amend such provisions so as to equitably reflect such changes with the desired result that the criteria for evaluating the financial condition of the REIT or Borrower shall be the same after such changes as if such changes had not been made; <u>provided</u>, <u>however</u>, that no change in GAAP that would affect the method of calculation of any of the financial covenants,

standards or terms shall be given effect in such calculations until such provisions are amended pursuant to <u>Section 12.4</u>, to so reflect such change in accounting principles.

- Amendments and Waivers. (i) No amendment or modification of any provision of this Agreement shall be effective without the written agreement of Requisite Lenders (after notice to all Lenders) and Borrower (except for amendments to Section 11.4.1 which do not require the consent of Borrower), and (ii) no termination or waiver of any provision of this Agreement, or consent to any departure by Borrower therefrom (except as expressly provided in Section 11.11.1 with respect to waivers of late fees), shall in any event be effective without the written concurrence of Requisite Lenders (after notice to all Lenders), which Requisite Lenders shall have the right to grant or withhold at their sole discretion, except that:
  - 12.4.1 The following amendments, modifications or waivers shall require the consent of all Lenders:
    - (a) increasing the Facility or increasing any Lender's Commitment or decreasing any Lender's Commitment (other than a pro-rata decrease of all Lender's Commitments) without the consent of the affected Lender;
      - (b) changing the principal amount or final maturity of the Loans;
      - (c) reducing the interest rates applicable to the Loans;
      - (d) reducing the rates on which fees payable pursuant hereto are determined;
      - (e) forgiving or delaying any amount payable or receivable under <u>Article II</u> (other than late fees);
      - (f) changing the definition of "Requisite Lenders", "Pro Rata Shares" or "Event of Default";
      - (g) changing any provision contained in this <u>Section 12.4</u>;
    - (h) releasing any obligor or guarantor under any Loan Document or amending the Guaranty to reduce the guarantor's liability thereunder;
    - (i) consent to assignment by Borrower of all of its duties and Obligations hereunder pursuant to Section 12.14; or
    - (j) changing any of the financial covenants set forth in Article IX or any of the definitions used in the computation of such covenants or waiving any failure of the Borrower to comply with any one of such covenants for two or more consecutive Fiscal Quarters.
- 12.4.2 No amendment, modification, termination or waiver of any provision of <u>Article XI</u> or any other provision referring to Agent shall be effective without the written concurrence of Agent, but only if such amendment, modification, termination or waiver alters the obligations or rights of Agent.

Any waiver or consent shall be effective only in the specific instance and for the specific purpose for which it was given. No notice to or demand on Borrower in any case shall entitle Borrower to any other further notice or demand in similar or other circumstances. Any amendment, modification, termination, waiver or consent effected in accordance with this <u>Section 12.4</u> shall be binding on each assignee, transferee or recipient of Agent's or any Lender's Commitment under this Agreement or the Loans at the time outstanding.

- 12.5 <u>Independence of Covenants</u>. All covenants hereunder shall be given independent effect so that if a particular action or condition is not permitted by any of such covenants, the fact that it would be permitted by an exception to, or be otherwise within the limitations of, another covenant shall not avoid the occurrence of an Event of Default or Unmatured Event of Default if such action is taken or condition exists, and if a particular action or condition is expressly permitted under any covenant, unless expressly limited to such covenant, the fact that it would not be permitted under the general provisions of another covenant shall not constitute an Event of Default or Unmatured Event of Default if such action is taken or condition exists.
- 12.6 <u>Notices and Delivery</u>. Unless otherwise specifically provided herein, any consent, notice or other communication herein required or permitted to be given shall be in writing and may be personally served, telecopied or sent by courier service or United States mail and shall be deemed to have been given when delivered in person or by courier service, upon receipt of a telecopy (or on the next Business Day if such telecopy is received on a non-Business Day or after 5:00 p.m. (at the office of the recipient) on a Business Day) or four (4) Business Days after deposit in the United States mail (registered or certified, with postage prepaid and properly addressed). Notices to Agent pursuant to <u>Article II</u> shall not be effective until received by Agent. For the purposes hereof, the addresses of the parties hereto (until notice of a change thereof is delivered as provided in this <u>Section 12.6</u>) shall be as set forth below each party's name on the signature pages hereof, or, as to each party, at such other address as may be designated by such party in a written notice to all of the other parties. All deliveries to be made to Agent for distribution to the Lenders shall be made to Agent at the addresses specified for notice on the signature page hereto and in addition, a sufficient number of copies of each such delivery shall be delivered to Agent for delivery to each Lender at the address specified for deliveries on the signature page hereto or such other address as may be designated by Agent in a written notice.

- 12.7 <u>Survival of Warranties, Indemnities and Agreements</u>. All agreements, representations, warranties and indemnities made or given herein shall survive the execution and delivery of this Agreement and the other Loan Documents and the making and repayment of the Loans hereunder and such indemnities shall survive termination hereof.
- 12.8 <u>Failure or Indulgence Not Waiver; Remedies Cumulative</u>. No failure or delay on the part of Agent or any Lender in the exercise of any power, right or privilege under any of the Loan Documents shall impair such power, right or privilege or be construed to be a waiver of any default or acquiescence therein, nor shall any single or partial exercise of any such power, right or privilege preclude other or further exercise thereof or of any other right, power or privilege. All rights and remedies existing under the Loan Documents are cumulative to and not exclusive of any rights or remedies otherwise available.
- 12.9 <u>Setoff; Payments Set Aside</u>. Borrower hereby grants to each Lender, a continuing lien, security interest and right of setoff as security for all liabilities and obligations to such Lender hereunder, whether now existing or hereafter arising, upon and against all deposits, credits, collateral and property, now or hereafter in the possession, custody, safekeeping or control of such Lender or in transit to it. At any time, without demand or notice (any such notice being expressly waived by Borrower), any Lender may, WITH THE PRIOR APPROVAL OF THE AGENT (which approval will not be unreasonably withheld), setoff the same or any part thereof and apply the same to any liability or obligation of Borrower hereunder even though unmatured and regardless of the adequacy of any other remedies to collect or collateral securing the Obligations. To the extent that Borrower makes a payment or payments to Agent or the Lenders or Agent or the Lenders exercise their rights of setoff, and such payment or payments or the proceeds of such setoff or any part thereof are subsequently invalidated, declared to be fraudulent or preferential, set aside and/or required to be repaid to a trustee, receiver or any other party under any bankruptcy law, state or federal law, common law or equitable cause, then to the extent of such recovery, the Obligation or part thereof originally intended to be satisfied, and all rights and remedies therefor, shall be revived and continued in full force and effect as if such payment had not been made or such setoff had not occurred.
- 12.10 <u>Severability</u>. In case any provision in or obligation under this Agreement or the other Loan Documents shall be invalid, illegal or unenforceable in any jurisdiction, the validity, legality and enforceability of the remaining provisions or obligations, or of such provision or obligation in any other jurisdiction, shall not in any way be affected or impaired thereby, <u>provided</u>, <u>however</u>, that if the rates of interest or any other amount payable hereunder, or the collectibility thereof, are declared to be or become invalid, illegal or unenforceable, Lenders' obligations to make Loans shall not be enforceable.
- 12.11 <u>Heading</u>. Section headings in this Agreement are included herein for convenience of reference only and shall not constitute a part of this Agreement for any other purpose or be given any substantive effect.
- 12.12 <u>Governing Law.</u> THIS AGREEMENT AND THE RIGHTS AND OBLIGATIONS OF THE PARTIES HEREUNDER SHALL BE GOVERNED BY, AND SHALL BE CONSTRUED AND INTERPRETED IN ACCORDANCE WITH THE LAWS OF THE COMMONWEALTH OF MASSACHUSETTS (EXCLUDING THE LAWS APPLICABLE TO CONFLICTS OR CHOICE OF LAW).
- 12.13 <u>Limitation of Liability</u>. To the extent permitted by applicable law, no claim may be made by Borrower, any Lender or any other Person against Agent, Co-Agent or any Lender, or the affiliates, directors, officers, employees, attorneys or agents of any of them, for any special, indirect, consequential or punitive damages in respect of any claim for breach of contract or any other theory of liability arising out of or related to the transactions contemplated by this Agreement, or any act, omission or event occurring in connection therewith; and Borrower and each Lender hereby waive, release and agree not to sue upon any claim for any such damages, whether or not accrued and whether or not known or suspected to exist in its favor.
- 12.14 <u>Successors and Assigns</u>. This Agreement and the other Loan Documents shall be binding upon the parties hereto and their respective successors and assigns and shall inure to the benefit of the parties hereto and the successors and permitted assigns of Agent and Lenders. The terms and provisions of this Agreement shall inure to the benefit of any assignee or transferee of the Loans and the Commitments of Lenders under this Agreement, and in the event of such transfer or assignment, the rights and privileges herein conferred upon Agent and Lenders shall automatically extend to and be vested in such transferee or assignee, all subject to the terms and conditions hereof. Borrower's rights or any interest therein hereunder, and Borrower's duties and Obligations hereunder, shall not be assigned without the consent of all Lenders.
- Consent to Jurisdiction and Service of Process; Waiver of Jury Trial and Certain Damage Claims. ALL JUDICIAL PROCEEDINGS BROUGHT AGAINST BORROWER WITH RESPECT TO THIS AGREEMENT OR ANY OTHER LOAN DOCUMENT MAY BE, AND ALL JUDICIAL PROCEEDINGS BROUGHT BY BORROWER WITH RESPECT TO THIS AGREEMENT OR ANY OTHER LOAN DOCUMENT SHALL BE, BROUGHT IN ANY STATE OR FEDERAL COURT OF COMPETENT JURISDICTION HAVING SITUS WITHIN THE COMMONWEALTH OF MASSACHUSETTS, AND BY EXECUTION AND DELIVERY OF THIS AGREEMENT, BORROWER ACCEPTS, FOR ITSELF AND IN CONNECTION WITH ITS PROPERTIES, GENERALLY AND UNCONDITIONALLY, THE JURISDICTION OF THE AFORESAID COURTS, AND IRREVOCABLY AGREES TO BE BOUND BY ANY FINAL JUDGMENT RENDERED THEREBY FROM WHICH NO APPEAL HAS BEEN TAKEN OR IS AVAILABLE. BORROWER IRREVOCABLY CONSENTS TO THE SERVICE OF PROCESS OF ANY OF THE AFOREMENTIONED COURTS IN ANY SUCH ACTION OR PROCEEDING BY THE MAILING OF COPIES THEREOF BY REGISTERED OR CERTIFIED MAIL, POSTAGE PREPAID, TO ITS NOTICE ADDRESS SPECIFIED ON THE SIGNATURE PAGES HEREOF. BORROWER, AGENT AND LENDERS EACH IRREVOCABLY WAIVES ANY OBJECTION (INCLUDING, WITHOUT LIMITATION, ANY OBJECTION OF THE LAYING OF VENUE OR BASED ON THE GROUNDS OF FORUM NON CONVENIENS) WHICH IT MAY NOW OR HEREAFTER HAVE TO THE BRINGING OF ANY SUCH ACTION OR PROCEEDING WITH RESPECT TO THIS AGREEMENT OR ANY OTHER LOAN DOCUMENT IN ANY JURISDICTION SET FORTH ABOVE. EXCEPT TO THE EXTENT EXPRESSLY

PROHIBITED BY LAW, THE BORROWER HEREBY WAIVES ANY RIGHT IT MAY HAVE TO CLAIM OR RECOVER IN ANY ACTION OR PROCEEDING REFERRED TO IN THE PRECEDING SENTENCE ANY SPECIAL, EXEMPLARY, PUNITIVE OR CONSEQUENTIAL DAMAGES OR ANY DAMAGES OTHER THAN, OR IN ADDITION TO, ACTUAL DAMAGES, INCLUDING ANY DAMAGES PURSUANT TO M.G.L. C.93A ET SEQ. NOTHING HEREIN SHALL AFFECT THE RIGHT TO SERVE PROCESS IN ANY OTHER MANNER PERMITTED BY LAW OR SHALL LIMIT THE RIGHT OF AGENT OR ANY LENDER TO BRING PROCEEDINGS AGAINST BORROWER IN THE COURTS OF ANY OTHER JURISDICTION. THE BORROWER, THE AGENT AND EACH LENDER HEREBY MUTUALLY HEREBY KNOWINGLY, VOLUNTARILY AND INTENTIONALLY WAIVE THE RIGHT TO A TRIAL BY JURY IN RESPECT OF ANY CLAIM BASED HEREON, ARISING OUT OF, UNDER OR IN CONNECTION WITH THIS AGREEMENT OR ANY OTHER LOAN DOCUMENTS CONTEMPLATED TO BE EXECUTED IN CONNECTION HEREWITH OR ANY COURSE OF CONDUCT, COURSE OF DEALINGS, STATEMENTS (WHETHER VERBAL OR WRITTEN) OR ACTIONS OF ANY PARTY, INCLUDING, WITHOUT LIMITATION, ANY COURSE OF CONDUCT, COURSE OF DEALINGS, STATEMENTS OR ACTIONS OF AGENT OR ANY LENDER RELATING TO THE ADMINISTRATION OF THE LOANS OR ENFORCEMENT OF THE LOAN DOCUMENTS, AND AGREE THAT NO PARTY WILL SEEK TO CONSOLIDATE ANY SUCH ACTION WITH ANY OTHER ACTION IN WHICH A JURY TRIAL CANNOT BE OR HAS NOT BEEN WAIVED. EXCEPT AS PROHIBITED BY LAW, BORROWER HEREBY WAIVES ANY RIGHT IT MAY HAVE TO CLAIM OR RECOVER IN ANY LITIGATION ANY SPECIAL, EXEMPLARY, PUNITIVE OR CONSEQUENTIAL DAMAGES OR ANY DAMAGES OTHER THAN, OR IN ADDITION TO, ACTUAL DAMAGES. BORROWER CERTIFIES THAT NO REPRESENTATIVE, AGENT OR ATTORNEY OF AGENT OR ANY LENDER HAS REPRESENTED, EXPRESSLY OR OTHERWISE, THAT AGENT OR ANY LENDER WOULD NOT, IN THE EVENT OF LITIGATION, SEEK TO ENFORCE THE FOREGOING WAIVER. THIS WAIVER CONSTITUTES A MATERIAL INDUCEMENT FOR LENDERS TO PROVIDE THE COMMITMENTS AND MAKE THE LOANS.

- 12.16 <u>Counterparts; Effectiveness; Inconsistencies</u>. This Agreement and any amendments, waivers, consents or supplements may be executed in counterparts, each of which when so executed and delivered shall be deemed an original, but all such together shall constitute but one and the same instrument. This Agreement shall become effective when Borrower, the initial Lenders and Agent have duly executed and delivered execution pages of this Agreement to each other (delivery by Borrower to Lenders and by any Lender to Borrower and any other Lender being deemed to have been made by delivery to Agent). This Agreement and each of the other Loan Documents shall be construed to the extent reasonable to be consistent one with the other, but to the extent that the terms and conditions of this Agreement are actually and directly inconsistent with the terms and conditions of any other Loan Document, this Agreement shall govern.
- 12.17 <u>Construction</u>. The parties acknowledge that each party and its counsel have reviewed and revised this Agreement and that the normal rule of construction to the effect that any ambiguities are to be resolved against the drafting party shall not be employed in the interpretation of this Agreement or any amendments or exhibits hereto.
- 12.18 <u>Obligations Unsecured</u>. It is the intent of the parties that the Obligations shall constitute unsecured obligations of Borrower. Neither the restrictions and prohibitions set forth herein with respect to the creation, incurrence, assumption or existence of any Lien on any Property of Borrower or any other Person, nor those set forth in any other Loan Document, are intended to create or constitute a Lien of any nature upon any Property of Borrower or any other Person, and no such restriction or prohibition shall be deemed to constitute any such Lien. This <u>Section 12.18</u> shall not be deemed to prevent the Agent or any Lender from obtaining a Lien as security for the Obligations at any time hereafter pursuant to a mutual agreement among the parties hereto expressly providing for such Lien or during the continuance of any Event of Default.
- 12.19 Replacement of Loan Notes. Upon receipt of an affidavit of an officer of any Lender as to the loss, theft, destruction or mutilation of its Loan Note or any other security document which is not of public record, and, in the case of any such loss, theft, destruction or mutilation, upon cancellation of such Loan Note or other security document, Borrower will issue, in lieu thereof, a replacement Loan Note or other security document in the same principal amount thereof and otherwise of like tenor. The Lender requesting any such replacement Loan Note shall agree to indemnify and hold harmless Borrower from any loss or liability arising from any claims made upon the cancelled Loan Note that was replaced by such replacement Loan Note.
- 12.20 <u>Entire Agreement</u>. This Agreement, taken together with all of the other Loan Documents and all certificates and other documents delivered by Borrower to Agent (including documents incorporating separate agreements relating to the payment of fees), embodies the entire agreement and supersede all prior agreements, written and oral, relating to the subject matter hereof.

IN WITNESS WHEREOF, this Amended and Restated Credit Agreement has been duly executed on the date set forth above.

| BORROWER: | CPG PARTNERS, L.P., a Delaware limited partnership |
|-----------|--|
|           |  |
|           |  |

By: CHELSEA PROPERTY GROUP, INC., a Maryland corporation, its general partner

| Ву  |  |  |  |
|-----|--|--|--|
| Its |  |  |  |

## ADDRESS FOR NOTICE AND DELIVERY:

103 Eisenhower Parkway Roseland, NJ 07068 Attn: Michael J. Clarke, Chief Financial Officer

Tel: (973)228-6111 Fax: (973)228-1694

| FLEET NATIONAL BANK  |
|--|
| By<br>Printed Name   |
| Title  |
| ADDRESS FOR NOTICE AND DELIVERYS                             |
| Fleet National Bank  |
| 100 Federal Street   |
| Boston, MA 02110   |
| Attn: Structured Real Estate                                 |
| with a copy to the following address:<br>Fleet National Bank |
| 115 Perimeter Center Place N.E.                              |
| Suite 500  |
| Atlanta, GA 30346  |
| Attn: Lori Y. Litow  |
| Telephone: (770) 390-6544                                    |
| Telecopy: (770) 390-8434                                     |
| D D-t Cl 200/  |
| Pro Rata Share: 20%<br>Loan Commitment: \$40,000,000         |
| Loan Communent. \$40,000,000                                 |
| LIBOR OFFICE:  |
| Fleet National Bank  |
| 100 Federal Street   |
| Boston, MA 02110   |
|  |
| BANK OF AMERICA, N.A.  |
|  |
| By   |
| Printed Name   |
| Title  |
| ADDRESS FOR NOTICE AND DELIVERY                              |
| Address:   |
|  |

Bank of America, N.A. Real Estate Group - SDG 12th Floor Mail Center IL-1231-12-12 231 South LaSalle Street Chicago, IL 60697 Attn: Jeff Kahl

Telephone: (312) 828-1739 Telecopy: (312) 974-4970

Pro Rata Share: 17.5%

Loan Commitment: \$35,000,000

LIBOR OFFICE:

Address:

Bank of America, N.A. Real Estate Group - SDG

12th Floor

Mail Code: IL1-231-12-08 231 South LaSalle Street Chicago, IL 60697 Attn: Arveste Spencer

Telephone: (312) 828-5146 Telecopy: (312) 828-3950

| ICN    | DER: |
|--------|------|
| LLLIN. | DĽR. |

#### COMMERZBANK AG, NEW YORK BRANCH

| By           |  |
|--------------|--|
| Printed Name |  |
| Title        |  |
|              |  |
| By           |  |
| Printed Name |  |
| Title        |  |

#### ADDRESS FOR NOTICE AND DELIVERY:

### Address:

Commerzbank AG, New York Branch Two World Financial Center 34th Floor New York, NY 10281 Attn: Douglas P. Traynor

Telephone:(212)266-7569 Telecopy: (212)266-7530

Pro Rata Share: 17.5%

Loan Commitment: \$35,000,000

### LIBOR OFFICE:

Address:

Commerzbank AG, New York Branch

Two World Financial Center

34th Floor

New York, NY 10281 Attn: Douglas P. Traynor

| _  | _    |      | _            | _   | _  |
|----|------|------|--------------|-----|----|
| т. | - 17 | NI   | $\mathbf{r}$ | IL. | n. |
|    | . г  | 1.71 |              | ,,, | к  |
|    |      |      |              |     |    |

## PNC BANK, NATIONAL ASSOCIATION

| By            |  |
|---------------|--|
| Printed Name_ |  |
| Title         |  |

#### ADDRESS FOR NOTICE AND DELIVERY:

Address:

PNC Bank, National Association Two Tower Center, 18th Floor East Brunswick, NJ 08816

Attn: Thomas Hyland

Telephone: (732) 220-3561 Telecopy: (732) 220-3744

Pro Rata Share: 17.5%

Loan Commitment: \$35,000,000

LIBOR OFFICE:

Address:

PNC Bank, National Association Two Tower Center, 18th Floor East Brunswick, NJ 08816 Attn: Thomas Hyland

Telephone: (732) 220-3561 Telecopy: (732) 220-3744

LENDER:

WACHOVIA BANK, NATIONAL ASSOCIATION

| By   |  |  |  |
|------|--|--|--|
| Its_ |  |  |  |

#### ADDRESS FOR NOTICE AND DELIVERY:

Address:

Wachovia Bank, National Association

Mail Code: NC 0172 One First Union Center 301 S. College Street Charlotte, NC 28288-0172

Attn: David B. Hoagland, Vice President

Tele: (704) 374-4809 Fax: (704) 383-6205

Pro Rata Share: 17.5%

Loan Commitment: \$35,000,000

LIBOR OFFICE:

Address:

Wachovia Bank, National Association

Mail Code: NC 5604 One First Union Center 301 S. College Street Charlotte, NC 28288-0166

Attn: David B. Hoagland, Vice President

Tele: (704) 374-4809 Fax: (704) 383-6205

LENDER:

BAYERISCHE LANDESBANK, CAYMAN ISLANDS BRANCH

| By_     |     |
|---------|-----|
| ]       | Its |
|         |     |
|         |     |
| By      |     |
| <i></i> | Its |

# ADDRESS FOR NOTICE AND DELIVERY:

## Address:

Bayerische Landesbank, Cayman Islands Branch

560 Lexington Avenue New York, NY 10022 Attn: Daniel J. Reddy

Tele: 212-310-9881 Fax: 212-230-9114

Pro Rata Share: 10%

Loan Commitment: \$20,000,000

## LIBOR OFFICE:

Address:

Bayerische Landesbank, Cayman Islands Branch

560 Lexington Avenue New York, NY 10022 Attn: Patricia Sanchez

Tele: 212-310-9810 Fax: 212-310-9930